%% Santosh Bhattarai.
%%
%%

\documentclass[12pt]{report}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% Standard AMS packages
%\usepackage{charter} % The "charter" font
\usepackage{textcomp}
\usepackage{amssymb, amsmath, amsthm, enumerate}
\usepackage{verbatim}
%\usepackage{amsfonts}
\usepackage{makeidx}
\usepackage[british]{babel}
\usepackage[section,verbose]{placeins}
\usepackage{fancyhdr}
\usepackage{sectsty}
\usepackage{url}
\usepackage{appendix}
\usepackage{afterpage}
\usepackage{microtype}
\usepackage{amsmath,amssymb,amscd,mathrsfs,amscd}
\usepackage{graphics,verbatim}
\usepackage{tikz}
\pagestyle{plain}
%\usepackage{fullpage}
\usepackage{amsthm}
\usepackage{graphicx}
\usepackage{amsfonts}
\usepackage{amsopn}
\usepackage{amsmath, amsfonts, amssymb}

\usepackage[nottoc]{tocbibind}
\usepackage[mathscr]{eucal}



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%  For mathematical diagrams
\usepackage{xypic}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% For inserting graphic files
\usepackage{graphicx} %



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% Graduate college required margins
\usepackage[left=1.7in, top=1.2in, right=1.2in, bottom=1in]{geometry}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% line spacing
\usepackage{setspace}
    \doublespacing
    %\onehalfspacing
    %\singlespacing
\usepackage{yhmath}
\usepackage{mathtools}
\usepackage[all]{xy}
\usepackage{eufrak}
\usepackage{arial}
\usepackage{titlesec}
\usepackage{sectsty}

\sectionfont{\small}
\subsectionfont{\small}


\newcommand{\Z}{\mathbb{Z}}
\newcommand{\N}{\mathbb{N}}
\newcommand{\Q}{\mathbb{Q}}
\newcommand{\C}{\mathbb{C}}
\newcommand{\R}{\mathbb{R}}
\newcommand{\h}{\mathbf{H}}
\newcommand{\V}{\mathbf{V}}
\newcommand{\T}{\mathbf{T}}
\newcommand{\p}{\mathbf{P}.}
\newcommand{\x}{X_0^+}
\newcommand{\End}{\operatorname{End}}
\newcommand{\df}{\operatorname{def}}
\newcommand{\osp}{\mathfrak{osp}}
\newcommand{\s}{\mathfrak{sl}}
\newcommand{\Ext}{\operatorname{Ext}}
\newcommand{\Hom}{\operatorname{Hom}}
\newcommand{\atyp}{\operatorname{atyp}}
\newcommand{\fg}{\mathfrak{g}}
\newcommand{\fh}{\mathfrak{h}}
\newcommand{\fa}{\mathfrak{a}}
\newcommand{\fb}{\mathfrak{b}}
\newcommand{\fp}{\mathfrak{p}}
\newcommand{\ff}{\mathfrak{f}}
\newcommand{\gl}{\mathfrak{gl}}
\newcommand{\F}{\mathcal{F}}
\newcommand{\fv}{\mathcal{V}}
\newcommand{\X}{\mathcal{X}}
\newcommand{\0}{\bar{0}}
\renewcommand{\1}{\bar{1}}
\newcommand{\tf}{\tilde{f}}


\def \cas{\mathop{\hbox{\rm Cas}}\nolimits}
\def \diag{\mathop{\mathrm{diag}}\nolimits}
\def \dd {D(2,1;\al)}
\def \tq {\ |\ \ }
\def\blouk{\vphantom{\int}}
\def\toto#1{\noindent{\em{(#1)}}\ }
\def \dd {D(2,1;\alpha)}
\def \gmod {\fg\hbox{-}\mathrm{mod}}
\let\noi=\noindent
\let\dspt=\displaystyle


\def \cas{\mathop{\hbox{\rm Cas}}\nolimits}
\def \diag{\mathop{\mathrm{diag}}\nolimits}
\def \dd {D(2,1;\al)}
\def \tq {\ |\ \ }
\def\blouk{\vphantom{\int}}
\def\toto#1{\noindent{\em{(#1)}}\ }
\def \dd {D(2,1;\alpha)}
\def \gmod {\fg\hbox{-}\mathrm{mod}}
\let\noi=\noindent
\let\dspt=\displaystyle
\let \l=\lambda
\let \Ga=\Gamma
\let \a=\alpha
\let \b=\beta
\let \d=\delta
\let \Ga=\Gamma
\let \l=\lambda
\let \L=\Lambda
\let \eps=\varepsilon
\let \w=\omega
\let \e=\eps



%
%\titleformat{\chapter}[display]
%{\normalfont\huge\bfseries}{\chaptertitlename\ \thechapter}{20pt}{\Huge}
%\titlespacing*{\chapter}{0pt}{0pt}{40pt}

\titleformat{\chapter}[display]
{\normalfont\large\bfseries}{\chaptertitlename\ \thechapter}{14pt}{\large}
\titlespacing*{\chapter}{0pt}{0pt}{40pt}




\setcounter{secnumdepth}{3} %subsubsections are numbered
\setcounter{tocdepth}{3} %subsubsections are added to the TOC


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% Personal Environments and Commands
\newtheorem{thm}{Theorem}[chapter]
\newtheorem{lem}[thm]{Lemma}
\newtheorem{cor}[thm]{Corollary}
%\theoremstyle{definition}
\newtheorem{dfn}[thm]{Definition}
%\theoremstyle{remark}
\newtheorem{rk}[thm]{Remark}
\newtheorem{ex}[thm]{Example}
\newtheorem{prop}[thm]{Proposition}
\newtheorem{conj}[thm]{Conjecture}
\numberwithin{equation}{chapter}
\numberwithin{figure}{chapter}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\renewcommand{\bibname}{References}

%%%%

\DeclareMathOperator{\func{}}\text{} %to write real and imaginary parts as Re and Im.
\DeclareMathOperator{\tciFourier}{\mathscr{F}} %for fourier operator
\DeclareMathOperator{\dint}{\int}

%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% THESIS %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{document}
\begin{thebibliography}{99}

{\singlespacing
\bibitem{BC} D.~Benson and J.~Carlson, Diagrammatic methods for modular representations and cohomology, \emph{Communications in Algebra}, 15(1\&2), 1987, 53-121.
\bibitem{BKN1} B.~D.~Boe, J.~R.~Kujawa, and D.~K.~Nakano, Complexity for modules over the classical Lie superalgebra $\mathfrak{gl}(m|n)$, \emph{Compositio Mathematica}, \textbf{148}, 2012, 1561-1592.
\bibitem{BKN2}------, Complexity and module varieties for classical Lie superalgebras, \emph{Int. Math. Res. Not.}, 2011, Issue 3, 696-724.
\bibitem{BKN3} ------, Cohomology and support varieties for Lie superalgebras, Trans. Amer. Math. Soc. \textbf{362}, 2010, no. 12, 6551-6590.
\bibitem{BKN4} ------, Cohomology and support varieties for Lie superalgebras II, \emph{Proc. London Math. Soc.}, \textbf{98} 2009, no. 1, 19-44.
\bibitem{kujawa} J.~Kujawa, The generalized Kac-Wakimoto conjecture and support varieties for the Lie superalgebra $\osp(2|2n)$, \emph{SE Lie Theory Conf. Proc., Proc. Symp. Pure Math.}, Volume 86, 2012.
\bibitem{Car} J.~F.~Carlson, The varieties and the cohomology ring of a module, \emph{Journal of Algebra}, 85, no.~1, 1983, 104-143.
\bibitem{CWZ} S.~Cheng, W.~Wang, R.~B.~Zhang, A fock space approach to representation theory of $\osp(2|2n)$, \emph{Transformation Groups}, \textbf{12}, No. 2, 2007, 209-225.
\bibitem{DS} M.~Duflo and V.~Serganova, On associated variety for Lie superalgebras, 2005, arXiv:math/0507198.
\bibitem{Germoni} J.~Germoni, Indecomposable representations of $\osp(3,2)$, $\dd$ and $G(3)$, \emph{Colloquium on Homology and Representation Theory} (Spanish) (Vaqueras, 1998) \emph{Bol.  Acad. Nac. Cienc.} (Crdoba) \textbf{65} (2000), 147-163.
\bibitem{FH} W.~Fulton and J.~Harris, Representation Theory, A First Course, Graduate Texts in Mathematics 129, Berlin, New York: \emph{Springer-Verlag}, ISBN 978-0-387-97495-8, MR 1153249.
\bibitem{GS}C.~Gruson and V.~Serganova, Cohomology of generalized supergrassmannians and character formulae for basic classical Lie superalgebras, \emph{Proc. Lond. Math. Soc.} (3) 101, 2010, no. 3, 852-892.
\bibitem{Pe} I.~Penkov, Generic representations of classical Lie superalgebras and their localization, \emph{Monatsh. Math.} \textbf{118} 1994, no. 3-4, 267-313.
\bibitem{S} V.~Serganova, On superdimension of an irreducible representation of a basic classical Lie superalgebra, \emph{Supersymmetry in Mathematics and Physics,} Springer, Heidelberg, 2011, 253-273.
%\bibitem{S1} V.~Serganova, Blocks in the category of finite-dimensional representations of $\mathfrak{gl}(m|n)$, ********.
\bibitem{Kac} V.~Kac, Lie superalgebras, \emph{Adv.\  Math.} \textbf{26}, 1977, 8-96.
\bibitem{Kac1} V.~Kac, Representations of classical Lie superalgebras, in \emph{Differential geometrical methods in mathematical physics II (Proceedings, University of Bonn, July 13-16, 1977)}, Lecture Notes in Mathematics, \textbf{676}, Springer, Berlin, 1978, 597-626.
\bibitem{KW} V.~Kac and M.~Wakimoto, Integrable highest weight modules over affine superalgebras and number theory, \emph{Progr.\   Math.}, \textbf{123}, Birkhauser Boston, Boston, MA, 1994, 415-456.
\bibitem{Alp} J.~L.~Alperin, Periodicity in groups, \emph{Illinois J. Math}, \textbf{21}, 1977, no. 4,776-783.
\bibitem{FS} E.~M.~Friedlander and A.~Suslin, Cohomology of finite group schemes over a field, \emph{Invent. Math.}, \textbf{127}, 1997, no. 2, 209-270.
\bibitem{HUM} J.~E.~Humphreys, Introduction to Lie algebras and representation theory, \emph{Springer} (1972).
\bibitem{L} P.~Littleman, A generalization of the Littlewood-Richardson Rule, \emph{Journal of Algebra}, \textbf{130}, 328-368 (1990).
\bibitem{Mart} L.~Martirosyan, The representation theory of the exceptional Lie superalgebras $F(4)$ and $G(3)$, arXiv:1309.0418.
\bibitem{Musson} I.~M.~Musson, Lie superalgebras and enveloping algebras, \emph{Graduate Studies in Mathematics, AMS} vol. \textbf{131} (2012).
\bibitem{Sch} M.~Scheunert, The theory of Lie superalgebras, \emph{Lecture Notes in Mathematics,} 716, Springer–Verlag, Berlin, Heidelberg, New York (1979).
}


\end{thebibliography}


\title{Complexity of Modules over Lie Superalgebras}
\author{Houssein El Turkey}
\date{June 2014}

\pagenumbering{roman}
{%% PAGE NUMBERING ROMAN

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% TITLE PAGE  %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
{\singlespacing

\newpage
\thispagestyle{empty}
\begin{center}
{ %\large
UNIVERSITY OF OKLAHOMA
\par
\vspace{0.16in}
GRADUATE COLLEGE
\par
\vspace{1.2in}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%5
COMPLEXITY OF MODULES OVER LIE SUPERALGEBRAS
\par
\vspace{0.17in}
\par
\vspace{1.2in}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
A DISSERTATION
\par
\vspace{0.17in}
SUBMITTED TO THE GRADUATE FACULTY
\par
\vspace{0.17in}
in partial fulfillment of the requirements for the
\par
\vspace{0.17in}
Degree of
\par
\vspace{0.17in}
DOCTOR OF PHILOSOPHY
\par
\vfill
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
By
\par
\vspace{0.17in}
HOUSSEIN EL TURKEY
\par
%\vspace{0.05in}
%Department of Mathematics
%\par
%\vspace{0.05in}
Norman, Oklahoma
%\par
%Norman, OK 73019 USA
\par
%\vspace{0.05in}
2014
}
\end{center}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% SIGNATURE PAGE  %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newpage
\thispagestyle{empty}
\begin{center}
{%\large
COMPLEXITY OF MODULES OVER LIE SUPERALGEBRAS
\par
%\vspace{0.17in}
\par
\vspace{0.9in}
A DISSERTATION APPROVED FOR THE
\par
%\vspace{0.17in}
DEPARTMENT OF MATHEMATICS
\par
\vspace{0.9in}
BY
\par
\vfill
\begin{flushright}
\begin{tabular}{cr}
\hline
\  \  \  \  \  \  \  & Dr.~Jonathan Kujawa, Chair\\
 \\ &  \\
  \\ &  \\
\hline
  & Dr.~Ralf Schmidt \\
\\ & \\
 \\ &  \\
\hline
 & Dr.~Alan Roche \\
\\   & \\
 \\ &  \\
\hline
 & Dr.~Tomasz Przebinda \\
 \\ &  \\
  \\ &  \\
\hline
   & Dr.~Patrick Livingood \\
\end{tabular}
\end{flushright}
}
\end{center}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% COPYRIGHT PAGE  %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newpage
\thispagestyle{empty}
\   \
\par
\vfill
\begin{center}
\copyright \    Copyright by HOUSSEIN EL TURKEY \ 2014

All Rights Reserved.
\end{center}
}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% ACKNOWLEDGEMENTS PAGE  %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\chapter*{Acknowledgements}

{\doublespacing
I have been blessed to have many amazing people in my life whom I am very grateful for their encouragement and support.
Thanks to my sister Salam Turki, whose constant love and support made me able to achieve what I did not even dream about achieving. Salam, you have been always there for me especially during my last year of graduate school when applying for jobs, working on the dissertation, and teaching allied to form a stressful burden on my academic and personal life. Thank you Salam, you deserve much greater recognition than I could offer.

Thanks to my parents for their unwavering love and support. They worked extremely hard to help me get to where I am at. I hope I made you proud. I am thankful for you for bringing up a big loving caring family. Thanks to my brothers and sisters who supported me financially and emotionally throughout my life. I know that you missed going to college because of financial reasons and the war in Lebanon, but I hope I will help your kids as much as you helped me.

Thanks to Dr. Jonathan Kujawa. I am so fortunate to have had you as my PhD advisor. Your continuous unlimited support was the main reason behind this work. You have taught me lots of mathematics, you motivated me to go to conferences, you supported me financially to go to conferences, you encouraged me to present my work anytime I get the chance, you nominated me for the outstanding graduate student (which I received thanks to your letter), you helped me immensely in the job hunting process. You were even as stressed as I was about finding a decent job. Many times I wondered how can you do all of that besides being a great teacher and a great researcher. But then I remember my spontaneous joke about you at the graduate student conference at KSU when I was presenting our joint work on Schur Superalgebras. I accidently said: ‘’This is a joint work with my \textbf{SUPER ADVISOR} Jonathan Kujawa.” Your dedication to support me is something I can never match. With no intention to offend the other faculty in the mathematics department, but you are ''The Advisor`` to have.

Thanks to Dr. Ralf Schmidt, Dr. Alan Roche, and Dr. Tomasz Przebinda for overlooking my progress in the doctoral program. You are inspiring mathematicians and I look up for you. I am thankful for teaching me a great deal of mathematics and for helping me out in some Lie theoretic problems. On that note, special Thanks goes to Julia Maddox for helping me out with many aspects of the Lie algebra $\mathfrak{sp}(2n)$. Your remarks were very useful and informative.

Thanks to Dr. Sepideh Stewart and Dr. Milos Savic. I believe that my conversations with you made me a better teacher. I learned many teaching methodologies that has been helping me connect with the students more and present mathematics in a very clear way.

Thanks to all my teachers from elementary school to graduate school. It is you who made me succeed. It is you who showed me the power of proper teaching. I am lucky to have had an amazing group of teachers throughout my life. Special Thanks goes to Dr. Abdul-Naser Kassar, Dr. Nazih Nahlus, Dr. Hazar abu Khuzam (my master thesis advisor), and  Dr. Sabine El Khoury.

Thanks to Ms. Anne Jones for being a warm, friendly face in the department during my years in graduate school. Thanks to Ms. Cristin Sloan for picking up where Anne left off. Cristin, you are such a wonderful person to be around. I am thankful for your tolerance of me invading your office on a lot of occasions.

Lastly, I would like to thank all my friends. Special thanks goes to my friend Dania Sheaib for providing that extra boost of motivation to finish my undergraduate degree, my master’s degree, and my doctoral degree all on time. We have known each other for twelve years during which we went through joyful and stressful moments. Thanks to all my friends in the Lebanese Student Association. Having this support system around is undeniably one of the best resources for success. I greatly enjoyed all of our events especially working with you to organize the Lebanese Cultural Night. I will forever remember our late night hangouts. Thanks to my friend and long time tennis partner Tyler Diedrick. I will definitely miss our long tennis matches.

Thanks to my friends and colleagues in the mathematics department for battling with me in the academic trenches of graduate school.  Thanks to my friends who graduated with me from BAU and AUB in Lebanon. I miss you all.


}




%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% TABLE OF CONTENTS PAGE  %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


{\onehalfspacing
\tableofcontents

%\listoffigures
%\listoftables
}
\newpage

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% ACKNOWLEDGEMENTS PAGE  %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\chapter*{Abstract}
{\doublespacing

In this dissertation, a fair amount of work is dedicated to computing the complexity of modules over a classical Lie superalgebra $\fg=\fg_{\0}\oplus \fg_{\1}$ over the complex numbers $\C$. We will consider the category $\F$ of finite dimensional $\fg$-supermodules which are completely reducible as $\fg_{\0}$-modules.  Every module $M\in \F$ admits a minimal projective resolution whose terms have dimensions which increase at a polynomial rate of growth. This rate of growth is called the \emph{complexity} of $M$. In \cite{BKN1} the authors compute the complexity of the simple and the Kac modules over the general linear Lie superalgebra $\gl(m|n)$ of type $A$. A natural continuation to their work is computing the complexity of the same family of modules over the ortho-symplectic Lie superalgebra $\osp(2|2n)$ of type $C$. The two Lie superalgebras are both of \emph{Type I}, thus the Kac modules in the two cases are constructed by the same induction functor. This similarity will result in similar computations. In fact, our geometric interpretation of the complexity agrees with theirs. The complexity is not a categorical invariant. However, we compute a categorical invariant called the $z$-complexity, introduced in \cite{BKN1}, and we interpret this invariant geometrically in terms of a specific detecting subsuperalgebra. In addition, we compute the complexity and the $z$-complexity of the simple modules over the \emph{Type II} Lie superalgebras $\osp(3|2)$, $\dd$, $G(3)$, and $F(4)$.
}

\newpage
} %% END PAGE NUMBERING ROMAN



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\pagenumbering{arabic}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% INTRODUCTION %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\chapter{Introduction}

Let $A$ be a %finite-dimensional cocommutative Hopf algebra over an algebraically closed field $k$ (or equivalently a
finite group scheme and consider the category of rational modules for $A$. It is well known that %$A$ is a Frobenius algebra, meaning that
an $A$-module is projective if and only if it is injective. Moreover, %by deep results involving the finite generation of the cohomology ring %(due to Friedlander and Suslin),
 every rational $A$-module admits a minimal projective resolution which has polynomial rate of growth. This rate of growth, called the \emph{complexity}, was first introduced by Alperin in 1977 (cf. \cite{Alp}). Carlson \cite{Car} later introduced the idea of the support variety of a module over group algebras whose dimension coincides with the complexity of the module. %Support varieties were later defined for other finite-dimensional cocommutative Hopf algebras. \\

In this dissertation, we will be considering a classical Lie superalgebra $\fg=\fg_{\0}\oplus\fg_{\1}$ over the complex numbers $\C$. By definition there exists a connected reductive algebraic group $G_{\0}$ such that $\operatorname{Lie}(G_{\0})=\fg_{\0}$. The simple classical Lie superalgebras were classified by Kac \cite{Kac}. The authors in \cite{BKN3} used relative cohomology for the pair $(\fg,\fg_{\0})$ to investigate the combinatorics and representation theory of the category finite-dimensional representations of $\fg$. In fact, the cohomology ring $R=\operatorname{H}^\bullet(\fg,\fg_{\0};\C)$ is finitely generated because $G_{\0}$ is reductive. In \cite{BKN3,BKN4} the authors initiated a study of support varieties for $\F$ using relative cohomology for the pair $(\fg,\fg_{\0})$.  The construction of these support varieties provides a homological framework which encapsulates the important combinatorial notions of atypicality and defect as defined by Kac and Wakimoto.

We will be working with the category $\F:=\F_{(\fg,\fg_{\0})}$ of finite-dimensional $\fg$-supermodules which are completely reducible over $\fg_{\0}$. The authors in \cite{BKN2} showed that $\F$ has enough projectives and it satisfies: $(i)$ it is a self-injective category and $(ii)$ every module in this category admits a projective resolution which has a polynomial rate of growth. For a module $M\in \F$, the complexity $c_{\F}(M)$ is the rate of growth of the minimal projective resolution of $M$. It should be noted that the support varieties (mentioned above) of a supermodule do not have their dimension equal to the complexity of the supermodule. For example, for the simple Lie superalgebra $\gl(1|1)$, the complexity of the trivial supermodule is equal to 2, but the dimension of the $(\fg,\fg_{\0})$-support variety is $1$ (which is also the atypicality).

This suggests that it is not sufficient to consider only these cohomological varieties when studying the category $\F$ and leads us to introduce new tools in the study of $\F$. Recently Duflo and Serganova introduced associated varieties for objects in $\F$. These varieties detect projectivity (that is, are trivial for a given supermodule if and only if the supermodule is projective). However, they do not have a direct connection to the cohomology of the category $\F$. For type I Lie superalgebras, the authors in \cite{BKN2} introduced new cohomologically defined varieties which are naturally subvarieties of the Duflo-Serganova associated varieties. Using these newly defined subvarieties together with support varieties we give a geometric interpretation of the complexity. This was first done in \cite{BKN1} for the type $A$ Lie superalgebra $\gl(m|n)$. This geometric interpretation of the complexity will be shown to hold over the type $C$ Lie superalgebra $\osp(2|2n)$, the type $B$ Lie superalgebra $\osp(3|2)$, and the three exceptional ones $D(2,1,\alpha)$, $G(3)$, and $F(4)$.

In chapter~2 we compute the complexity of the simple and the Kac modules for the orthosymplectic Lie superalgebra $\osp(2|2n)$. Let $K(\lambda)$ (resp. $L(\lambda)$) be the Kac (resp. simple) module of highest weight $\lambda$. Let $\atyp(\lambda)$ denote the atypicality of $\lambda$ (see section~\ref{ss:atyp}). We will see that $\atyp(\lambda)$ is either zero or one. For typical $\lambda$ (i.e. $\atyp(\lambda)=0$), the simple and the Kac modules are projective and hence they have a zero complexity. For atypical $\lambda$ (i.e. $\atyp(\lambda)=1$), the complexity is computed in Theorems~\ref{t2:atypcomplexitysimple} and \ref{t1:atypcomplexityKac}
\[c_{\F}(L(\lambda))=2n+1,\quad c_{\F}(K(\lambda))=2n.\]

These computations can be interpreted geometrically as follows. For a module $M$, let $\X_M$ denotes the associated variety defined by Duflo and Serganova \cite {DS}, and $\fv_{(\fg,\fg_{\0})}(M)$ the support variety as defined in \cite{BKN4}. Then, if $X(\lambda)$ is a Kac or a simple $\osp(2|2n)$-module, we have the geometric interpretation of the complexity in Theorem~\ref{ospgeometric1}~:
\begin{equation}\label{compinterpret}
 c_{\F}(X(\lambda))=\dim \X_{X(\lambda)}+\dim \fv_{(\fg,\fg_{\0})}(X(\lambda)).
\end{equation}

The authors in \cite{BKN1} introduced a categorical invariant called the $z$-complexity of modules and denoted by $z_{\F}(-)$ (see \cite[Section~9]{BKN1}). They computed the $z$-complexity of the simple and the Kac modules over $\gl(m|n)$ and then used a detecting subsuperalgebra $\ff$ to interpret their computations geometrically. In chapter~2, we carry these computations over $\osp(2|2n)$ and conclude in Theorem~\ref{ospgeometric2} that if $X(\lambda)$ is a Kac or a simple module, we have
\begin{equation}\label{zcompinterpret}
z_{\F}(X(\lambda))=\dim \fv_{(\ff,\ff_{\0})}(X(\lambda)).
\end{equation}
It is worth noting that our geometric interpretations of the complexity and the $z$-complexity over $\osp(2|2n)$ agree with the results obtained in \cite{BKN1}. This was expected since both types $A$ and $C$ are \emph{Type I} Lie superalgebras (Section~\ref{ss:atyp}). In particular, the Kac modules are defined similarly in the two cases and they are finite-dimensional. It was interesting to know if these interpretations would hold over \emph{Type II} Lie superalgebras, hence we computed the complexity and the $z$-complexity of the simple (finite-dimensional) modules over $\osp(3|2)$, and the three exceptional Lie superalgebras $\dd$, $G(3)$, and $F(4)$. In chapters three through six, we show that equations~\eqref{compinterpret} and \eqref{zcompinterpret} hold for the simple modules over these Lie superalgebras. The results in this paper lead us to believe that a more general result holds~:
\begin{conj}\label{conj}
Let $\fg=\fg_{\0}\oplus\fg_{\1}$ be a basic classical Lie superalgebra over the complex numbers and let $\F:=\F_{(\fg,\fg_{\0})}$ be the category of finite-dimensional $\fg$-supermodules which are completely reducible over $\fg_{\0}$. If $M\in \F$, then
\begin{itemize}
\item[(a)] $c_{\F}(M)=\dim \X_{M}+\dim \fv_{(\fg,\fg_{\0})}(M).$
\item[(b)] $z_{\F}(M)=\dim \fv_{\ff}(M).$
\end{itemize}
\end{conj}

\begin{rk}
Through out this dissertation, we will be working over the complex numbers $\C$.\\
\end{rk}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\chapter{Preliminaries}

In this chapter I give a quick introduction to Lie superalgebras and their representations. I also introduce the notions of complexity, support varieties, associated varieties, $z$-complexity, and detecting subsuperalgebras. Some examples will be provided from the work in \cite{BKN1} over the Lie superalgebra $\gl(m|n)$. For more about Lie superalgebras, I refer the reader to \cite{Kac}, \cite{Musson}, and \cite{Sch}.

\section{Lie superalgebras}

\begin{dfn}
A superspace $V$ is a $\Z_2$-graded vector space, i.e., a direct sum of two vector spaces $V=V_{\0}\oplus V_{\1}$. Elements of $V_{\0}$ (resp. $V_{\1}$) are called \emph{even} (resp. \emph{odd}). Nonzero elements of $V_{\0}\cup V_{\1}$ are \emph{homogeneous} and for a homogeneous vector $v$, we write $\bar{v}\in \Z_2$ for the parity (or degree) of $v$.
\end{dfn}
We will be using the following important convention throughout this dissertation: if $\bar{v}$ appears in some formula or expression, then $v$ is assumed to be homogeneous.
\begin{dfn}
A superalgebra is a $\Z_2$-graded algebra $A=A_{\0}\oplus A_{\1}$ which satisfies $A_rA_s\subseteq A_{r+s}$ for all $r,\,s\in \Z_2$. A \emph{Lie superalgebra} is a $\Z_2$-graded vector space $\fg=\fg_{\0}\oplus\fg_{\1}$ together with a bilinear map $[\,,\,]:\fg\times\fg\rightarrow \fg$ which preserves the $\Z_2$-grading and satisfies graded versions of the usual Lie bracket axioms. Explicitly, for all $a,b,c\in \fg$ we have

 \begin{enumerate}
 \item[(a)] $[\fg_r,\fg_s]\subseteq \fg_{r+s}$ for $r,s\in \Z_2$ (preserves the $\Z_2$-grading),
 \item[(b)] $[a,b]=-(-1)^{\bar{a}\bar{b}}[b,a]$ (graded skew-symmetry),
 \item[(c)] $(-1)^{\bar{a}\bar{c}}[a,[b,c]]+(-1)^{\bar{a}\bar{b}}[b,[c,a]]+(-1)^{\bar{b}\bar{c}}[c,[a,b]]=0$ (graded Jacobi identity).
 \end{enumerate}
 \end{dfn}

 \begin{dfn}
 A Lie superalgebra $\fg$ is abelian if $[\fg,\fg]=0.$
 \end{dfn}

 \begin{dfn}
 A $\Z_2$-graded subspace $\fa$ of $\fg$ is an ideal if $[\fa,\fg]\subseteq \fa$.
 \end{dfn}
\section{Simple Lie superalgebras}
\begin{dfn}
A Lie superalgebra $\fg$ is \emph{simple} if it is not abelian and the only $\Z_2$-graded ideals of $\fg$ are $0$ and $\fg$.
\end{dfn}

The subspace $\fg_{\0}$ is a Lie algebra under the bracket and $\fg_{\1}$ is a $\fg_{\0}$-module. A finite dimensional Lie superalgebra $\fg$ is called \emph{classical} if there is a connected reductive algebraic group $G_{\0}$ such that $\operatorname{Lie}(G_{\0})=\fg_{\0}$, and an action of $G_{\0}$ on $\fg_{\0}$ which differentiates to the adjoint action of $\fg_{\0}$ on $\fg_{\1}$.\footnote{Unlike in Kac's original definition \cite{Kac}, we do not require a classical Lie superalgebra to be simple.} In particular, if $\fg$ is classical, then $\fg_{\0}$ is a reductive Lie algebra and $\fg_{\1}$ is semisimple as a $\fg_{\0}$-module. A \emph{basic classical} Lie superalgebra is a classical Lie superalgebra with a nondegenerate invariant supersymmetric even bilinear form. The simple (basic) classical Lie superalgebras were classified by Kac \cite{Kac}:

\begin{thm}
Let $\fg$ be a finite dimensional classical simple (basic) Lie superalgebra. Then either $\fg$ is a simple Lie algebra or $\fg$ is isomorphic to one of the following algebras:
\[A(m,n)=\operatorname{sl}(m+1,n+1)\quad\text{with $m>n\geq 0$},\]
\[A(n,n)=\operatorname{psl}(n+1,n+1)\quad\text{with $n\geq 1$},\]
\[B(m,n)=\operatorname{osp}(2m+1,2n)\quad\text{with $m\geq 0,n> 0$},\]
\[C(n)=\operatorname{osp}(2,2(n-1))\quad\text{with $n\geq 2$},\]
\[D(m,n)=\operatorname{osp}(2m,2n)\quad\text{with $m\geq 2,n\geq 1$},\]
\[D(2,1;\alpha)=\Gamma(1,-1-\alpha,\alpha)\quad \alpha\neq0,-1,\]
\[\operatorname{p}(n),\quad n\geq2;\quad \operatorname{q}(n),\quad n\geq2,\]
\[G(3), \,\,\text{a simple algebra of dimension $31$},\]
\[F(4), \,\,\text{a simple algebra of dimension $40$}.\]
\end{thm}
Among the Lie superalgebras listed above, we have the following isomorphisms:
\[A(1,0)\cong C(2)\quad D(2,1)\cong D(2,1,1).\]
In addition, there are some isomorphisms between the various $D(2,1;\alpha)$. There are no further isomorphisms between the algebras listed in the theorem.

\section{Roots}
Let $\fg$ be any finite dimensional Lie superalgebra such that $\fg_{\0}$ is reductive and $\fg_{\1}$ is a semisimple $\fg_{\0}$-module. Let $\fh_{\0}$ be a Cartan subalgebra of $\fg_{\0}$. For $\alpha\in \fh_{\0}^*$, set
\[\fg^\alpha=\{x\in \fg\,|\,[h,x]=\alpha(h)x \,\,\text{for all $h\in \fh_{\0}$}\},\]
and let
\[\Phi=\{\alpha\in \fh_{\0}^*\,|\,\alpha\neq 0, \fg^\alpha\neq 0\}\]
be the set of \emph{roots} of $\fg$. Since the action of $\fh_{\0}$ on any simple $\fg_{\0}$-modules is diagonalizable, it follows that the adjoint action of $\fh_{\0}$ on $\fg$ is diagonalizable. Thus there is a \emph{root space decomposition}
\[\fg=\fh\oplus \bigoplus_{\alpha\in \Phi}\fg^\alpha,\]
where $\fh=\fg^0$ is the centralizer of $\fh_{\0}$ in $\fg$.

Suppose $\mathfrak{b}$ is a subalgebra of $\fg$ containing $\mathfrak{h}$. We say that $\mathfrak{b}$ is a \emph{Borel subalgebra} of $\fg$ if
\begin{enumerate}
\item $\mathfrak{b}_{\0}=\mathfrak{b}\cap \fg$ is a Borel subalgebra of $\fg_{\0}$,
\item $\mathfrak{b}=\mathfrak{h}\oplus \mathfrak{n}$ with $\mathfrak{n}$ a nilpotent ideal of $\mathfrak{b}$,
\end{enumerate}
and $\mathfrak{b}$ is maximal with these properties. The root vectors that lie in $\mathfrak{b}$ will be called \emph{positive root vectors} and their corresponding roots $\l\in \Phi$ will be called the \emph{positive roots}. The set of positive roots will be denoted by $\Phi^+$. The \emph{negative roots} are $\Phi^-:=\Phi\setminus \Phi^+$.

\section{Supermodules}
 \begin{dfn}
 A $\fg$-supermodule $M$ is a superspace together with an action $\fg\times M\rightarrow M$ such that $\fg_r.M_s\subseteq M_{r+s}$ for $r,s\in \Z_2$.
 \end{dfn}

If $M$ and $M'$ are two superspaces, then the space $\operatorname{Hom}_{\C}(M,M')$ is naturally $\Z_2$-graded by declaring $f\in \Hom_{\C}(M,M')_r$ ($r\in \Z_2$) if $f(M_s)\subseteq M'_{r+s}$ for all $s\in \Z_2$.\\
Let $U(\fg)$ be the universal enveloping superalgebra of $\fg$. Let us describe the category of $\fg$-supermodules. The objects are all left $U(\fg)$-modules which are $\Z_2$-graded; that is, superspaces $M=M_{\0}\oplus M_{\1}$ satisfying $U(g)_rM_s\subseteq M_{r+s}$ for all $r,\,s\in \Z_2$. If $M$ is a $\fg$-supermodule, then $N\subseteq M$ is a subsupermodule if it is a supermodule which inherits its grading from $M$ in the sense that $M_r\cap N=N_r$ for $r\in \Z_2$. Given $\fg$-supermodules $M$ and $N$ one can define a $\fg$-supermodule struture on the contragradient dual $M^*$ and the tensor product $M\otimes N$ using the antipode and coproduct of $U(\fg)$. A morphism of $U(\fg)$-supermodules is an element of $\Hom_{\C}(M,M')$ satisfying $f(xm)=(-1)^{\bar{f}\bar{x}}xf(m)$ for all $m\in M$ and $x\in U(\fg)$. In this definition, $f$ and $x$ are assumed to be homogeneous. The general case can be obtained by linearity.

Let $\F:=\F_{(\fg,\fg_{\0})}$ be the category of finite-dimensional $\fg$-supermodules which are completely reducible over $\fg_{\0}$. In the case $\fg$ is one of the Lie superalgebras $\osp(3|2)$, $\dd$, $G(3)$, and $F(4)$, the even part $\fg_{\0}$ is semisimple, thus the category $\F$ is the category of finite-dimensional $\fg$-supermodules. Note that we allow all (not only graded) module homomorphisms, thus for a simple module $S$, it can happen that $\dim_{\C}\Hom_{\F}(S,S)$ is either one or two. This category decomposes into a direct sum of full subcategories called \emph{blocks} $\F^\chi$, where $\F^\chi$ consists of all finite dimensional modules with generalized central character $\chi$. A block having more than one element is called an \emph{atypical} block. The interested reader can find more about central characters and blocks in \cite[Section~8.2.4]{Musson}.

The category $\F$ has some properties that are of great importance to the computations in this work. To state these properties, we will need to use a fair amount of homological algebra. The interested reader can find the definitions of projective and injective modules, minimal projective resolutions, extensions, cohomology rings, etc. in any text on homological algebra.
\begin{dfn}
 A module $P$ is projective if for any modules $B,\,C$ with module homomorphisms $f:B  \twoheadrightarrow C$ and $g:P\rightarrow C$, there is a module homomorphism $h: P\rightarrow B$ such that $f\circ h=g$.
 \end{dfn}

 \begin{thm}
 Projective modules satisfy the following properties:
 \begin{itemize}
 \item A module $P$ is projective iff the functor $\Hom(P,\,\,)$ is exact.
 \item A module $P$ is projective iff it is a summand of a free module.
 \item The direct sum of projective modules is projective.
 \item Every summand of a projective module is projective.
 \end{itemize}
 \end{thm}
We can also define injective modules that satisfy similar (dual) properties. In our category $\F$, injective modules will not be used since:
\begin{prop}\cite[Prop.~2.2.2]{BKN2}
$\F$ is self-injective; that is, a module $M\in \F$ is projective if and only if it is injective.
\end{prop}

\begin{dfn}
A minimal projective resolution (in $\F$) of a module $M\in \F$ is an exact sequence:
\[\cdots\rightarrow P_d\rightarrow\cdots \rightarrow P_1\rightarrow P_0\rightarrow M\rightarrow 0,\]
such that each $P_d$ is projective with minimal dimension. The module $P_0$ is called the \textbf{projective cover} of $M$ since it is the minimal projective module that surjects onto $M$.
\end{dfn}

\begin{rk}
In our consideration of projective resolutions, we always assume the maps to be even.
\end{rk}

\begin{prop}\cite[Theorem~2.5.1]{BKN2}
Every module in $\F$ admits a minimal projective resolution whose terms have dimensions which increase at a polynomial
rate of growth.
\end{prop}
\iffalse
\section{Roots}
Let $\mathfrak{t}_{\0}$ be a Cartan subalgebra of $\fg_{\0}$. For $\l\in \mathfrak{t}_{\0}^*$, set
\[\fg^{\l}=\{x\in \fg\,|\, [h,x]=\l(h)x \quad \text{for all}\quad h\in \mathfrak{t}_{\0}\}.\]
and let
\[\Phi=\{\l\in \mathfrak{t}_{\0}^*\,|\, \l\neq 0,\,\fg^{\l}\neq 0\}\]
be the set of \emph{roots} of $\fg$. We have a root space decomposition
\[\fg=\mathfrak{h}\oplus \bigoplus_{\l\in \Phi} \fg^\l,\]
where $\mathfrak{h}=\fg^0$ is the centralizer of $\mathfrak{t}_{\0}$ in $\fg$. A nonzero element of $\fg^{\l}$ is called a \emph{root vector}.

Suppose $\mathfrak{b}$ is a subalgebra of $\fg$ containing $\mathfrak{h}$. We say that $\mathfrak{b}$ is a \emph{Borel subalgebra} of $\fg$ if
\begin{enumerate}
\item $\mathfrak{b}_{\0}=\mathfrak{b}\cap \fg$ is a Borel subalgebra of $\fg_{\0}$,
\item $\mathfrak{b}=\mathfrak{h}\oplus \mathfrak{n}$ with $\mathfrak{n}$ a nilpotent ideal of $\mathfrak{b}$,
\end{enumerate}
and $\mathfrak{b}$ is maximal with these properties. The root vectors that lie in $\mathfrak{b}$ will be called \emph{positive root vectors} and their corresponding roots $\l\in \Phi$ will be called the \emph{positive roots}. The set of positive roots will be denoted by $\Phi^+$. The \emph{negative roots} are $\Phi^-:=\Phi\setminus \Phi^+$.
\fi

\section{Type I/II Lie superalgebras}\label{ss:atyp}
A Lie superalgebra is said to be of \emph{Type I} if it admits a $\Z$-grading
\[\fg=\fg_{-1}\oplus\fg_{0}\oplus\fg_{1},\]
concentrated in degrees $-1,\,0,\,1$ with $\fg_{\0}=\fg_0$ and $\fg_{\1}=\fg_{-1}\oplus \fg_{1}$ and if the bracket respects this grading. Otherwise, $\fg$ is of \emph{Type II}. Examples of Type I Lie superalgebras include: $\mathfrak{gl}(m|n)$ and the simple Lie superalgebras of types $\operatorname{A}(m,n),\,\operatorname{C}(n)$ and $\operatorname{p}(n)$.

The simple modules for a Type I classical Lie superalgebra $\fg$ can be constructed as follows. Let $\mathfrak{t}_{\0}$ be a Cartan subalgebra of $\fg_{\0}$ and $X_0^+\subseteq \mathfrak{t}_{\0}^*$ be the set of dominant integral weights for $\fg_{\0}$ with respect to a fixed Borel subalgebra of $\fg_{\0}$. For $\lambda \in X_0^+$, let $L_{\0}(\lambda)$ be the simple finite dimensional module over the Lie algebra $\fg_{\0}$ of highest weight $\lambda$. Set
 \[\fp^{+}=\fg_{\0}\oplus\fg_{1}\quad \fp^{-}=\fg_{\0}\oplus\fg_{-1}.\]
Since $\fg$ is a Type I Lie superalgebra, $\fg_{\pm1}$ is an abelian ideal of $\fp^{\pm}$. We can then view $L_{\0}(\lambda)$ as a simple finite dimensional $\fp^{\pm}$-module via inflation through the canonical quotient $\fp^{\pm}\twoheadrightarrow \fg_{\0}$. For $\lambda \in X_0^+$, the \emph{Kac module} $K(\lambda)$ is defined by:
\[K(\lambda):=U(g)\otimes_{U(\fp^+)}L_{\0}(\lambda).\]

The Kac module $K(\lambda)$ has a unique maximal submodule. The head of $K(\lambda)$ is the simple finite dimensional $\fg$-module $L(\lambda)$. The set $\{L(\lambda)\mid \lambda\in X_0^+\}$ is a complete set of non-isomorphic simple modules in $\F=\F_{(\fg,\fg_{\0})}$. The projective cover of $L(\lambda)$ will be denoted by $P(\lambda)$. These are all finite dimensional.


\begin{dfn}
Let $\rho$ be half the sum of the positive even roots minus half the sum of the positive odd roots. Define the atypicality of $\lambda$ to be the maximal number of pairwise orthogonal positive isotropic (odd) roots which are also orthogonal to $\lambda+\rho$ with respect to the bilinear form on $\fh^*$. We will write $\atyp(\lambda)$ for the atypicality of $\lambda$. Note that $\atyp(\l)\leq \df(\fg)$ for all $\l$. We say that $\l$ is typical if $\atyp(\l)=0$, otherwise $\l$ is atypical. If $L(\l)$ is a simple $\fg$-modules of highest weight $\l$, then we define $\atyp(L(\l))~:=\atyp(\l)$.
\end{dfn}

If $L(\l)$ is a simple $\fg$-modules of highest weight $\l$, then we define $\atyp(L(\l))~:=\atyp(\l)$. It is known that the atypicality of a simple module is independent of the choice of Cartan and Borel subalgebras and, furthermore, is the same for all simple modules in a given block. Hence it makes sense to refer to the atypicality of a block. The principal block is the block containing the trivial module. It is known that (\cite[Theorem~1]{Kac1}) if $\atyp(\lambda)=0$, then $P(\lambda)=L(\lambda)$ and hence $L(\lambda)$ is projective. If $\fg$ is of \emph{Type I} and $\atyp(\l)=0$, then $P(\lambda)=L(\lambda)=K(\lambda)$.

\begin{dfn}
Let $\fg$ be a basic classical Lie superalgebra with a root system $\Phi$. Kac and Wakimoto \cite[Section~2]{KW} define the \emph{defect} of $\fg$, denoted by $\df(g)$, to be the maximal atypicality.
\end{dfn}

The defects for the various simple basic Lie superalgebras are given as follows \cite[Section~4]{DS}:
 $$\df(\mathfrak{sl}(m|n))=\min (m,n),\quad \df(\mathfrak{psl}(n|n))=\min (n,n),$$
$$\quad \df(\osp(2m|2n))=\df(\osp(2m+1|2n))=\min (m,n),$$
and the exceptional Lie superalgebras $\dd$, $G(3)$, and $F(4)$ all have defect $1$.

\begin{ex}\label{IntroEx1}
The standard example of a \emph{Type I} classical Lie superalgebra is $\fg=\mathfrak{gl}(m|n)$. The even component $\fg_{\0}$ is the span of the matrix units $E_{i,j}$ where $1\leq i,\,j\leq m$ or $m+1\leq i,\,j\leq m+n$. A basis for $\fg_{\1}$ is given by $E_{i,j}$ where $1\leq i\leq m$ and $m+1\leq j\leq m+n$ or $m+1\leq i\leq m+n$ and $1\leq j\leq n$. We can choose the Cartan subalgebra $\fh$ to be set of the diagonal matrices and the Borel subalgebra $\fb$ to be the set of all upper triangular matrices. The dual of the Cartan $\fh^*$ has basis $\{\eps_i~:~\fh\rightarrow \C\,|\, 1\leq i\leq m+n \}$ where $\eps_i$ is the linear functional which picks out the $i^{th}$ diagonal entry. With respect to this basis we define a bilinear form on $\fh$ by
\[(\eps_i,\eps_j)=\begin{cases}
                   \d_{i,j},&\quad 1\leq i,\,j\leq m;\\
                   -\d_{i,j},&\quad m+1\leq i,\,j\leq m+n;\\
                  0, & \quad \text{otherwise.}
                  \end{cases}\]
The set of roots for $\fg$ is
\[\Phi=\{\eps_i-\eps_j\,|\,1\leq i,\,j\leq m+n,\,i\neq j\}\]
and the positive roots are
\[\Phi^{+}=\{\eps_i-\eps_j\,|\, 1\leq i< j\leq m+n\}.\]
The even roots of $\fg$ are:
\[
\Phi_{\0}=\{\eps_i-\eps_j\,|\, 1\leq i\neq j\leq m\,\,\text{or}\,\,m+1\leq i\neq j\leq m+n\},\]
while the odd roots are $\Phi_{\1}=\Phi\setminus \Phi_{\0}.$ Using this data, we can compute
\[\rho=\frac{m-n-2}{2}\eps_1+\frac{m-n-4}{2}\eps_2+\ldots+\frac{-m-n}{2}\eps_m+\frac{m+n}{2}\eps_{m+1}+\ldots+\frac{m-n+2}{2}\eps_{m+n}.\]
The set of dominant integral weights for $\mathfrak{gl}(m|n)$ is given in \cite[Subsection~2.4]{BKN1} by
\[\x=\Bigg\{\l=\sum_{i=1}^{m+n}\l_i\eps_i\,|\, \l_i-\l_{i+1}\in \Z_{\geq 0}\,\,\text{for}\,\,i\neq m,\,m+n\Bigg\}.\]
For a weight $\l\in \x$, $\atyp(\l)=k$ is equivalent to having a set of $k$ positive odd pairwise orthogonal roots
\[ \{\eps_{i_t}-\eps_{j_t}\,|\, t=1,\ldots,k\},\]
such that $(\l+\rho,\eps_{i_t}-\eps_{j_t})=0$ for all $t=1,\ldots,k$ where $1\leq i_1,\ldots,i_k\leq m$ and $m+1\leq j_1,\ldots,j_k\leq m+n$. Then the defect of $\fg$ is $\min (m,n)$. We can then define the \emph{core} of $\l$ as the pair of multisets
\[\Bigg(\{(\l+\rho,\eps_r)\},\{(\l+\rho,\eps_s)\}\Bigg),\]
where $r\in\{1,\ldots,m\}\setminus \{i_1,\ldots,i_k\}$ and $s\in \{m+1,\ldots, m+n\}\setminus \{j_1,\ldots, j_k\}$. From \cite[Section~5]{GS} showed that two simple modules $L(\l)$ and $L(\mu)$ lie in the same block if and only if $\atyp(\l)=\atyp(\mu)$ and the core of $\l$ equals the core of $\mu$.

The trivial module has weight $\l=0$ whose atypicality is the same as $\df(\fg)$. If we take $m=n=1$, the simple modules in the principal block are one dimensional and indexed by $L(\l\eps_1-\l\eps_2)$ where $\l\in \Z$.
\end{ex}
\section{Complexity}
The authors in \cite{BKN2} showed that $\F$ has enough projectives, $(i)$ it is a self-injective category and $(ii)$ every module in this category admits a projective resolution which has a polynomial rate of growth. For a module $M\in \F$, define the complexity $c_{\F}(M)$ to be the rate of growth of the minimal projective resolution of $M$. More explicitly,
\begin{dfn}
Let $M$ be a $\fg$-module in the category $\F$ and $\p\twoheadrightarrow M$ be a minimal projective resolution. The rate of growth of $\p$, $r(\p)$, is the smallest nonnegative integer $c$ such that there exists a constant $C>0$ with $\dim P_d\leq Cd^{c-1}$ for all $d$. If no such integer exists, then $\p$ is said to have infinite rate of growth. Define the complexity of $M$ to be $c_{\F}(M):=r(\p)$.
\end{dfn}
Fortunately, the complexity of all modules in $\F$ is finite:
\begin{thm}\cite[Theorem~2.5.1]{BKN2}
Let $M$ be an object of $\F$. Then $c_{\F}(M)\leq \dim \fg_{\1}$.
\end{thm}
This concept of complexity was first introduced by Alperin for finite modules over groups \cite{Alp} in $1977$. One of the important features of the complexity is the fact that it measures how far the module is from being projective:
\begin{lem}\cite[Corollary~2.7.1]{BKN2}
$c_{\F}(M)=0$ if and only if $M$ is projective.
\end{lem}

\begin{cor}\label{typcomplexity}
If $\lambda$ is typical (i.e. $\atyp(\lambda)=0$), then $c_{\F}(L(\lambda))=0$. If $\fg$ is of \emph{Type I}, then $c_{\F}(K(\lambda))=0$.
\end{cor}
\begin{ex} \label{IntroEx2}
Let $\fg=\gl(1|1)$. The simple modules in the principal block have highest weights $\l\eps_1-\l\eps_2,$ $\l\in \Z$. Boe, Nakano, and Kujawa showed in \cite{BKN2} that $c_{\F}(L(\l\eps_1-\l\eps_2))=2$ ($\l\in \Z$). The projective cover $P(\l|-\l)$ of $L(\l|-\l)$ is four-dimensional with three radical layers. The head and socle of $P(\l|-\l)$ are both isomorphic to $L(\l|-\l)$, and the second layer is isomorphic to  $L(\l+1|-\l-1)\oplus L(\l-1|-\l+1)$. For simplicity, let us choose $\l=0$. They constructed an explicit minimal projective resolution of the trivial module $L(0|0)$:
\[\ldots \rightarrow P(2|-2)\oplus P(0|0)\oplus P(-2|2)\rightarrow P(1|-1)\oplus P(-1|1)\rightarrow P(0|0)\rightarrow L(0|0) \rightarrow 0.\]
Therefore, $\dim P_d=4(d+1)$ and $c_{\F}(L(0|0))=2$.
\end{ex}
More generally,
\begin{thm}\cite[Theorem~8.2.1]{BKN1}
Let $L(\l)$ be a simple $\gl(m|n)$-module of atypicality $k$. Then
\[c_{\F}(L(\l))=(m+n)k-k^2+k.\]
\end{thm}

\begin{ex} \label{IntroEx3}
Let $\fg=\gl(1|1)$. Given the structure of the projective indecomposable modules given in the previous section, we can verify that the Kac module $K(0|0)$ admits a minimal projective resolution of the form
\[\ldots \rightarrow P(2|-2)\rightarrow  P(1|-1)\rightarrow P(0|0)\rightarrow K(0|0) \rightarrow 0.\]
Therefore, $\dim P_d=4$ and $c_{\F}(K(0|0))=1$.
\end{ex}

More generally,

\begin{thm}\cite[Theorem~6.3.1]{BKN1}
Let $K(\l)$ be a Kac $\gl(m|n)$-module of atypicality $k$. Then
\[c_{\F}(K(\l))=(m+n)k-k^2.\]
\end{thm}

We write $\Ext_{(\fg,\fg_{\0})}^\bullet(M,N)$ for the relative cohomology for the pair $(\fg,\fg_{\0})$. For $M,N\in \F$, we will use the isomorphism:
\[\Ext_{\F}^\bullet(M,N)\cong\Ext_{(\fg,\fg_{\0})}^\bullet(M,N).\]
 The following proposition provides a characterization of the complexity via rates of growth of these extension groups:
\begin{prop}\cite[Proposition~2.8.1]{BKN2}\label{compcharacterization}
Let $\fg$ be a classical Lie superalgebra, and let $M$ be an object in $\F$. Then
\[c_{\F}(M)=r\Big(\Ext_{(\fg,\fg_{\0})}^\bullet(M,\bigoplus S^{\dim P(S)})\Big)\]
where the sum is over all the simple modules in $\F$, and $P(S)$ is the projective cover of $S$.
\end{prop}

\begin{proof}
Let $\p\twoheadrightarrow M$ be a minimal projective resolution of $M$, then we have
\[\Ext_{\F}^n\Big(M,\bigoplus S^{\dim P(S)}\Big)\cong \Hom_{\F}\Big(P_n,\bigoplus S^{\dim P(S)}\Big).\]
Let $[P_n:P(S)]$ be the number of times the projective cover $P(S)$ appears as a summand of the projective module $P_n$. Then we have,
\begin{align*}
\dim_{\C}\Ext_{\F}^n\Big(M,\bigoplus S^{\dim P(S)}\Big)& = \sum_S \dim_{\C}P(S)\dim_{\C}\Hom_{\F}(P_n,S)\\
                                                       & = \sum_S \dim_{\C}P(S)\dim_{\C}\Hom_{\F}(S,S)[P_n:P(S)]\\
                                                      % & \leq 2\dim_{\C} P_n.
\end{align*}
Then, by Schur's Lemma in the super setting, we have
\[ \dim_{\C} P_n\leq \dim_{\C}\Ext_{\F}^n\Big(M,\bigoplus S^{\dim P(S)}\Big)\leq 2\dim_{\C} P_n.\]
This proves the result.
 \end{proof}

%In \cite{BKN1} the authors compute the complexity of the simple and the Kac modules over the general linear Lie superalgebra $\gl(m|n)$ of type $A$. A natural continuation to their work is computing the complexity of the same family of modules over the ortho-symplectic Lie superalgebra $\osp(2|2n)$ of type $C$. The two Lie superalgebras are both of \emph{type I}, thus the Kac modules in the two cases are constructed by the same induction functor. This similarity will result in similar computations.\\

In the group case, the cohomology ring of $G$ is finitely generated and we can construct the support variety, $\fv_G(M)$, of a module $M$, with
$\dim \fv_G(M)=c_G(M)$. However, the relative cohomology $H^{\bullet}(\fg,\fg_{\0};\C)$ for $\gl(1|1)$ has Krull dimension one, and so is not large enough to use to construct a support variety theory which measures the complexity. This geometric interpretation will be completed by using both the cohomological variety and the associated variety of Duflo and Serganova \cite{DS}.


\section{Support varieties}
Let $R=H^{\bullet}(\fg,\fg_{\0};\C)$ be the cohomology ring of $\fg$ and let $M_1,\,M_2\in \F$. According to \cite[Theorem~2.7]{BKN3}, $\Ext_{\F}^{\bullet}(M_1,M_2)$ is a finitely generated $R$-module. Set $$J_{(\fg,\fg_{\0})}(M_1,M_2):=Ann_R(\Ext_{\F}^{\bullet}(M_1,M_2)).$$
 The \emph{relative support variety} of the pair $(M_1,M_2)$ is defined by
\[\fv_{(\fg,\fg_{\0})}(M_1,M_2):=MaxSpec(R/J_{(\fg,\fg_{\0})}(M_1,M_2)).\]
In the case $M=M_1=M_2$, set $J_{(\fg,\fg_{\0})}(M)=J_{(\fg,\fg_{\0})}(M,M)$, and
 \[\fv_{(\fg,\fg_{\0})}(M):=\fv_{(\fg,\fg_{\0})}(M,M).\]
 The variety $\fv_{(\fg,\fg_{\0})}(M)$ is called the \emph{support variety} of $M$. Note that in this situation $J_{(\fg,\fg_{\0})}(M)=Ann_R(Id)$ where $Id$ is the identity morphism in $\operatorname{Hom}_{\F}(M,M)$.\\
Boe, Kujawa, and Nakano had computed the support varieties of modules over certain families of Lie superalgebras. For example,

\begin{lem}\cite[Corollary~3.2]{BKN4}
Let $K(\l)$ be a Kac $\gl(m|n)$-module. Then
 $$\fv_{(\fg,\fg_{\0})}(K(\l))=0.$$
\end{lem}

 \begin{lem}\cite[Theorem~4.7]{BKN4}
 Let $L(\l)$ be a simple $\gl(m|n)$-module of atypicality $k$. Then $\fv_{(\fg,\fg_{\0})}(L(\l)\cong \mathbb{A}^k$. Hence $\dim \fv_{(\fg,\fg_{\0})}(L(\l))=\atyp(\l).$
 \end{lem}


\section{Rank varieties for \emph{Type I} Lie superalgebras}\label{ss:rankvar}
Let $\fg=\fg_{-1}\oplus\fg_{0}\oplus\fg_{1}$ be a \emph{Type I} Lie superalgebra. Observe that $\fg_{\pm1}$ is an abelian Lie superalgebra and, consequently,
 \[R_{\pm}:=H^{\bullet}(\fg_{\pm1},\{0\};\C)\cong S(\fg_{\pm1}^*)\]
 as graded algebras. Let $\F(\fg_{\pm1})$ be the category of finite dimensional $\fg_{\pm 1}$-modules. Define the $\fg_{\pm1}$ \emph{support variety} of $M\in \F(\fg_{\pm1})$ by:
  \[\fv_{\fg_{\pm1}}(M):=\fv_{(\fg_{\pm1},0)}(M).\]

 From \cite[Section~5]{BKN4}, we have $\fv_{\fg_{\pm1}}(M)$ is canonically isomorphic to the following rank variety:
\[\fv_{\fg_{\pm1}}^{rank}(M):=\{x\in \fg_{\pm1} \mid \text{M is not projective as a $U(\langle x \rangle)$-module}\}\cup\{0\},\]
where $U(\langle x \rangle)$ denotes the enveloping algebra of the Lie superalgebra generated by $x\in \fg_{\pm1}$. As a consequence of this isomorphism, we can use the various properties of a rank variety. For example, \cite[Sections~5,6]{BKN4}, it satisfies the tensor product rule
\[\fv_{\fg_{\pm1}}^{rank}(M\otimes N)=\fv_{\fg_{\pm1}}^{rank}(M)\cap \fv_{\fg_{\pm1}}^{rank}(N).\]
 and detects projectivity:
\begin{lem}\cite[Theorem~3.5.1(a)]{BKN2}
Let $M$ be a $\fg$-module in the category $\F$. Then $M$ is projective if and only if $\fv_{\fg_1}^{rank}(M)=\fv_{\fg_{-1}}^{rank}(M)=0$.
\end{lem}

As a consequence, if $\lambda\in \x$ is typical, $L(\lambda)$ and $K(\lambda)$ are projective and hence
\[\fv_{\fg_1}^{rank}(L(\lambda))=\fv_{\fg_1}^{rank}(K(\lambda))=0.\]

\section{Associated variety}
Let
\[\X=\{x\in \fg_{\1} \mid [x,x]=0\}.\]
Let $G_{\0}$ be an algebraic group with Lie algebra $\fg_{\0}$. Then $\X$ is invariant under the adjoint action of $G_{\0}$. It was shown in \cite{DS} that for any $x\in \X$, there exists $g\in G_{\0}$ and isotropic mutually orthogonal linearly independent roots $\alpha_1,\ldots, \alpha_k$ such that $Ad_g(x)=x_1+\ldots+x_k$ with $x_i\in \fg_{\alpha_i}$. The number $k$ does not depend on the choice of $g$ and is called the \emph{rank} of $x$ (notation $rk(x)=k$). Let
\[X_k=\{x\in \X\,|\,rk(x)=k\}.\]
If $s$ is the defect of $\fg$, then $\X=X_0\cup \ldots \cup X_s$ is a stratification of $\X$ such that $\overline{X_k}=X_0\cup \ldots \cup X_k$.

If $M \in \F$, then Duflo and Serganova \cite{DS} define an associated variety of $M$ which is equivalent to:
\[\X_M=\{x\in \X \mid \text{M is not projective as a $U(\langle x \rangle)$-module}\}\cup \{0\}.\]
As an example, we consider the work done in \cite{BKN1}:
\begin{thm}\cite[Corollary~6.4.1]{BKN1}
 Let $K(\l)$ be a Kac $\gl(m|n)$-module of atypicality $k$. Then $\X_{K(\l)}=\overline{(\fg_1)_k}$ where $\overline{(\fg_1)_k}$ is the closure of  the $G_{\0}$-orbit, $(\fg_1)_k$ described in \cite[Section~2.6]{BKN1}. Hence, $\dim \X_{K(\l)}= (m+n)k-k^2$.
 \end{thm}

On the other hand,

 \begin{thm}\cite[Theorems~4.5 and 5.4]{DS}
 Let $L(\l)$ be a simple $\gl(m|n)$-module of atypicality $k$. Then $ \X_{L(\l)}=\overline{(\fg_1)_k}$. Hence $\dim \X_{L(\l)}= (m+n)k-k^2$.
 \end{thm}

 Combining the results stated in the previous three sections, the following geometric interpretation of the complexity holds:

 \begin{thm}\label{thm1}
If $X(\lambda)$ is a Kac or a simple $\gl(m|n)$-module, then
\[c_{\F}(X(\lambda))=\dim \X_{X(\lambda)}+\dim \fv_{(\fg,\fg_{\0})}(X(\lambda)).\]
\end{thm}

\section{$z$-complexity}\label{zcompdef}
Complexity is not a categorical invariant; however, the authors in \cite{BKN1} defined a closely related invariant, called the $z$-complexity:
\begin{dfn}
Let $M$ be a module in $\F$. Consider
\[z_{\F}(M)=r\Big(\Ext_{(\fg,\fg_{\0})}^\bullet(M,\bigoplus S)\Big),\]
where the direct sum runs over all simple modules of $\F$. Unlike complexity, $z_{\F}(-)$ has the advantage of being invariant under category equivalences.
\end{dfn}

\begin{prop} \label{zcompcharacterization}
Let $M\in F$ and $\p \rightarrow M$ be a minimal projective resolution of $M$. Define $s(\p)$ to be the rate of growth of the number of summands in $P_d$. Explicitly, $s(\p)$ is the smallest nonnegative integer $c$ such that there exists a constant $C>0$ with
\[\text{number of summands in $P_d$}\leq Cd^{c-1},\quad \text{for all $d$.}\]
 We have $z_{\F}(M)=s(\p)$.
\end{prop}

\begin{proof}
The proof follows the same argument in the proof of Proposition~\ref{compcharacterization} where $\dim P(S)$ is removed from the computations.
\end{proof}

\begin{ex} \label{IntroEx4}
Let $\fg=\gl(1|1)$.  Recall the explicit minimal projective resolution of the trivial module $L(0|0)$:
\[\ldots \rightarrow P(2|-2)\oplus P(0|0)\oplus P(-2|2)\rightarrow P(1|-1)\oplus P(-1|1)\rightarrow P(0|0)\rightarrow L(0|0) \rightarrow 0.\]
The number of summands in each $P_d$ is $d+1$, thus the rate of growth of this number is $2$ and hence $z_{\F}(L(0|0))=2$.
\end{ex}
More generally,

\begin{thm}\cite[Theorem~9.1.1]{BKN1}
Let $L(\l)$ be a simple $\gl(m|n)$-module of atypicality $k$. Then
\[z_{\F}(L(\l))=2k.\]
\end{thm}

\begin{ex} \label{IntroEx3}
Let $\fg=\gl(1|1)$. The Kac module $K(0|0)$ admits a minimal projective resolution of the form
\[\ldots \rightarrow P(2|-2)\rightarrow  P(1|-1)\rightarrow P(0|0)\rightarrow K(0|0) \rightarrow 0.\]
The number of summands in each $P_d$ is $1$, thus the rate of growth of this number is $1$ and hence $z_{\F}(K(0|0))=1$.
\end{ex}

More generally,

\begin{thm}\cite[Theorem~9.1.1]{BKN1}
Let $K(\l)$ be a Kac $\gl(m|n)$-module of atypicality $k$. Then
\[z_{\F}(K(\l))=k.\]
\end{thm}

This invariant can be also interpreted geometrically, this time in terms of detecting subalgebras which were introduced in \cite{BKN4}:

\section{Detecting subsuperalgebras}
Boe, Kujawa, and Nakano \cite[Section~4]{BKN3} constructed two subsuperalgebras of $\fg$, namely $\ff$ and $\mathfrak{e}$, which will detect the relative cohomology ring of $\fg$ \cite[Theorem~4.1.1]{BKN3}. We will only use the detecting subsuperalgebra $\ff$ throughout this work. An explicit definition of $\ff_{\1}$ is given in \cite[Table~2]{BKN3} using root vectors. The even part is defined by $\ff_{\0}=[\ff_{\1},\ff_{\1}]$. Then
\[\ff:=\ff_{\0}\oplus \ff_{\1}.\]
Since $\fg$ is classical, the Lie superalgebra $\ff$ is also classical \cite[Theorem~4.1.1]{BKN3}. We can then define the support variety of $M$ relative to $\ff$, $\fv_{(\ff,\ff_{\0})}(M)$, in the same way we defined $\fv_{(\fg,\fg_{\0})}(M)$. If $[\ff_{\0},\ff_{\1}]=0$, these varieties admit a rank variety description and, in particular, can be identified as subvarieties of $\ff_{\1}$, i.e.,
\begin{align*}
\fv_{(\ff,\ff_{\0})}(M)&=\fv_{\ff_{\1}}^{rank}(M)
                       &=\{y\in \ff_{\1}\mid \text{M is not projective as $U(\langle y \rangle)$-module}\}\cup\{0\}.\\
\end{align*}
Note that $\fv_{(\ff,\ff_{\0})}(\C)=\ff_{\1}$. Boe et al. computed the dimensions of $\fv_{(\ff,\ff_{\0})}(M)$ when $M$ is either a simple or a Kac $\gl(m|n)$-module. They came to the conclusion:

\begin{thm}\cite[Theorem~9.2.1]{BKN1}
If $X(\lambda)$ is a Kac or a simple $\gl(m|n)$-module, then
\[z_{\F}(X(\lambda))=\dim \fv_{(\ff,\ff_{\0})}(X(\lambda)).\]
\end{thm}


\chapter{Computing the complexity over $\osp(2|2n)$}


\section{The Lie superalgebra $\osp(m|n)$}

We first discuss the orthosymplectic Lie superalgebra $\operatorname{osp}(m|n)$. Let $V=V_{\0}\oplus V_{\1}$ be a $\Z_2$-graded space, with $\dim V_{\0}=m$ and $\dim V_{\1}=n$. Let $F$ be a nondegenerate consistent supersymmetric bilinear form on $V$, so that $V_{\0}$ and $V_{\1}$ are orthogonal and the restriction of $F$ to $V_{\0}$ is symmetric and to $V_{\1}$ a skew-symmetric form. Define the Lie superalgebra $\gl(m|n):=\End(V)$ with the $\Z_2$-grading:
\[\End(V)_i=\{f\in \End V\mid f(V_s)\subseteq V_{i+s},\,s\in \Z_2\},\quad i\in \Z_2,\]
and the bracket is given by $[f,f']=ff'-(-1)^{\bar{f}\bar{f'}}f'f$. We define in $\gl(m|n)$ the subalgebra $\osp(m|n)=\osp(m|n)_{\0}\oplus \osp(m|n)_{\1}$ by setting
\[\osp(m|n)_i=\{f\in l(m|n)_i \mid F(f(x),y)=-(-1)^{i\bar{x}}F(x,f(y))\},\quad (i\in \Z_2).\]
\vspace{-2cm}
\section{The Lie superalgebra $\osp(2|2n)$}

In this chapter we will be working with the Lie superalgebra $\fg=\osp(2|2n)$ of type $C(n+1)$. The bilinear form $F$ can be represented by the matrix:
 \[
F=\left[\begin{array}{cc}
G & 0\\
0 & H\\
\end{array}\right],\]
where \[G=\left[\begin{array}{cc}
0 & 1\\
1 & 0\\
\end{array}\right]\quad\text{and}\quad
H= \left[\begin{array}{cc}
0 & I_n\\
-I_n & 0\\
\end{array}\right],
\]
where $I_n$ is the identity $n\times n$ matrix. Then a matrix $X=\begin{bmatrix} A & B\\ C & D \end{bmatrix}$ is in $\osp(2|2n)$ if and only if $$A^tG+GA=B^tG-HC=D^tH+HD=0.$$
Using this equation, we can see that $\fg$ has a $\Z$-grading
\[\fg=\fg_{-1}\oplus\fg_{0}\oplus\fg_{1},\]
given by the following matrix realization:
\[
\fg_{-1}=\left[\begin{array}{c|c}
\begin{array}{cc}
0 & 0\\
0 & 0\\
\end{array}  & \begin{array}{cc}
0 & 0\\
y & y_1\\
\end{array}\\ \hline
\begin{array}{cc}
y_{1}^t & 0\\
-y^t & 0\\
\end{array} & \begin{array}{cc}
0 & 0\\
0 & 0\\
\end{array}\\
\end{array}
\right],
\]

\[
\fg_{0}=\left[\begin{array}{c|c}
\begin{array}{cc}
\alpha & 0\\
0 & -\alpha\\
\end{array}  & \begin{array}{cc}
0 & 0\\
0 & 0\\
\end{array}\\ \hline
\begin{array}{cc}
0 & 0\\
0 & 0\\
\end{array} & \begin{array}{cc}
a & b\\
c & -a^t\\
\end{array}\\
\end{array}
\right],
\]
and
\[
\fg_{1}=\left[\begin{array}{c|c}
\begin{array}{cc}
0 & 0\\
0 & 0\\
\end{array}  & \begin{array}{cc}
x & x_1\\
0 & 0\\
\end{array}\\ \hline
\begin{array}{cc}
0 & x_{1}^t\\
0 & -x^t\\
\end{array} & \begin{array}{cc}
0 & 0\\
0 & 0\\
\end{array}\\
\end{array}
\right].
\]
In the above, $x,\,x_1,\,y,\,y_1$ are $1\times n$ matrices; $a,\,b,\,c$ are $n\times n$ matrices where $b$ and $c$ are symmetric; $\alpha$ is a scalar in $\C$. The $\Z_2$-grading is given by: $\fg_{\1}=\fg_{-1}\oplus \fg_{1}$ and $\fg_{\0}=\fg_0$. The Lie super-bracket is defined by
 \[[A,B]=AB-(-1)^{\bar{A}\bar{B}}BA,\]
 for homogeneous elements $A,\,B\in \fg_{\0}$ or $\fg_{\1}$. We then extend the definition of the bracket to all of $\fg$ by bilinearity. In the above $AB$ denotes the matrix multiplication of $A$ and $B$. The degree $\bar{A}$ is
 \[\bar{A}=\begin{cases}
          0\quad \text{if $A\in \fg_{\0},$}\\
          1\quad \text{if $A\in \fg_{\1}.$}
          \end{cases}\]
 Using this bracket, we can easily check that $\fg_{-1}$ and $\fg_1$ are both abelian.

Note that $\fg_{\0}\cong \C\oplus \mathfrak{sp}(2n)$, hence the choice of the Cartan subalgebra:
\[
\fh=\left[\begin{array}{c|c}
\begin{array}{cc}
\alpha & 0\\
0 & -\alpha\\
\end{array}  & \begin{array}{cc}
0 & 0\\
0 & 0\\
\end{array}\\ \hline
\begin{array}{cc}
0 & 0\\
0 & 0\\
\end{array} & \begin{array}{cc}
a & 0\\
0 & -a\\
\end{array}\\
\end{array}
\right],
\]
where $a$ is $n\times n$ diagonal matrix. Let $\varepsilon_1:\fh\rightarrow \C$ be the linear map that takes an element of $\fh$ to its first diagonal entry ($\alpha$) and let $\delta_i:\fh\rightarrow \C$ be the linear map that takes an element of $\fh$ to the $i$th diagonal entry of the matrix $a$. The set $\{\varepsilon_1,\,\delta_1,\,\delta_2,\ldots,\delta_n \}$ forms a basis of $\fh^*$ and we define a bilinear form, $(\,,\,)$, on $\fh^*$ by setting
\[
(\varepsilon_1,\varepsilon_1)=1,\quad (\delta_i,\delta_j)=-\delta_{ij},\quad (\varepsilon_1,\delta_i)=0,
\]
for all $1\leq i \leq n$.
The Borel subalgebra for $\fg_{\0}$ is chosen as follows:
\[
\fb_{\0}=\left[\begin{array}{c|c}
\begin{array}{cc}
\alpha & 0\\
0 & -\alpha\\
\end{array}  & \begin{array}{cc}
0 & 0\\
0 & 0\\
\end{array}\\ \hline
\begin{array}{cc}
0 & 0\\
0 & 0\\
\end{array} & \begin{array}{cc}
a & b\\
0 & -a^t\\
\end{array}\\
\end{array}
\right],
\]
where $a$ is upper-triangular and $b$ is symmetric. With our choice of the above Cartan subalgebra the root system of $\fg_{\0}$, that is the even roots of $\fg$, is
\[
\Phi_{\0}=\{\pm\delta_i\pm\delta_j\mid 1\leq i\neq j\leq n\}\cup\{\pm2\delta_i\mid 1\leq i \leq n\}.
\]
The set of odd roots is
\[
\Phi_{\1}=\{\pm\varepsilon_1\pm\delta_i\mid 1\leq i \leq n\}.
\]
The Borel superalgebra for $\fg$ will be given by $\fb=\fb_{\0}\oplus\fg_1$. With this choice, the set of positive roots of $\fg$ is given by
\[
\Phi^{+}=\{\delta_i+\delta_j\mid 1\leq i\leq j\leq n\}\cup\{\delta_i-\delta_j\mid 1\leq i<j \leq n\}\cup\{\varepsilon_1\pm\delta_i\mid 1\leq i \leq n\},
\]
and the set of simple roots is
\[
\Delta=\{\delta_i-\delta_{i+1}\mid 1\leq i\leq n\}\cup\{2\delta_n,\,\varepsilon_1-\delta_1\}.
\]

Let $\x\subseteq\fh^*$ be the parameterizing set of highest weights for the simple finite dimensional $\fg_{\0}$-modules with respect to the pair $(\fh,\,\fb_{\0})$. By \cite[Section~21.1]{HUM}, we can show that the simple finite-dimensional $\mathfrak{sp}(2n)$-modules have integral weights. An explicit description of $\x$ is
\[\x=\{\lambda=\lambda_{-1}\varepsilon_1+\sum_{i=1}^{n}\lambda_i\delta_i \mid \l_{-1}\in \C,\, \lambda_i\in \Z,\, \forall i\geq 1;\,\,\lambda_1\geq \lambda_2\geq \ldots \geq \lambda_n\geq 0\}.\]

For $\lambda\in \x$, let $L_{\0}(\lambda)$ be the simple finite dimensional $\fg_{\0}$-module of highest weight $\lambda$. Since $\osp(2|2n)_{\0}\cong \C\oplus \mathfrak{sp}(2n)$, the simple $\fg_{\0}$-modules are of the form
\[L_{\0}(\lambda)=\C_{\lambda_{-1}}\boxtimes L_0(\lambda),\]
 where $L_0(\lambda)$ is the simple $\mathfrak{sp}(2n)$-module of weight $\displaystyle{\sum_{i=1}^{n}\lambda_i\delta_i}$. Note that there is a slight abuse of notation where we used $\l$ for the simple $\mathfrak{sp}(2n)$-module even though we removed the $\l_{-1}$-part. The action of $\fg_{\0}$ on $L_{\0}(\lambda)=\C_{\lambda_{-1}}\boxtimes L_0(\lambda)$ is given by
 \[(\alpha\oplus T).(c_{\lambda_{-1}}\otimes v)=\alpha.c_{\lambda_{-1}}\otimes v+c_{\lambda_{-1}}\otimes T.v,\]
  for $\alpha,\, c_{\lambda_{-1}} \in \C,\, T\in \mathfrak{sp}(2n)$ and $v\in L_0(\lambda)$. It is worth mentioning that if $v_1$ is the highest weight vector for $L_0(\lambda)$, then $w_1=1\otimes v_1$ is the highest weight vector in $L_{\0}(\lambda)$.

Let $\rho$ be half the sum of the positive even roots minus half the sum of the positive odd roots. Then
\[\rho=-n\varepsilon_1+\sum_{i=1}^{n}(n-i+1)\delta_i.\]
%Note that if $\lambda\in \x$ and $\lambda+\rho=a_1\varepsilon_1+\sum_{i=1}^{n}b_i\delta_i$, then $a_1\in \Z$ and $b_1>b_2>\cdots>b_n\geq1$.
%The isotropic roots are the odd ones but they are not pairwise orthogonal, hence $\atyp(\lambda)$ is either one or zero. In fact, $\atyp(\lambda)=1$ if there is an odd positive root $\varepsilon_1\pm\delta_i$ such that $(\lambda+\rho,\varepsilon_1\pm\delta_i)=0$ for some $1\leq i \leq n$ which is equivalent to having $a_1=\pm b_i$ in $\lambda+\rho$ for some $1\leq i \leq n$. If $L(\lambda)$ is a simple $\fg$-module of highest weight $\lambda$, then we define $\atyp(L(\lambda)):=\atyp(\lambda)$. It is known that the atypicality of a simple module is independent of the choice of Cartan and Borel subalgebras and, furthermore, is the same for all simple modules in a given block. Hence it makes sense to refer to the atypicality of a block. If $\atyp(\lambda)=0$, then $P(\lambda)=L(\lambda)=K(\lambda)$ hence $L(\lambda)$ and $K(\lambda)$ are projective.
We can easily check that the atypicality of $\l$ is either zero or one. In fact, $\atyp(\lambda)=1$ if and only if there is an odd positive root $\varepsilon_1\pm\delta_i$ such that $(\lambda+\rho,\varepsilon_1\pm\delta_i)=0$ for some $1\leq i \leq n$. A weight is called \emph{atypical} if its atypicality is equal to 1, and is called \emph{typical} otherwise. It is important to note that if $\l=\lambda_{-1}\varepsilon_1+\sum_{i=1}^{n}\lambda_i\delta_i$ is atypical, then $\l_{-1}\in  \Z$.

If $L(\lambda)$ is a simple $\fg$-module of highest weight $\lambda$, then we define $\atyp(L(\lambda)):=\atyp(\lambda)$. It is known that the atypicality of a simple module is independent of the choice of Cartan and Borel subalgebras and, furthermore, is the same for all simple modules in a given block. Hence it makes sense to refer to the atypicality of a block. It is known that (\cite[Theorem~1]{Kac1}) if $\atyp(\lambda)=0$, then $P(\lambda)=L(\lambda)=K(\lambda)$ hence $L(\lambda)$ and $K(\lambda)$ are projective.

\section{Computing the complexity}\label{s:computecomplexity}
For $\lambda\in \x$, we want to find the complexity of $L(\lambda)$ and $K(\l)$. If $\l$ is typical, the simple module $L(\l)$ is projective and hence it has zero complexity. We only need to consider the case when $\lambda$ is atypical since the complexity of the typical simple and Kac modules is zero. In this section, we will refer to \cite[Sections~1.2,\,3.1]{CWZ} to get a description of the projective covers.

Let $W$ be the Weyl group of $\fg$ which is, by definition, the Weyl group of the $\mathfrak{sp}(2n)$-subalgebra, generated by the reflections corresponding to the even simple roots of $\fg$. If $\lambda\in \x$ is atypical with respect to an odd positive root $\gamma$, the authors in \cite{CWZ} defined an ``L-operator" given by:
\[\lambda^L:=\omega(\lambda+\rho-k\gamma)-\rho,\]
where $k$ is the smallest positive integer such that $(\lambda+\rho-k\gamma,\alpha_i)\neq 0$ for all the even simple roots $\alpha_i$ and $\omega$ is the unique element in the Weyl group of $\mathfrak{sp}(2n)$ rendering $\lambda^L$ dominant. Given an atypical $\lambda\in \x$, we shall write
\begin{equation}\label{eq1:inductivepower}
\lambda^{(0)}=\lambda,\quad \lambda^{(l+1)}=(\lambda^{(l)})^L,\quad l\geq 0.
\end{equation}
Let $\omega_0$ be the longest element in the Weyl group of $\mathfrak{sp}(2n)$ and let $\beta=2n\varepsilon_1$ be the sum of all positive roots. Let $\lambda$ be atypical, we can use \cite[equation~(4)]{CWZ} to get:
\[\lambda=\omega_0^{-1}\Big(\beta-(\beta-\omega_0\lambda^L)^L\Big),\]
which proves that the $L$-operator is one-to-one. We will prove that $L$ is onto on the set of atypical weights in $\x$ using a representation theoretical approach (Proposition~\ref{Lsurjective}). Using this we can define $\lambda^{-L}$ to be the unique weight $\mu$ such that $\mu^L=\lambda$, hence we can extend the definition in \eqref{eq1:inductivepower} to any $l\in \Z$.

 Consider the set $Y_+^{1|n}$ of $(n+1)$-tuples of integers:
\[Y_+^{1|n}=\{f=(f_{-1}|f_1,f_2,\ldots,f_n)\mid f_i\in \Z,\,\, f_1<f_2<\cdots<f_n<0\}.\]
There is a bijection $\x\rightarrow Y_+^{1|n}$ given by $\lambda\mapsto f_\lambda$, where
\[(f_\lambda)_{-1}=(\lambda+\rho,\varepsilon_1),\quad (f_\lambda)_i=(\lambda+\rho,\delta_i),\,i\geq 1.\]
Since $\l$ is atypical, we have $(f_\lambda)_{-1}\in \Z$ and $|f_{-1}|=-f_i$ for some $i\geq 1$. We set $f^L:=f_{\lambda^L}$. Note that we can compute $\lambda^L$ for each atypical $\lambda$ by passing over to the set $Y_+^{1|n}$. The description of $f^L$ is divided into three cases as follows (cf. \cite[Section~1.2]{CWZ}:

\begin{enumerate}
\item [(I)] $f_{-1}=f_i<0$ for some $1\leq i\leq n$. let $c$ be the largest integer such that $c<f_i$ and $d\notin \{f_1,f_2,\ldots,f_n\}$. Then
\[f^L=(c|f_1,\ldots \widehat{f_i},\ldots,f_n,c)^+,\]
 where $\widehat{f_i}$ denotes the removal of $f_i$, and the $+$ denotes the rearrangement of $$f_1,\ldots \widehat{f_i},\ldots,f_n,c$$ in an increasing order.
\item [(II)] $f_{-1}=-f_i>0$ for some $1\leq i\leq n$ and $\{f_1,f_2,\ldots,f_n\}$ does not contain $\{-1,-2,\ldots,f_i\}$ as a subset. Let $c$ be the largest integer such that $f_i<c\leq -1$ and $-c\notin \{f_1,f_2,\ldots,f_n\}$. Then
\[f^L=(c|f_1,\ldots \widehat{f_i},\ldots,f_n,-c)^+.\]
\item [(III)] $f_{-1}=-f_i>0$ for some $1\leq i\leq n$ and $\{f_1,f_2,\ldots,f_n\}$ contains $\{-1,-2,\ldots,f_i\}$ as a subset. Then
\[f^L=(-f_{-1}|f_1,\ldots, f_n).\]
\end{enumerate}
Using these cases, we can show the following:

\begin{lem}\label{t1:projcover(2,2n)}
We have:
\begin{enumerate}
\item For $d\geq 0$, $(-d|d,0,\ldots,0)^L=(-d-1|d+1,0,\dots,0)$.
\item For $d\geq 1$, $(2n+d|d,0,\ldots,0)^L=(2n+(d-1)|d-1,0,\ldots,0)$.
\item  $(2n|0,\dots,0)^L=(0|0,\ldots,0)$.
\item For $\l=(0|0,\ldots,0)=0$ and $d\geq 0$, we have
\[\l^{(d)}=(-d|d,0,\ldots,0)\,\, \text{and}\,\, \l^{(-d-1)}=(2n+d|d,0,\ldots,0).\]
\end{enumerate}
\end{lem}

\begin{proof}
For $(1)$, $f_{\lambda}=(-n-d|-(n+d),-(n-1),-(n-2),\ldots,-2,-1)$, hence $f_{-1}=f_{1}<0$, and case $(I)$ applies. The required integer $c$ is $-n-d-1$, then
\[f^L_{\lambda}=f_{\lambda^L}=(-n-d-1|-n-d-1,-(n-1),-(n-2),\ldots,-2,-1),\]
which gives $\lambda^L=(-d-1|d+1,0,\ldots,0)$.

For $(2)$, $f_{\lambda}=(n+d|-(n+d),-(n-1),-(n-2),\ldots,-2,-1)$, hence $f_{-1}=-f_{1}>0$, and case $(II)$ applies. The required integer $c$ is $n+d-1$, then
\[f^L_{\lambda}=f_{\lambda^L}=(n+d-1|-n-d+1,-(n-1),-(n-2),\ldots,-2,-1),\]
which gives $\lambda^L=(2n+d-1|d-1,0,\ldots,0)$.

For $(3)$, $f_{\lambda}=(n|-n,-(n-1),-(n-2),\ldots,-2,-1)$, hence $f_{-1}=-f_{1}>0$, and case $(III)$ applies. Then
\[f^L_{\lambda}=f_{\lambda^L}=(-n|-n,-(n-1),-(n-2),\ldots,-2,-1),\]
which gives $\lambda^L=(0|0,0,\ldots,0)$.

The last part of the Lemma follows easily from the earlier parts.
\end{proof}

Let $P(\lambda)$ be the projective cover of $L(\lambda)$. From \cite[Theorem~7]{CWZ}, we have the following 2-step Kac flag:
\[0\rightarrow K(\lambda)\rightarrow P(\lambda^L)\rightarrow K(\lambda^L)\rightarrow 0,\]
which implies that $\dim P(\lambda^L)=\dim K(\lambda)+\dim K(\lambda^L)$. Moreover, using \cite[Corollary~8]{CWZ}, there is a short exact sequence:
\[0\rightarrow L(\lambda^L)\rightarrow K(\lambda)\rightarrow L(\lambda) \rightarrow 0.\]
This shows that the Kac module has two layers: The head of $K(\lambda)$ is $L(\lambda)$ and the socle is $L(\lambda^L)$. On the other hand, the projective module $P(\lambda^L)$ decomposes as follows: the head and socle of $P(\lambda^L)$ are both isomorphic to $L(\lambda^L)$ and the second layer is isomorphic to $L(\lambda)\oplus L(\lambda^{LL})$. It has the form:

\[\begin{tikzpicture}

  \node (n1) at (1,1) {$L(\l^L)$};
  \node (n2) at (0,0)  {$L(\l)$};
  \node (n3) at (2,0)  {$L(\l^{LL})$};
  \node (n4) at (1,-1) {$L(\l^L)$};


  \foreach \from/\to in {n1/n2,n1/n3,n2/n4,n3/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}\]

%\[ P(\lambda^L)=\begin{array}{c}
% L(\l^L)\\

%  L(\l)\ \ \ L(\l^{LL}) \\

% L(\l^L)
% \end{array}. \]
Note that by the Kac filtration of $P(\l^L)$ and the composition factors of $K(\l)$ we know that the socle of $P(\l^L)$ has only one submodule, namely $L(\l^L)$. Similarly the head of $P(\l^L)$ is the head of $K(\l^L)$ which is $L(\l^L)$. We can also see this from the fact that the $P(\l^L)$ is the projective cover and injective hull of $L(\l^L)$, $L(\l^L)$ is the unique head and socle at the same time.  We will prove later in \eqref{eq4:dimExt} that
\[\dim \Ext^1(L(\l),L(\mu))=1\Leftrightarrow \l=\mu^L\,\,\text{or}\,\, \l^L=\mu.\]
Thus there is no indecomposable module $M$ such that the following sequence is exact:
 \[0\rightarrow L(\lambda)\rightarrow M\rightarrow L(\lambda^{LL}) \rightarrow 0.\]
This shows that the middle layer can not be of the form
\[\begin{tikzpicture}

  \node (n1) at (1,1) {$L(\l^{LL})$};
  \node (n2) at (1,0)  {$L(\l)$};
  \foreach \from/\to in {n1/n2}
    \draw (\from) -- (\to);

\end{tikzpicture}\]

%\[\begin{array}{cc}
                       % L(\l^{LL})\\
                       % L(\l)
                        %\end{array}.\]
\subsection{Complexity of simple modules}\label{SS:simplecomplexity}
To compute the complexity of the simple modules $L(\l)$, we need the following bounds on the dimension of the simple $\mathfrak{sp}(2n)$-module $L_0(r,0,\ldots,0)$:
\begin{lem}\label{lem:bddimsimple}
There are positive constants $C$ and $C'$ that depend only on $n$ such that
\[Cr^{2n-1}\leq \dim L_0(r,0,\ldots,0)\leq C'r^{2n-1}.\]
\end{lem}
\begin{proof}
 The dimension of a simple module of highest weight $\lambda$ is given by the Weyl-dimension formula (\cite[Section~24.3]{HUM}):
\[\dim L_0(\lambda)=\frac{\prod_{\alpha\in \Phi_{\0}^+}(\lambda+\delta,\alpha)}{\prod_{\alpha\in \Phi_{\0}^+}(\delta,\alpha)},\]
where $\delta$ is half the sum of the positive roots in $\mathfrak{sp}(2n)$, i.e., half the sum of the positive even roots in $\osp(2|2n)$.
\\
Recall
\[
\Phi_{\0}^{+}=\{\delta_i+\delta_j\mid 1\leq i\leq j\leq n\}\cup\{\delta_i-\delta_j\mid 1\leq i<j \leq n\},
\]
then $\delta=\displaystyle{\sum_{i=1}^{n}(n-i+1)\delta_i}$. Thus,

\[\dim L_0(r,0,\ldots,0)=\frac{(2n+r)\prod_{j=2}^n(r+j-1)(2n+r-j+1)}{(2n)\prod_{j=2}^n(j-1)(2n-j+2)}.\]
Let
\[
C=\frac{1}{(2n)\prod_{j=2}^n (j-1)(2n-j+2)}.
\]
Note that $C$ is a positive constant depending only on $n$ and $\dim L_0(r,0,\ldots,0)$ is a polynomial in $r$ of degree $2n-1$ with a positive leading coefficient. Moreover,
\iffalse
\begin{align*}
\dim L_0(r,0,\ldots,0)&=C(2n+2r)\prod_{j=2}^n(r+j-1)(2n+r-j+1)\\
                      &\geq C(2n+2r)(r+1)^{n-1}(n+r+1)^{n-1}\\
                      &\geq Cr^{2n-1}.
\end{align*}
On the other hand,
\begin{align*}
\dim L_0(r,0,\ldots,0)&=C(2n+2r)\prod_{j=2}^n(r+j-1)(2n+r-j+1)\\
                      &\leq C(2n+2r)(r+n-1)^{n-1}(2n+r-1)^{n-1}\quad \text{($2\leq j\leq n$).}\\
\end{align*}
\fi
\begin{align*}
\dim L_0(r,0,\ldots,0)&=C_1(2n+2r)\prod_{j=2}^n(r+j-1)(2n+r-j+1)\\
                      &\geq C_1(2n+2r)(r+1)^{n-1}(2n+r-n+1)^{n-1}\\
                      &=C_1(2n+2r)(r+1)^{n-1}(n+r+1)^{n-1}\\
                      &\geq C_1r^{2n-1}.
\end{align*}
On the other hand,
\begin{align*}
\dim L_0(r,0,\ldots,0)&=C_1(2n+2r)\prod_{j=2}^n(r+j-1)(2n+r-j+1)\\
                      &\leq C_1(2n+2r)(r+n-1)^{n-1}(2n+r-2+1)^{n-1}\\
                      &=C_1(2n+2r)(r+n-1)^{n-1}(2n+r-1)^{n-1},
\end{align*}
Let us pick positive constants $C_2,C_3,C_4$ depending only on $n$ such that
\[2n+2r\leq C_2r,\quad r+n-1\leq C_3r,\quad 2n+r-1\leq C_4r,\]
then
\[\dim L_0(r,0,\ldots,0)\leq C.C_2.{C_3}^{n-1}.{C_4}^{n-1}r^{2n-1}=C'r^{2n-1},\]
where $C'$ is a positive constant that depends only on $n$. Therefore, there are positive constants $C$ and $C'$ that depend only on $n$ such that
\[Cr^{2n-1}\leq \dim L_0(r,0,\ldots,0)\leq C'r^{2n-1}.\]
\end{proof}
Recall that
\[0^{(d)}=(-d|d,0,\ldots,0) \quad \text{and}\quad 0^{(-d-1)}=(2n+d|d,0,\ldots,0).\]
For $i\in \Z$, set
$$[i]=L(0^{(i)})\quad P^{(i)}=P(0^{(i)}),\,\,\text{and}\,\, K^{(i)}=K(0^{(i)}).$$
Using these notations, the projective covers $P^{(i)}$ have the following radical layer structure:
\[\begin{tikzpicture}

  \node (n1) at (1,1) {$[i]$};
  \node (n2) at (0,0)  {$[i-1]$};
  \node (n3) at (2,0)  {$[i+1]$};
  \node (n4) at (1,-1) {$[i]$};


  \foreach \from/\to in {n1/n2,n1/n3,n2/n4,n3/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}\]

\begin{lem}\label{firsttermsproj}
Let
\[P_0=P(0)=\raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[-1]$};
  \node (n2) at (0,0)  {$[-2]$};
  \node (n3) at (2,0)  {$[0]$};
  \node (n4) at (1,-1) {$[-1]$};


  \foreach \from/\to in {n1/n2,n1/n3,n2/n4,n3/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}} \]

\[P_1=P^{(-1)}\oplus P^{(1)}\raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[-1]$};
  \node (n2) at (0,0)  {$[-2]$};
  \node (n3) at (2,0)  {$[0]$};
  \node (n4) at (1,-1) {$[-1]$};


  \foreach \from/\to in {n1/n2,n1/n3,n2/n4,n3/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\oplus \raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[1]$};
  \node (n2) at (0,0)  {$[0]$};
  \node (n3) at (2,0)  {$[2]$};
  \node (n4) at (1,-1) {$[1]$};


  \foreach \from/\to in {n1/n2,n1/n3,n2/n4,n3/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\]

\[P_2=P^{(-2)}\oplus P(0)\oplus P^{(2)}=\raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[-2]$};
  \node (n2) at (0,0)  {$[-3]$};
  \node (n3) at (2,0)  {$[-1]$};
  \node (n4) at (1,-1) {$[-2]$};


  \foreach \from/\to in {n1/n2,n1/n3,n2/n4,n3/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\oplus \raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[0]$};
  \node (n2) at (0,0)  {$[-1]$};
  \node (n3) at (2,0)  {$[1]$};
  \node (n4) at (1,-1) {$[0]$};


  \foreach \from/\to in {n1/n2,n1/n3,n2/n4,n3/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\oplus \raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[2]$};
  \node (n2) at (0,0)  {$[1]$};
  \node (n3) at (2,0)  {$[3]$};
  \node (n4) at (1,-1) {$[2]$};


  \foreach \from/\to in {n1/n2,n1/n3,n2/n4,n3/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\]

The first three terms of a minimal projective resolution of $L(0)$ are:
\[P_2 \stackrel{f_2}{\rightarrow} P_1 \stackrel{f_1}{\rightarrow}P_0 \stackrel{f_0}{\rightarrow} [0]\rightarrow 0\]
\end{lem}
\begin{proof}
To prove this result we use the diagrammatic method for modular representations given in \cite{BC}. In particular, we use the description given in \cite[Section~10.3]{BC} for the kernel of the surjective map $P(M) \twoheadrightarrow M$ where $P(M)$ is the projective cover of $M\in \F$. We also use the fact that $P(M)$ is the direct sum of the projective covers of the simple modules in the head of $M$.

Let $f_0:P(0)\twoheadrightarrow L(0)=[0]$ be the projection map from $P(0)$ onto its head. Let $Ker(f_0)$ be the kernel of $f_0$. Sine the map $f_0$ projects the head of $P(0)$ onto $L(0)$, we have $P(0)/Ker(f_0)\cong [0]$. Thus the kernel of $f_0$ is the submodule of $P(0)$ with the layer structure:
 \[Ker(f_0)=\raisebox{-3em}{\begin{tikzpicture}

  %\node (n1) at (1,1) {$L(\l^L)$};
  \node (n2) at (0,0)  {$[-1]$};
  \node (n3) at (2,0)  {$[1]$};
  \node (n4) at (1,-1) {$[0]$};


  \foreach \from/\to in {n2/n4,n3/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\]
 By taking a quotient map we have
 \[\raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[-1]$};
  \node (n2) at (0,0)  {$[-2]$};
  \node (n3) at (2,0)  {$[0]$};
  \node (n4) at (1,-1) {$[-1]$};


  \foreach \from/\to in {n1/n2,n1/n3,n2/n4,n3/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\oplus \raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[1]$};
  \node (n2) at (0,0)  {$[0]$};
  \node (n3) at (2,0)  {$[2]$};
  \node (n4) at (1,-1) {$[1]$};


  \foreach \from/\to in {n1/n2,n1/n3,n2/n4,n3/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}} \twoheadrightarrow \raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[-1]$};
 % \node (n2) at (0,0)  {$[-1]$};
  %\node (n3) at (2,0)  {$[1]$};
  \node (n4) at (1,-1) {$[0]$};


  \foreach \from/\to in {n1/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\oplus \raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[1]$};
  %\node (n2) at (0,0)  {$[1]$};
  %\node (n3) at (2,0)  {$[1]$};
  \node (n4) at (1,-1) {$[0]$};


  \foreach \from/\to in {n1/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\stackrel{\phi_1}{\twoheadrightarrow} \raisebox{-3em}{\begin{tikzpicture}

  %\node (n1) at (1,1) {$L(\l^L)$};
  \node (n2) at (0,1)  {$[-1]$};
  \node (n3) at (2,1)  {$[1]$};
  \node (n4) at (1,-1) {$[0]$};


  \foreach \from/\to in {n2/n4,n3/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\hookrightarrow \raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[0]$};
  \node (n2) at (0,0)  {$[-1]$};
  \node (n3) at (2,0)  {$[1]$};
  \node (n4) at (1,-1) {$[0]$};


  \foreach \from/\to in {n1/n2,n1/n3,n2/n4,n3/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\]
 The map $\phi_1$ can be chosen to satisfy $\phi_1(x,y)=x-y$ for $x\in [0]$ in the first summand and $y\in [0]$ in the second. Define $f_1$ to be the composition of these maps. Note that to find the projective cover of a module $M$, we take the direct sum of the projective covers of the simple modules which appear in the top level of $M$. Thus $P_1$ is the projective cover of
 \[ \raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[-1]$};
 % \node (n2) at (0,0)  {$[-1]$};
  %\node (n3) at (2,0)  {$[1]$};
  \node (n4) at (1,-1) {$[0]$};


  \foreach \from/\to in {n1/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\oplus \raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[1]$};
  %\node (n2) at (0,0)  {$[1]$};
  %\node (n3) at (2,0)  {$[1]$};
  \node (n4) at (1,-1) {$[0]$};


  \foreach \from/\to in {n1/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\]
 This means that $P_1$ is the minimal projective module that we can choose to make the sequence exact. The Kernel of $f_1$ is

\[Ker(f_1)=\raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[-2]$};
  \node (n2) at (2,0)  {$[-1]$};
  \node (n3) at (3,1)  {$[0]$};
  \node (n4) at (4,0) {$[1]$};
  \node (n5) at (5,1) {$[2]$};

  \foreach \from/\to in {n1/n2, n2/n3, n3/n4, n4/n5}
    \draw (\from) -- (\to);

\end{tikzpicture}}\]

We again take a quotient map to get
\[P_2\twoheadrightarrow  \raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[-2]$};
  %\node (n2) at (0,0)  {$[1]$};
  %\node (n3) at (2,0)  {$[1]$};
  \node (n4) at (1,0) {$[-1]$};


  \foreach \from/\to in {n1/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}} \oplus \raisebox{-3em}{\begin{tikzpicture}

  %\node (n1) at (1,1) {$L(\l^L)$};
  \node (n2) at (1,1)  {$[0]$};
  \node (n3) at (0,0)  {$[-1]$};
  \node (n4) at (2,0) {$[1]$};


  \foreach \from/\to in {n2/n3,n2/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\oplus \raisebox{-3em}{\begin{tikzpicture}
  \node (n1) at (1,1) {$[2]$};
  %\node (n2) at (0,0)  {$[1]$};
  %\node (n3) at (2,0)  {$[1]$};
  \node (n4) at (1,0) {$[1]$};


  \foreach \from/\to in {n1/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\stackrel{\phi_2}{\twoheadrightarrow} \raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[-2]$};
  \node (n2) at (2,0)  {$[-1]$};
  \node (n3) at (3,1)  {$[0]$};
  \node (n4) at (4,0) {$[1]$};
  \node (n5) at (5,1) {$[2]$};

  \foreach \from/\to in {n1/n2, n2/n3, n3/n4, n4/n5}
    \draw (\from) -- (\to);

\end{tikzpicture}}\hookrightarrow P_1\]
  The map $\phi_2$ can be defined in a similar fashion as $\phi_1$. Define $f_2$ to be the composition of these maps. As before, $P_2$ is the minimal projective module we could choose to make the sequence exact. This is the case because it is the projective cover of
  \[\raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[-2]$};
  %\node (n2) at (0,0)  {$[1]$};
  %\node (n3) at (2,0)  {$[1]$};
  \node (n4) at (1,0) {$[-1]$};


  \foreach \from/\to in {n1/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}} \oplus \raisebox{-3em}{\begin{tikzpicture}

  %\node (n1) at (1,1) {$L(\l^L)$};
  \node (n2) at (1,1)  {$[0]$};
  \node (n3) at (0,0)  {$[-1]$};
  \node (n4) at (2,0) {$[1]$};


  \foreach \from/\to in {n2/n3,n2/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\oplus \raisebox{-3em}{\begin{tikzpicture}
  \node (n1) at (1,1) {$[2]$};
  %\node (n2) at (0,0)  {$[1]$};
  %\node (n3) at (2,0)  {$[1]$};
  \node (n4) at (1,0) {$[1]$};


  \foreach \from/\to in {n1/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\]
The kernel of $f_2$ is
\[Ker(f_2)=\raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[-3]$};
  \node (n2) at (2,0)  {$[-2]$};
  \node (n3) at (3,1)  {$[-1]$};
  \node (n4) at (4,0) {$[0]$};
  \node (n5) at (5,1) {$[1]$};
  \node (n6) at (6,0) {$[2]$};
  \node (n7) at (7,1) {$[3]$};

  \foreach \from/\to in {n1/n2, n2/n3, n3/n4, n4/n5, n5/n6, n6/n7}
    \draw (\from) -- (\to);

\end{tikzpicture}}\]

 The exactness of the sequence follows from the construction of the maps $f_i$.
\end{proof}


\begin{lem}
The minimal projective resolution of $L(0)$ is
\begin{equation}\label{projrestrivial}
 \ldots P_d \stackrel{f_d}{\rightarrow} \ldots \stackrel{f_1}{\rightarrow} P(0)\stackrel{f_0}{\rightarrow} L(0) \rightarrow 0,
\end{equation}
where the $d^{th}$ term in this resolution is given by: If $d$ is even,

%\begin{align*}
%P_d=& P(2n+d-1|d-1,0,\ldots,0)\oplus P(2n+d-3|d-3,\ldots,0)\\
 %  &\oplus \cdots\oplus P(2n+1|1,\ldots,0) \oplus P(0|0,\ldots,0)\\
  %  & \oplus P(-d|d,0,\ldots,0)\oplus P(-d+2|d-2,\ldots,0)\oplus \cdots \oplus P(-2|2,\ldots,0) ,
%\end{align*}
\begin{align*}
P_d=P^{(-d)}\oplus P^{(-d+2)}\oplus\cdots \oplus P^{(-2)}\oplus P^{(0)}\oplus P^{(d)}\oplus P^{(d-2)}\oplus\cdots \oplus P^{(2)},
\end{align*}
  and if $d$ is odd,
%\begin{align*}
%P_d=& P(2n+d-1|d-1,0,\ldots,0)\oplus P(2n+d-3|d-3,\ldots,0)\oplus \cdots \oplus P(2n|0,\ldots,0)\\
 %   & \oplus P(-d|d,0,\ldots,0)\oplus P(-d+2|d-2,\ldots,0)\oplus \cdots \oplus P(-1|1,\ldots,0).
%\end{align*}
\begin{align*}
P_d=P^{(-d)}\oplus P^{(-d+2)}\oplus\cdots \oplus P^{(-1)}\oplus P^{(d)}\oplus P^{(d-2)}\oplus\cdots \oplus P^{(1)}.
\end{align*}
The kernel of $f_d$ is
\[Ker(f_d)=\raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[-d-1]$};
  \node (n2) at (2,0)  {$[-d]$};
  \node (n3) at (3,1)  {$[-d+1]$};
  \node (n4) at (4,0)  {$\cdots$};
  \node (n5) at (5,1) {$[-1]$};
  \node (n6) at (6,0)  {$[0]$};
  \node (n7) at (7,1)  {$[1]$};
  \node (n8) at (8,0)  {$\cdots$};
  \node (n9) at (9,1) {$[d-1]$};
  \node (n10) at (10,0)  {$[d]$};
  \node (n11) at (11,1)  {$[d+1]$};

  \foreach \from/\to in {n1/n2, n2/n3, n3/n4, n4/n5, n5/n6, n6/n7, n7/n8, n8/n9, n9/n10, n10/n11}
    \draw (\from) -- (\to);

\end{tikzpicture}}\]

\end{lem}


\begin{proof}
An inductive argument proves the result. The base case is proved in lemma~\ref{firsttermsproj}. We again use the diagrammatic method for modular representations given in \cite{BC}. In particular, we use the description given in \cite[Section~10.3]{BC} for the kernel of the surjective map $P(M) \twoheadrightarrow M$ where $P(M)$ is the projective cover of $M\in \F$. We also use the fact that $P(M)$ is the direct sum of the projective covers of the simple modules in the head of $M$.
\end{proof}


\begin{thm}\label{t2:atypcomplexitysimple}
For atypical $\lambda\in \x$, $c_{\F}(L(\lambda))=2n+1$.
\end{thm}

\begin{proof}
\iffalse
First, we will find $c_{\F}(L(0|0,\ldots,0))$. Recall that
\[0^{(d)}=(-d|d,0,\ldots,0) \quad \text{and}\quad 0^{(-d-1)}=(2n+d|d,0,\ldots,0).\]
Set $P^{(i)}=P(0^{(i)})$ and $K^{(i)}=K(0^{(i)})$. Using these notations, the projective covers $P^{(i)}$ have the following radical layer structure:
\[\begin{tikzpicture}

  \node (n1) at (1,1) {$[i]$};
  \node (n2) at (0,0)  {$[i-1]$};
  \node (n3) at (2,0)  {$[i+1]$};
  \node (n4) at (1,-1) {$[i]$};


  \foreach \from/\to in {n1/n2,n1/n3,n2/n4,n3/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}\]
The minimal projective resolution of $L(0)$ is
\begin{equation}\label{projrestrivial}
 \ldots P_d \stackrel{f_d}{\rightarrow} \ldots \stackrel{f_1}{\rightarrow} P(0)\stackrel{f_0}{\rightarrow} L(0) \rightarrow 0,
\end{equation}
where the $d^{th}$ term in this resolution is given by: If $d$ is even,

%\begin{align*}
%P_d=& P(2n+d-1|d-1,0,\ldots,0)\oplus P(2n+d-3|d-3,\ldots,0)\\
 %  &\oplus \cdots\oplus P(2n+1|1,\ldots,0) \oplus P(0|0,\ldots,0)\\
  %  & \oplus P(-d|d,0,\ldots,0)\oplus P(-d+2|d-2,\ldots,0)\oplus \cdots \oplus P(-2|2,\ldots,0) ,
%\end{align*}
\begin{align*}
P_d=P^{(-d)}\oplus P^{(-d+2)}\oplus\cdots \oplus P^{(-2)}\oplus P^{(0)}\oplus P^{(d)}\oplus P^{(d-2)}\oplus\cdots \oplus P^{(2)},
\end{align*}
  and if $d$ is odd,
%\begin{align*}
%P_d=& P(2n+d-1|d-1,0,\ldots,0)\oplus P(2n+d-3|d-3,\ldots,0)\oplus \cdots \oplus P(2n|0,\ldots,0)\\
 %   & \oplus P(-d|d,0,\ldots,0)\oplus P(-d+2|d-2,\ldots,0)\oplus \cdots \oplus P(-1|1,\ldots,0).
%\end{align*}
\begin{align*}
P_d=P^{(-d)}\oplus P^{(-d+2)}\oplus\cdots \oplus P^{(-1)}\oplus P^{(d)}\oplus P^{(d-2)}\oplus\cdots \oplus P^{(1)}.
\end{align*}
The kernel of $f_d$ is
\[Ker(f_d)=\raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$[-d-1]$};
  \node (n2) at (2,0)  {$[-d]$};
  \node (n3) at (3,1)  {$[-d+1]$};
  \node (n4) at (4,0)  {$\cdots$};
  \node (n5) at (5,1) {$[-1]$};
  \node (n6) at (6,0)  {$[0]$};
  \node (n7) at (7,1)  {$[1]$};
  \node (n8) at (8,0)  {$\cdots$};
  \node (n9) at (9,1) {$[d-1]$};
  \node (n10) at (10,0)  {$[d]$};
  \node (n11) at (11,1)  {$[d+1]$};

  \foreach \from/\to in {n1/n2, n2/n3, n3/n4, n4/n5, n5/n6, n6/n7, n7/n8, n8/n9, n9/n10, n10/n11}
    \draw (\from) -- (\to);

\end{tikzpicture}}\]
An inductive argument proves the result. We use the diagrammatic method for modular representations given in \cite{BC}. In particular, we use the description given in \cite[Section~10.3]{BC} for the kernel of the surjective map between a module $M$ and its projective cover $P(M)$. From the same reference, we also use the fact that the projective cover of $M$ is the direct sum of the projective covers of the simple modules in the head of $M$.

The minimal projective resolution of $L(0|0,\ldots,0)$ is given by:
\begin{equation}\label{projrestrivial}
 \ldots P_d\rightarrow \ldots \rightarrow P(0)\rightarrow L(0) \rightarrow 0,
\end{equation}
where the $d^{th}$ term in this resolution is given by: If $d$ is even,
%\begin{align*}
%P_d=& P(2n+d-1|d-1,0,\ldots,0)\oplus P(2n+d-3|d-3,\ldots,0)\\
 %  &\oplus \cdots\oplus P(2n+1|1,\ldots,0) \oplus P(0|0,\ldots,0)\\
  %  & \oplus P(-d|d,0,\ldots,0)\oplus P(-d+2|d-2,\ldots,0)\oplus \cdots \oplus P(-2|2,\ldots,0) ,
%\end{align*}
\begin{align*}
P_d=P^{(-d)}\oplus P^{(-d+2)}\oplus\cdots \oplus P^{(-2)}\oplus P^{(0)}\oplus P^{(d)}\oplus P^{(d-2)}\oplus\cdots \oplus P^{(2)},
\end{align*}
  and if $d$ is odd,
%\begin{align*}
%P_d=& P(2n+d-1|d-1,0,\ldots,0)\oplus P(2n+d-3|d-3,\ldots,0)\oplus \cdots \oplus P(2n|0,\ldots,0)\\
 %   & \oplus P(-d|d,0,\ldots,0)\oplus P(-d+2|d-2,\ldots,0)\oplus \cdots \oplus P(-1|1,\ldots,0).
%\end{align*}
\begin{align*}
P_d=P^{(-d)}\oplus P^{(-d+2)}\oplus\cdots \oplus P^{(-1)}\oplus P^{(d)}\oplus P^{(d-2)}\oplus\cdots \oplus P^{(1)}.
\end{align*}
%\[\dim P^{(-i-1)}=\dim K^{(-i-1)}+\dim K^{(-i)},\]
%\[\dim P(2n+r|r,0,\ldots,0)=\dim K(2n+r|r,0,\ldots,0)+\dim K(2n+r+1|r+1,0,\ldots,0),\]
%and
%\[\dim P(-r|r,0,\ldots,0)=\dim K(-r|r,0,\ldots,0)+\dim K(-r+1|r-1,0,\ldots,0).\]
%Moreover,
%\[\dim P^{(0)}=\dim K^{(0)}+\dim K^{(-1)},\]
%\[\dim P(0|0,0,\ldots,0)=\dim K(0|0,0,\ldots,0)+\dim K(2n|0,0,\ldots,0),\]
\begin{align*}
\dim P_d&=\sum_{r=0}^d \Big(\dim K(2n+r|r,0,\ldots,0)+\dim K(-r|r,0,\ldots,0)\Big)\\
        &=2^{2n}\Big(\sum_{r=1}^d 2\dim L_0(r,0,\ldots,0)+1\Big)\quad\quad(\dim L(0,0,\ldots,0)=1)\\
         &\leq k'd^{2n},\quad\quad\text{By Lemma~(\ref{lem:bddimsimple})}\\
\end{align*}
where $k'$ is a positive constant that depends only on $n$. On the other hand,
\begin{align*}
\dim P_d&=2^{2n}\Big(\sum_{r=1}^d 2\dim L_0(r,0,\ldots,0)+1\Big)\\
        &\geq 2^{2n+1}\Big(\sum_{r=1}^{\frac{d}{2}} Cr^{2n-1}+1\Big)\quad\quad\text{By Lemma~(\ref{lem:bddimsimple})}\\
        &\geq kd^{2n},
\end{align*}
\fi
Since $\dim P(\lambda^L)=\dim K(\lambda)+\dim K(\lambda^L)$, we have for $i\in \Z$,
\[\dim P^{(i)}=\dim K^{(i)}+\dim K^{(i-1)}.\]
By the PBW basis of $U(\fg)$, we have $\dim K(\lambda)=2^{\dim \fg_{-1}}.\dim L_{0}(\lambda)$ where $L_{0}(\lambda)$ is the simple $\mathfrak{sp}(2n)$-module of highest weight $\lambda$ (where we omit the $\lambda_{-1}$ from $\lambda)$. Hence for $i\in \Z$,
\[\dim K^{(i)}=2^{2n}.\dim L_0(i,0,\ldots,0).\]
Consider the case when $d$ is even. Using the minimal projective resolution in \eqref{projrestrivial} we have:
\begin{align*}
\dim P_d&=\dim P^{(-d)}+\dim P^{(d)}+\cdots+\dim P^{(-2)}+\dim P^{(2)}+\dim P^{(0)}\\
        %&=\sum_{i=0}^d \Big(\dim K^{(i)}+\dim K^{(i-1)}\Big)\\
        &=\sum_{i=-d}^d \dim K^{(i)}\\
        &=2^{2n}\sum_{i=1}^d \Big(2\dim L_0(i,0,\dots,0)\Big)+\dim L_0(0,0,\ldots,0)\\
        &\leq 2^{2n+1}\sum_{i=1}^d C'_1i^{2n-1}+1 \quad\quad(Lemma~\ref{lem:bddimsimple})\\
        &\leq 2^{2n+1}\cdot C'_1d\cdot d^{2n-1}+1\\
        &\leq 2^{2n+1}\cdot C'_1d^{2n}+d^{2n}\\
        &= k'd^{2n},
\end{align*}
where $k'$ is a positive constant that depends only on $n$. On the other hand,
\begin{align*}
\dim P_d&=2^{2n}\sum_{i=1}^d \Big(2\dim L_0(i,0,\ldots,0)\Big)+\dim L_0(0,0,\ldots,0)\\
        %&\geq 2^{2n+1}\sum_{i=1}^{\frac{d}{2}} C_1i^{2n-1}+1\\
        %& 2^{2n+1}\cdot C_1\cdot(\frac{d}{2}+1)\cdot(\frac{d}{2})^{2n-1}+1\\
        %&=2C_1d^{2n}+4C_1d^{2n-1}+2^{2n+1}\\
        &\geq 2^{2n+1}\sum_{i=1}^{d} C_1i^{2n-1}+1 \quad\quad(Lemma~\ref{lem:bddimsimple})\\
        &\geq 2^{2n+1}\cdot C_1\sum_{i=d/2}^{d} i^{2n-1}\\
        &\geq 2^{2n+1}\cdot C_1\sum_{i=d/2}^{d} (\frac{d}{2})^{2n-1}\\
        %&\geq 2^{2n+1}\cdot C_1\cdot(\frac{d}{2}+1)\cdot(\frac{d}{2})^{2n-1}+1\\
        &= kd^{2n},
\end{align*}
where $k$ is a positive constant that depends only on $n$. Thus when $d$ is even, there exist constants $k,k'$ depending only on $n$ such that:
\begin{equation}\label{eq2:dimPd}
kd^{2n}\leq \dim P_d \leq k'd^{2n}.
\end{equation}
The same argument can be used to show that inequality~\eqref{eq2:dimPd} holds when $d$ is odd. Therefore, $c_{\F}(L(0|0,\ldots,0)=2n+1$. By \cite[Theorem~4.1.1]{kujawa}, all simple modules of the same atypicality have the same complexity. Thus the complexity of all atypical simple $\osp(2|2n)$-modules is $2n+1$.
\end{proof}
\iffalse
\begin{rk}
We can always give bounds on the dimension of $P(\mu)$ in terms of the dimension of $L(\mu)$. As $\fg_{\0}$-module $L(\mu)$ contains $L_{\0}(\mu)=\C_{\mu_{-1}}\boxtimes L_0(\mu)$ as a composition factor where $L_0(\mu)$ is the simple $\mathfrak{sp}(2n)$-module where we omit the $\mu_{-1}$-part from $\mu$. Thus
 $$\dim P(\mu)\geq \dim L(\mu)\geq \dim L_{\0}(\mu)=\dim L_0(\mu).$$
On the other hand, by the PBW theorem for Lie superalgebras we see that $U(\fg)$ is a free right $U(\fg_{\0})$-module and thus, $U(\fg)\otimes_{U(\fg_{\0})}L_{\0}(\mu)$ is a projective $U(\fg)$-module. Hence it decomposes into a direct sum of projective indecomposable modules. Set $P:=U(\fg)\otimes_{U(\fg_{\0})}L_{\0}(\mu)$. Then by applying Frobenius reciprocity we have
\[\Hom_{\fg}(P,L(\gamma))=\Hom_{\fg_{\0}}(L_{\0}(\mu)),L(\gamma)\Big|_{\fg_{\0}})\neq 0\,\,\text{if}\,\,\mu=\gamma.\]
Thus $P$ surjects onto $L(\mu)$ which implies that $P(\mu)$ is a direct summand of $P$. However, by the PBW theorem, $$P=U(\fg)\otimes_{U(\fg_{\0})}L_{\0}(\mu)\cong \Lambda^{\bullet}(\fg_{\1})\otimes L_{\0}(\mu)$$
as vector spaces. Therefore, we have
\begin{equation}\label{eq1:Bdprojcover}
\dim L_0(\mu)\leq \dim P(\mu)\leq 2^{\dim \fg_{\1}}\dim L_0(\mu).
\end{equation}
Our calculations in the proof of Theorem~\ref{t2:atypcomplexitysimple} matches these bounds.
\end{rk}
\fi
\subsection{Complexity of Kac modules}
Using the computations of the complexity of the trivial module, we can compute the complexity of the Kac module $K(0)$:
\begin{prop}
\[c_{\F}(K(0))=2n.\]
\end{prop}
\begin{proof}
Using the notation developed in this section, the minimal projective resolution of $K(0)$ is given by:
\begin{equation}\label{Ktrivialprojres}
\ldots \rightarrow P^{(-2)}\rightarrow P^{(-1)}\rightarrow P(0)\rightarrow K(0) \rightarrow 0.
\end{equation}
For $d\geq 1$, we have
%\begin{align*}
%\dim P_d &=\dim K(2n+d-1|d-1,0,\ldots,0)+\dim K(2n+d|d,0,\ldots,0)\\
 %        &=2^{2n}\Big(\dim L_0(d-1,0,\ldots,0)+\dim L_0(d,0,\ldots,0)\Big),
%\end{align*}
\begin{align*}
\dim P_d &=\dim P^{(-d)}\\
         &=\dim K^{(-d)}+\dim K^{(-d-1)}\\
         &=2^{2n}\Big(\dim L_0(d,0,\ldots,0)+\dim L_0(d-1,0,\ldots,0)\Big),
\end{align*}
hence $\dim P_d$ is a polynomial in $d$ of degree $2n-1$. Therefore,
\[c_{\F}(K(0))=2n.\]
\end{proof}

We will use the complexity of $K(0)$ to compute the complexity of any atypical Kac module. First we show that that the $L$-operator on the weights does not change the complexity of the Kac modules. Then we give an explicit description of the block $\F^{\chi_{0}}$, where $\chi_{0}$ is the central character corresponding to the weight $\lambda=(0|0,\ldots,0)$. This description will be obtained using the notion of weight diagrams given in \cite[Section~6]{GS}, and the characterization of blocks given in \cite[Section~5]{GS}. We then use the fact that translation functors preserve the complexity to show that $c_{\F}(K(\lambda))=2n$.

\begin{lem}\label{L1:atypcomplexityKac}
For $\lambda\in \x$, $c_{\F}(K(\lambda))=c_{\F}(K(\lambda^{(l)}))$, $l\in \Z$.
\end{lem}

\begin{proof}
It is sufficient to prove that
\[c_{\F}(K(\lambda))=c_{\F}(K(\lambda^{L})).\]
The complexity of any module $M\in \F$ is given by:
\begin{equation}\label{eq1:complexityExt}
c_{\F}(M)=r(\Ext^{\bullet}_{(\fg,\fg_{\0})}(M,\bigoplus L(\mu)^{\dim P(\mu)})),
\end{equation}
where the sum is over all simple modules in $\F$. Let
$S=\bigoplus L(\mu)^{\dim P(\mu)}$. By applying the functor $\Hom(\,\,\,,S)$ to the following 2-step Kac flag:
\[0\rightarrow K(\lambda)\rightarrow P(\lambda^L)\rightarrow K(\lambda^L)\rightarrow 0,\]
we get:
\begin{align*}
& \ldots \rightarrow \Ext^{d+1}(P(\lambda^{L}),S)\rightarrow \Ext^{d+1}(K(\lambda),S)\rightarrow  \Ext^d(K(\lambda^L),S)\rightarrow  \\
& \ldots \rightarrow  \Ext^1(K(\lambda^L),S)\rightarrow \Ext^1(P(\lambda^L),S)\rightarrow\Ext^1(K(\lambda),S)\rightarrow\\
& \Hom(K(\lambda^L),S)\rightarrow \Hom(P(\lambda^L),S)\rightarrow \Hom(K(\lambda),S) \rightarrow 0.
\end{align*}
Since $P(\lambda^L)$ is projective, then $\Ext^d(P(\lambda^L),S)=0$ for all $d\geq 1$, which gives:
\[\Ext^d(K(\lambda^L),S)=\Ext^{d+1}(K(\lambda),S).\]
This implies that $\Ext^{\bullet}_{(\fg,\fg_{\0})}(K(\lambda),S)$ and $\Ext^{\bullet}_{(\fg,\fg_{\0})}(K(\lambda^L),S)$ will have the same rate of growth, hence the theorem follows.
\end{proof}

\subsection{Weight diagrams and translation functors}\label{ss:diagramsfunctors}
 Assume $\l$ is atypical
\[\lambda+\rho=a_1\varepsilon_1+b_1\delta_1+\ldots+b_n\delta_n,\]
 The weight diagram of $\l$ is the function $\tf_\lambda:\Z_{\geq 0}\rightarrow \{>,\,<,\,\times,\,0\}$ represented by a diagram according to the following algorithm:
\begin{enumerate}
\item Put the symbol $>$ in position $t$ if $t=|a_1|$.
\item Put the symbol $<$ in position $t$ for all i such that $t=b_i$.
\item If there are both $>$ and $<$ in the same position replace them by the symbol $\times$, repeat if possible.
\item Put $0$ otherwise.
\end{enumerate}
 For example, the weight diagram of $0=(0|0,\ldots,0)$ is given by:
\[0,<,<,\ldots,<,\times,\ldots,\]
where the first $0$ is at position $0$, the first $<$ is at position $1$, the last $<$ is at position $n-1$, $\times$ is at position $n$, the dots after that stand for empty positions (or zeros). Note that the $\times$ at position $n$ comes from having $|a_1|=b_1=n$ in $0+\rho$. In fact, the number of $\times$ in the weight diagram $\tf_\lambda$ equals the degree of atypicality of $\lambda$. In the case $\atyp(\lambda)=1$, we will not have any $>$ by itself in $\tf_\lambda$. Define the \emph{core} of $\lambda$ to have the same diagram as $\tf_\lambda$ but with replacing all $\times$'s by zeros. Thus the weight diagrams of $\lambda$ and its core have $<,>$ at the same positions. For example, $\tf_{core(0)}$ is given by:
\[0,<,<,\ldots,<,0,\ldots.\]

\begin{rk}
 It is important to note the relation with the notation developed in \cite[Section~1.2]{CWZ} to compute $\lambda^L$. In fact, $f_i=(\lambda+\rho,\delta_i)=-b_i$.
 \end{rk}

We have the following characterization of the blocks in $\F$ in \cite[Section~5]{GS}:
\begin{prop}\label{p1:blocks}
Two dominant weights $\lambda$ and $\mu$ belong to the same block if and only if they have the same atypicality and the same core.
\end{prop}

\begin{cor}\label{c1:blocks}
Let $\chi_{\lambda}$ be the central character corresponding to the weight $\lambda$ and $\F^{\chi_{\lambda}}$ be the corresponding block. Then $\lambda$ and $\lambda^{(l)}$, $l\in \Z$, belong to the same block $\F^{\chi_{\lambda}}$.
\end{cor}
\begin{proof}
It is sufficient to prove that $\lambda$ and $\lambda^{L}$ are in the same block. By the definition of the $L$-operator, $\lambda^L$ and $\lambda$ have the same atypicality. By proposition~\ref{p1:blocks}, we have to show that $\lambda$ and $\lambda^{L}$ have the same core. Consider the three cases that defined $f^L$ in Section~\ref{SS:simplecomplexity} and suppose $f_{-1}=\pm f_i$ for some $1\leq i\leq n$. In the weight diagram of $core(\lambda)$, we will have $<$ at the positions $-f_j$, $j\neq i$. However, by the definition of $f^L$, we will have $<$ at the same $-f_j$, $j\neq i$ in the weight diagram of $core(\lambda^L)$. The corollary follows.
\end{proof}

In the following we have an explicit description of the block $\F^{\chi_{0}}$:
\begin{lem}\label{L1:principalblock}
$\F^{\chi_{0}}=\{\lambda\in  \x\mid \lambda=0^{(l)}\,\,\text{for some}\,\, l\in \Z\}.$
\end{lem}
\begin{proof}
From Corollary~\ref{c1:blocks}, $0^{(l)}\in \F^{\chi_{0}}$, for all $l\in \Z$. Now assume $\lambda\in \F^{\chi_{0}}$. Then $\atyp(\lambda)=1$ and $\lambda$ has the same core as $0$. Thus the weight diagram of $\lambda$ has one $\times$ and $n-1$ $<$'s at positions $1,2,\ldots,n-1$ as in the weight diagram of $0$. The $\times$ cannot be at the zero position because $\lambda$ will not be dominant in that case, so it will be at some position $k\geq n$. The weight diagram of $\lambda$
\[0,<,<,\ldots,<,0,\ldots,0,\times,\ldots,\]
where the $\times$ is at position $k$, from which get:
\[\lambda+\rho=(-k|k,n-1,\ldots,2,1) \quad \text{or}\quad \lambda+\rho=(k|k,n-1,\ldots,2,1).\]
Thus we have two weights:
\[\lambda=(-k+n|k-n,0,\ldots,0) \quad \text{or}\quad \lambda=(k+n|k-n,0,\ldots,0).\]
Let $d=k-n\geq 0$. If $d=0$, then $\lambda=0$ or $\lambda=(2n|0,0,\ldots,0)$. By Lemma~\ref{t1:projcover(2,2n)}, $(2n|0,\ldots,0)^L=0$. Thus $\lambda=0^{(0)}$ or $\lambda=0^{(-1)}$.\\
If $d\geq 1$, then $\lambda=(-d|d,0,\ldots,0)$ or $\lambda=(2n+d|d,0,\ldots,0)$. By Lemma~\ref{t1:projcover(2,2n)}, $(-d|d,0,\ldots,0)=0^{(d)}$ and $(2n+d|d,0,\ldots,0)^{(d+1)}=0$. Thus $\lambda=0^{(d)}$ or $\lambda=0^{(-d-1)}$. This completes the proof.
\end{proof}

\iffalse
The translation functors defined in \cite[Section~5]{GS} can now be described explicitly. The translation functor $T^{\chi,\tau}$ moves a simple module $L(\lambda)\in \F^{\chi}$ to $L(\mu)\in\F^{\tau}$
such that $f_{\mu}$ is obtained from $f_\lambda$ by moving a symbol $<$ or $>$ at position $t$ to the next right position or to the next left position. We can also exchange these two symbols with $0$ and $\times$. We do not allow a symbol to move from or to the zero position.

Assume $\atyp(\lambda)=1$. Let $\chi_{\lambda}$ be the central character corresponding to the weight $\lambda$ and $\F^{\chi_{\lambda}}$ be the corresponding block. Let $\F^{\chi_{0}}$ be the block containing the trivial module. The translation functors define an equivalence of blocks between $\F^{\chi_{\lambda}}$ and $\F^{\chi_{0}}$. To see this, we note that $core(\lambda)$ has $<$ at $n-1$ positions, the same as $core(0)$. Assume the $\times$ in $f_\lambda$ is at position $i$ for some $i\in \N$, then the translation functors allow us to move the $<$'s to the positions $1,2,\ldots,n-1$ and move $\times$ to some position $k\geq n$. The diagram we get is $f_\mu$ where $\mu\in \F^{\chi_{0}}$. The same discussion as in  \cite[Section~6.3]{BKN1} shows that the translation functors preserve the complexity of any $\fg$-module, thus
\fi
Moreover,
\begin{lem}\label{L2:atypcomplexityKac}
If $\lambda \in \F^{\chi_{\lambda}}$ with $\atyp(\lambda)=1$, then
\[c_{\F}(K(\lambda)=c_{\F}(K(\mu)),\]
for some $\mu\in \F^{\chi_{0}}$.
\end{lem}

\begin{proof}
The translation functors defined in \cite[Section~5]{GS} moves a simple module $L(\lambda)\in \F^{\chi}$ to $L(\mu)\in\F^{\tau}$ such that $f_{\mu}$ is obtained from $f_\lambda$ by moving a symbol $<$ or $>$ at position $t$ to the next right position or to the next left position. We can also exchange these two symbols with $0$ and $\times$. We do not allow a symbol to move from or to the zero position.

Let $\chi_{\lambda}$ be the central character corresponding to the weight $\lambda$ and $\F^{\chi_{\lambda}}$ be the corresponding block. Let $\F^{\chi_{0}}$ be the block containing the trivial module. The translation functors define an equivalence of blocks between $\F^{\chi_{\lambda}}$ and $\F^{\chi_{0}}$. To see this, we note that $core(\lambda)$ has $<$ at $n-1$ positions, the same as $core(0)$. Assume the $\times$ in $f_\lambda$ is at position $i$ for some $i\in \N$, then the translation functors allow us to move the $<$'s to the positions $1,2,\ldots,n-1$ and move $\times$ to some position $k\geq n$. The diagram we get is $f_\mu$ where $\mu\in \F^{\chi_{0}}$. The same discussion as in  \cite[Section~6.3]{BKN1} shows that the translation functors preserve the complexity of any $\fg$-module. This completes the proof.
\end{proof}

Therefore,
\begin{thm}\label{t1:atypcomplexityKac}
For atypical $\lambda\in \x$, $c_{\F}(K(\lambda))=2n$.
\end{thm}

\begin{proof}
For an atypical $\lambda\in \x$, we use Lemmas~\ref{L2:atypcomplexityKac}, \ref{L1:atypcomplexityKac}, and \ref{L1:principalblock} to get
\begin{align*}
 c_{\F}(K(\lambda))&=c_{\F}(K(\mu))\quad\text{for some}\,\, \mu\in \F^{\chi_{0}}\\
                  &=c_{\F}(K(0^{(l)}))\quad\text{for some}\,\,l\in \Z\\
                  &=c_{\F}(K(0))=2n.
\end{align*}
\end{proof}

\section{Support, rank and associated varieties}\label{S:varKac}
\subsection{Support variety}
%Let $R=H^{\bullet}(\fg,\fg_{\0};\C)$ be the cohomology ring of $\fg$ and let $M_1,\,M_2\in \F$. According to \cite[Theorem~2.7]{BKN3}, %$\Ext_{\F}^{\bullet}(M_1,M_2)$ is a finitely generated $R$-module. Set $$J_{(\fg,\fg_{\0})}(M_1,M_2):=Ann_R(Ext_{\F}^{\bullet}(M_1,M_2)).$$
% The \emph{relative support variety} of the pair $(M_1,M_2)$ is defined by
%\[\fv_{(\fg,\fg_{\0})}(M_1,M_2):=MaxSpec(R/J_{(\fg,\fg_{\0})}(M_1,M_2)).\]
%In the case $M=M_1=M_2$, set $J_{(\fg,\fg_{\0})}(M)=J_{(\fg,\fg_{\0})}(M,M)$, and
% \[\fv_{(\fg,\fg_{\0})}(M):=\fv_{(\fg,\fg_{\0})}(M,M).\]
 %The variety $\fv_{(\fg,\fg_{\0})}(M)$ is called the \emph{support variety} of $M$. Note that in this situation $J_{(\fg,\fg_{\0})}(M)=Ann_R(Id)$ where %$Id$ is the identity morphism in $\operatorname{Hom}_{\F}(M,M)$.
 Support varieties were defined in chapter 1. The dimensions of the support varieties of the simple modules and the Kac modules in $\F$ are known:

\begin{lem}\cite[Cor~4.4.2]{kujawa}\label{suppvarsimple}\\
For $\lambda\in \x$, $\dim \fv_{(\fg,\fg_{\0})}(L(\lambda))=\atyp(\lambda).$
\end{lem}

\begin{lem}\cite[Cor~3.2]{BKN4}\label{suppvarKac}\\
For $\lambda\in \x$,  $\dim \fv_{(\fg,\fg_{\0})}(K(\lambda))=0$.
\end{lem}

\subsection{Rank variety}\label{ss:rankvar}
The rank variety will be introduced to give a geometric interpretation of the complexity of Kac modules. Furthermore, it will be used to compute the associated variety of the Kac module. For the original definition of the rank variety, and for more details, we refer the reader to \cite[Subsection~6.3]{BKN3}.

Let $\F(\fg_{\pm1})$ be the category of finite dimensional $\fg_{\pm}$-modules. Let $M\in \F(\fg_{\pm1})$
and consider the following rank variety of $M$:
\[\fv_{\fg_{\pm1}}^{rank}(M):=\{x\in \fg_{\pm1} \mid \text{M is not projective as a $U(\langle x \rangle)$-module}\}\cup\{0\},\]
where $U(\langle x \rangle)$ denotes the enveloping algebra of the Lie superalgebra generated by $x\in \fg_{\pm1}$. The rank variety has various properties such as detecting projectivity:

\begin{prop}\cite[Theorem~3.5.1(a)]{BKN2}\label{t2:varKac}
Let $M$ be a $\fg$-module in the category $\F$. Then $M$ is projective if and only if $\fv_{\fg_1}^{rank}(M)=\fv_{\fg_{-1}}^{rank}(M)=0$.
\end{prop}

As a consequence,
\begin{cor}
 If $\lambda\in \x$ is typical, then
\[\fv_{\fg_1}^{rank}(L(\lambda))=\fv_{\fg_1}^{rank}(K(\lambda))=0.\]
\end{cor}
\begin{proof}
$L(\lambda)$ and $K(\lambda)$ are projective and the result follows from Proposition~\ref{t2:varKac}.
\end{proof}

\subsection{Rank variety of Kac modules}
Consider the $\fg_{+1}$ case. We start by computing the rank variety, $\fv_{\fg_1}^{rank}(K(\l))$,  of an atypical Kac module in  $\fg=\osp(2|2)$.

\begin{prop}\label{t2:varKac(2,2)}
For $\fg=\osp(2|2)$ and for an atypical $\lambda\in \x$, $\fv_{\fg_1}^{rank}(K(\lambda))=\fg_1$.
\end{prop}
\begin{proof}
 Since $\fg_1$ is abelian, $[x,x]=0$ for all $x\in \fg_1$. By \cite[Proposition~5.4]{BKN3}, the only simple
$U(\langle x \rangle)$-module is the trivial module $\C$ and the indecomposable modules are $\C$ and $U(\langle x \rangle)$ where $U(\langle x \rangle)$ is two dimensional and it is the projective cover of $\C$. Hence, $K(\l)$ will be projective as a $U(\langle x \rangle)$-module if and only if it does not contain a copy of the trivial module. Thus, $x\in \fg_1$ is an element of $\fv_{\fg_1}^{rank}(K(\l))$ if and only if $K(\l)$ contains a copy of the trivial module when viewed as a $U(\langle x \rangle)$-module.

By \cite[3.8.4]{BKN2}, the action of $G_{\0}=\C^*\times Sp(2n)$ on $\fg_1$ has two orbits $\{0\}$ and $\fg_1\setminus\{0\}$. Note that $\fv_{\fg_1}^{rank}(K(\l))$ is a closed $G_{\0}$-invariant subvariety of $\fv_{\fg_1}^{rank}(\C)=\fg_1$.  Since the closed subvarieties of $\fv_{\fg_1}^{rank}(\C)=\fg_1$ are the closure of those orbits, we have $\fv_{\fg_1}^{rank}(K(\l))=\{0\}$ or $\fv_{\fg_1}^{rank}(K(\l))=\fg_1$.

Hence, to prove the Proposition it is enough to find $0\neq x \in \fg_1$ such that $K(\lambda)$ contains a copy of the trivial module when viewed as a $U(\langle x \rangle)$-module. Note that $\fg_1$ is spanned by the root vectors $x_1$ and $x_2$ with weights $wt(x_1)=\varepsilon_1-\delta_1$ and $wt(x_2)=\varepsilon_1+\delta_1$. On the other hand, $\fg_{-1}$ is spanned by $y_1$ and $y_2$ with weights $wt(y_1)=-wt(x_1)=-\varepsilon_1+\delta_1$ and $wt(y_2)=-wt(x_2)=-\varepsilon_1-\delta_1$. We will need the following elements of $\fg_{\0}$ as we proceed:
\[
z_{11}=1\oplus h,\quad z_{22}=-1\oplus h,\quad z_{12}=0\oplus -2f, \quad\text{and}\quad z_{21}=0\oplus 2e,
\]
where $\{e,f,h\}$ is the standard basis of $\operatorname{sl}(2)$. Since $\osp(2|2)_{\0}\cong \C\oplus \s_2$, the simple $\fg_{\0}$-modules are of the form $L_{\0}(\lambda)=\C_{\lambda_{-1}}\boxtimes L(d)$ where $L(d)$ is the simple $\s_2$-module of dimension $d+1$.  Note that the action of $\fg_{\0}$ on $L_{\0}(\lambda)=\C_{\lambda_{-1}}\boxtimes L(d)$ is given by
 \[(\alpha\oplus T).(c_{\lambda_{-1}}\otimes v)=\alpha.c_{\lambda_{-1}}\otimes v+c_{\lambda_{-1}}\otimes T.v,\]
  for $\alpha,\, c_{\lambda_{-1}} \in \C,\, T\in \operatorname{sl}(2)$ and $v\in L(d)$.

Let $\{v_k\mid 0 \leq k \leq d\}$ be the basis for $L(d)$ with $v_0$ being the highest weight vector given in \cite[Subsection~7.2]{HUM}. Then for $0\leq k\leq d$, we have
\[h.v_k=(d-2k)v_k,\quad e.v_k=(d-k+1)v_{k-1},\quad f.v_k=(k+1)v_{k+1},\]
with $v_{-1}=v_{d+1}=0$.

 Let $w_k=1\otimes v_k$ then $\{w_k\mid 0\leq k\leq d\}$ is a basis of $L_{\0}(\lambda)$ with $w_0$ being the highest weight vector. We can easily show that for $0\leq k\leq d$, we have
\[z_{11}.w_k=-2kw_k,\quad z_{12}.w_k=-2(k+1)w_{k+1},\quad z_{21}.w_k=2(d-k+1)w_{k-1},\]
with $w_{-1}=w_{d+1}=0$. By the PBW-basis theorem for $U(\fg)$, $\dim K(\lambda)=4(d+1)$ with basis $\{y_1^ay_2^b\otimes w_k\mid a,\,b\in \{0,1\},\,1\leq k\leq d\}$.

In the case of $\osp(2|2)$, $\rho=(-1|1)$ then $\lambda=(\lambda_{-1}|\lambda_1)=(\lambda_{-1}|d)$ is atypical if and only if $(\rho+\lambda,\varepsilon_1\pm\delta_1)=0$. Thus we have two cases:\\

\textbf{Case 1}: Using the odd root $wt(x_1)=\varepsilon_1-\delta_1$, we have $\lambda_{-1}=-\lambda_1$. We will write  $\lambda=(-d|d)$.

we can check that $x_1(1\otimes w_0)=1\otimes x_1w_0=0$ since $w_0$ is annihilated by $\fg_1$. By degrees (or $\Z$-grading), $y_i\otimes w_k$ could be in the $U(\langle x_1 \rangle)$-submodule generated by $1\otimes w_0$, but using weights, we can check that only $y_1\otimes w_0$ could land in that submodule. However,
\[x_1(y_1\otimes w_0)=-y_1\otimes x_1w_0+1\otimes z_{11}w_0=0+(-d+d)(1\otimes w_0)=0,\]
therefore, the $U(\langle x_1 \rangle)$-submodule generated by $1\otimes w_0$ forms a trivial direct summand of $K(\lambda)$ when viewed as a $U(\langle x_1 \rangle)$-module. Thus $0\neq x_1 \in \fv_{\fg_1}^{rank}(K(\lambda))$, the result follows in this case.

\textbf{Case 2}: Using the odd root $wt(x_2)=\varepsilon_1+\delta_1$, we have $\lambda_{-1}=\lambda_1+2$. We will write $\lambda=(d+2|d)$. Then we have:
\begin{align*}
x_2(y_1y_2\otimes w_0)= & (-y_1x_1+z_{21})y_2\otimes w_0\\
                      = &-y_1(-y_2x_2+z_{22})\otimes w_0+(2y_1+y_1z_{21})\otimes w_0\\
                      = & y_1y_2\otimes x_2w_0-y_1\otimes z_{22}w_0+2y_1\otimes w_0+y_1\otimes z_{21}w_0\\
                      = & 0-y_1\otimes (-2w_0)+2y_1\otimes w_0+0=0.
\end{align*}
In the above we used the fact the $x_2$ and $z_{21}$ both annihilate the highest weight vector $w_0$. Since $y_1y_2\otimes w_0$ has the largest possible degree (i.e. $-2$), we know that the $U(\langle x_2 \rangle)$-submodule generated by $y_1y_2\otimes w_0$ forms a trivial direct summand of $K(\lambda)$ when viewed as a $U(\langle x_2 \rangle)$-module. Thus $0\neq x_2 \in \fv_{\fg_1}^{rank}(K(\lambda))$, the result follows in this case.
\end{proof}

To generalize the above theorem, we need to use the equivalence of blocks stated in \cite[Theorem~2]{GS}. Indeed, we have:
\begin{thm}\label{t:varkac}
When $\lambda$ is atypical, $\fv_{\fg_1}^{rank}(K(\lambda))=\fg_1$.
\end{thm}

\begin{proof}
Assume $\atyp(\lambda)=k=1$. Let $\chi_{\lambda}$ be the central character corresponding to the weight $\lambda$ and let $\F^{\chi_{\lambda}}$ be the corresponding block. \cite[Theorem~2]{GS} implies that $\F^{\chi_{\lambda}}$ is equivalent to the maximal block of $\fg_k$ containing the trivial module, where $\fg_k=\osp(2|2)$ or $\fg_k=\osp(4|2)$. However, $\fg_k$ is chosen to correspond to a connected sub-Dynkin diagram containing the last node(s) of the diagram of $\fg$ and this is not the case if we use $\osp(4|2)$ because it has the type $D(2,1)$ while $\osp(2|2n)$ has type $C(n+1)$. Set $\fg'=\fg_k=\osp(2|2)$. Using the discussion in \cite[Section~4]{kujawa}, $\fg'$ is the subalgebra of $\fg$ whose roots lie in the intersection of $\Phi$ with the $\mathbb{R}$-span of $\{\pm\varepsilon_1\pm\delta_n,\pm2\delta_n\}$.

As discussed in \cite{DS}, the equivalence of blocks is a composition of translation functors between the blocks of $\F$, followed by a restriction functor from $\fg$ to $\fg'$. Upon restricting to $\fg'$, we have:
\[K(\lambda)=K(\lambda')\oplus M,\]
where $K(\lambda')$ is a Kac $\fg'$-module (the above equivalence of blocks takes Kac modules to Kac modules) and $M$ is a $\fg'$-module. By Serganova's equivalence, $\atyp(\lambda')=\atyp(\lambda)=1$, thus there exists $0\neq x' \in \fv_{\fg'_1}^{rank}(K(\lambda'))=\fg'_1$, (\ref{t2:varKac(2,2)}). By \cite[Theorem~6.6]{BKN3} the rank variety distributes over the direct sum as a union , thus
\[\fv_{\fg'_1}^{rank}(K(\lambda))=\fv_{\fg'_1}^{rank}(K(\lambda'))\cup\fv_{\fg'_1}^{rank}(M).\]
This shows that there exists $0\neq x' \in \fg'\subseteq \fv_{\fg'_1}^{rank}(K(\lambda))$. Using the embedding $\fg'\hookrightarrow\fg$, we can find
$0\neq x \in \fv_{\fg_1}^{rank}(K(\lambda))$ but the $G_{\0}$-orbits of $\fg_1$ are $\{0\}$ and $\fg_1\setminus\{0\}$ which imply that $\fv_{\fg_1}^{rank}(K(\lambda))=\fg_1$.
\end{proof}

Thus we can interpret the complexity of Kac modules geometrically in the following sense:
\begin{cor}
For $\lambda\in \x$, $c_{\F}(K(\lambda))=\dim \fv_{\fg_1}^{rank}(K(\lambda)).$
\end{cor}

\begin{proof}
If $\l$ is typical, the Kac module $K(\l)$ is projective in which case we have
$$c_{\F}(K(\lambda))=\dim \fv_{\fg_1}^{rank}(K(\lambda))=0.$$
If $\l$ is atypical, Theorem~\ref{t1:atypcomplexityKac} and Theorem~\ref{t:varkac} imply that
$$c_{\F}(K(\lambda))=\dim \fv_{\fg_1}^{rank}(K(\lambda))=\dim \fg_1=2n.$$
\end{proof}

After establishing the equivalence of blocks between $\fg$ and $\fg'$, we can show that the operator $L$ is onto without using its combinatorial definition:
\begin{prop}\label{Lsurjective}
The operator $L$ is surjective on the set of atypical weights in $\x$.
\end{prop}
\begin{proof}
First, we will show that $L$ is onto in the case of $\fg'=\osp(2|2)$. Let $\mu$ be an atypical weight, then $\mu=(d+2|d)$ or $\mu=(-d|d)$ for some $d\geq 0$. By Theorem~\ref{t1:projcover(2,2n)}, $(0|0)=(2|0)^L$, $(d+2|d)=(d+3|d+1)^L$ for any $d\geq 0$ and $(-d|d)=(-d+1|d-1)^L$ for any $d\geq 1$. Thus $\mu$ is the image under $L$ of some atypical weight.

Now Let $\mu$ be an atypical weight in the case of $\fg=\osp(2|2n)$. Let $L(\mu')$ be the image of the simple module $L(\mu)$ under the above equivalence of blocks between $\fg$ and $\fg'$. Then $\mu'$ is an atypical weight for $\fg'$. Thus there exists an atypical weight $\lambda'$ with $\lambda'^L=\mu'$. The head of $K(\lambda')$ is $L(\lambda')$ and the socle is $L(\lambda'^L)=L(\mu')$. This Kac module corresponds to a Kac module $K(\lambda)$ (where $\lambda$ corresponds to $\lambda'$ under the same equivalence) which has $L(\lambda)$ as its head and $L(\mu)$ as its socle. But the socle of $K(\lambda)$ is $L(\lambda^L)$, thus $L(\lambda^L)\cong L(\mu)$ which shows $\mu=\lambda^L$ for some atypical $\lambda$.
\end{proof}
\subsection{Associated variety}
Recall
\[\X=\{x\in \fg_{\1} \mid [x,x]=0\}.\]
If $M \in \F$, then Duflo and Serganova \cite{DS} define an associated variety of $M$ which is equivalent to:
\[\X_M=\{x\in \X \mid \text{M is not projective as a $U(\langle x \rangle)$-module}\}\cup \{0\}.\]

We will indicate how the support variety and the associated variety can be combined to measure the complexity of the simple and the Kac modules. First, we find the dimension of the associated variety of the simple and Kac modules:

\begin{lem}\label{assocvarsimple}
For $\lambda\in \x$, let $X(\lambda)$ be $L(\lambda)$ or $K(\lambda)$, then
 \[\dim \X_{X(\lambda)}=\begin{cases}
            2n,& \text{if $\atyp(\lambda)=1$};\\
            0,& \text{if $\atyp(\lambda)=0$.}
            \end{cases}\]
\end{lem}

\begin{proof}
When $\lambda$ is typical, $L(\lambda)$ and $K(\lambda)$ are projective, hence $\dim \X_{X(\lambda)}=0$ by \cite[Theorem~3.4]{DS}. For the rest of the proof, assume $\lambda$ is atypical. \cite[Cor~2.5]{S} implies that $ \X_{L(\lambda)}=\X$ when $\lambda$ is atypical. In $\osp(2|2n)$, $\X$ has two irreducible components each with dimension equals to $\displaystyle{\frac{\dim g_1}{2}}=2n$. Thus $\dim \X=2n$, (\cite[Cor~4.8, Cor~4.9]{DS}).

Now assume $X(\lambda)=K(\lambda)$. From the definitions of the rank variety and the associated variety, we have
\[\fv_{\fg_1}^{rank}(K(\lambda))=\X_{K(\lambda)}\cap \fg_1\subseteq \X_{K(\lambda)}.\]
Following the same proof of \cite[Theorem~6.4.1]{BKN1}, we show that the inclusion is in fact an equality. Thus $\dim \X_{K(\lambda)}=\dim \fv_{\fg_1}^{rank}(K(\lambda))=\dim \fg_{\1}=2n$, (using Theorem~\ref{t:varkac}).

\end{proof}
\subsection{Geometric interpretation of the complexity}
Combining the computations about the complexity, support variety, and the associated variety of the simple and Kac modules, we can conclude that:
\begin{thm}\label{ospgeometric1}
For $\lambda\in \x$, let $X(\lambda)$ be $L(\lambda)$ or $K(\lambda)$, then
$$c_{\F}(X(\lambda))=\dim \X_{X(\lambda)}+\dim \fv_{(\fg,\fg_{\0})}(X(\lambda)).$$
\end{thm}


\section{$z$-complexity}\label{zcomplexity}
The $z$-complexity of $M\in \F$ is defined in section~\ref{zcompdef}. We will compute this categorical invariant for the simple and the Kac modules in $\F$. Recall that if $\p\rightarrow M$ is a minimal projective resolution of $M$ and $s(\p)$ is the rate of growth of the number of projective summands in each $P_d$, then $z_{\F}(M)=s(\p)$.

\begin{lem}
If $L(\lambda)$ is the trivial module, then $z_{\F}(L(\lambda))=2$.
\end{lem}

\begin{proof}
The proof follows directly by counting the number of summands in each $P_d$ in the resolution \eqref{projrestrivial} of the trivial module. By doing so, we can see that the number of summands in each $P_d$ is a polynomial in $d$ of degree $1$. Hence, the rate of growth of this number is $s(\p)=2$. The result follows using Proposition~\ref{zcompcharacterization}.
\end{proof}

To find the $z$-complexity of all atypical simple modules, we will show that simple modules of the same atypicality have the same $z$-complexity.

\begin{lem}
For any module $X\in \F$ and a Kac module $K(\lambda)$, there exists a constant $D_X$ depending only on $X$ such that
\[K(\lambda)\otimes X\cong \bigoplus_{\gamma\in I}K(\gamma),\]
where $|I|\leq D_X$.
\end{lem}

\begin{proof}
By the definition of Kac modules, we have
\[K(\lambda)=U(g)\otimes_{U(\fp^+)}L_{\0}(\lambda).\]
 As a $\fg_{\0}$-module, $L_{\0}(\lambda)\otimes X$ decomposes into a direct sum $\bigoplus_{\mu\in I_X} L_{\0}(\lambda)\otimes L_{\0}(\mu)$, where $I_X$ is a finite indexing set depending only on $X$. By the generalized Littlewood-Richardson formula \cite[Subsection~A.4]{L}, each summand $L_{\0}(\lambda)\otimes L_{\0}(\mu)$ decomposes into a finite direct sum $\bigoplus_{\gamma\in I_{\lambda,\mu}} L_{\0}(\gamma)$ indexed by the $\mathfrak{sp}(2n)$-standard Young tableaux of shape $p(\mu)$ which are $\lambda$-dominant (see \cite[Subsection~A.4]{L} for definitions). Let $l(\mu)$ be the number of boxes in $p(\mu)$, then
 \[|I_{\lambda,\mu}|\leq (2n)^{l(\mu)}.\]
 By using the functor $U(g)\otimes_{U(\fp^+)}-$, we have the following isomorphisms as $U(\fg)$-modules:
\[K(\lambda)\otimes X\cong \bigoplus_{\mu\in I_X}\bigoplus_{\gamma\in I_{\lambda,\mu}} U(g)\otimes_{U(\fp^+)}L_{\0}(\gamma)\cong \bigoplus_{\mu\in I_X}\bigoplus_{\gamma\in I_{\lambda,\mu}} K(\gamma).\]
%where each summand $U(g)\otimes_{U(\fp^+)}L_{\0}(\gamma)$ is a Kac module (hence indecomposable) and
The number of summands is at most $D_X:=\sum_{\mu\in I_X}(2n)^{l(\mu)}$. This completes the proof.
\end{proof}

\begin{cor}
For any module $X\in \F$ and a projective cover $P(\lambda)$, there exists a constant $E_X$ depending only on $X$ such that
\[P(\lambda)\otimes X\cong \bigoplus_{\gamma\in J}P(\gamma),\]
where $|J|\leq E_X$.
\end{cor}
\iffalse
\begin{proof}
Recall the following 2-step Kac flag:
\[0\rightarrow K(\lambda)\rightarrow P(\lambda^L)\rightarrow K(\lambda^L)\rightarrow 0,\]
which can be rewritten as
\[0\rightarrow K(\lambda')\rightarrow P(\lambda)\rightarrow K(\lambda)\rightarrow 0,\]
with $\lambda'^L=\lambda$. By the exactness of the tensor functor (over $\C$), we have an exact sequence:
\[0\rightarrow K(\lambda')\otimes X\rightarrow P(\lambda)\otimes X\rightarrow K(\lambda)\otimes X\rightarrow 0.\]
Thus $P(\lambda)\otimes X$ has a Kac filtration where the factors $K(\lambda)\otimes X$ and $K(\lambda')\otimes X$ decompose into Kac modules by the previous lemma.
For any module $X\in \F$ and a projective cover $P(\lambda)$, there exists a constant $E_X$ depending only on $X$ such that
\[P(\lambda)\otimes X\cong \bigoplus_{\gamma\in J}P(\gamma),\]
where $|J|\leq E_X$.
\end{proof}
\fi
\begin{proof}
Recall the following 2-step Kac flag:
\[0\rightarrow K(\lambda)\rightarrow P(\lambda^L)\rightarrow K(\lambda^L)\rightarrow 0,\]
which can be rewritten as
\[0\rightarrow K(\lambda')\rightarrow P(\lambda)\rightarrow K(\lambda)\rightarrow 0,\]
with $\lambda'^L=\lambda$. This Kac filtration of $P(\l)$ together with the exactness of the tensor functor (over $\C$) imply the exact sequence:
\[0\rightarrow K(\lambda')\otimes X\rightarrow P(\lambda)\otimes X\rightarrow K(\lambda)\otimes X\rightarrow 0.\]
The factors $K(\lambda)\otimes X$ and $K(\lambda')\otimes X$ decompose into Kac modules by the previous lemma. Thus $P(\lambda)\otimes X$ has a Kac filtration. Moreover, there are two numbers $D_X$ and $D'_X$ such that
%\[P(\lambda)\otimes X \cong \begin{array}{cc}
                    %    K(\lambda)\otimes mma. Then there are two numbers $D_X$ and $D'_X$ such that
%\[P(\lambda)\otimes X \cong \begin{array}{cc}
                    %    K(\lambda)\otimes X\\
                     %   K(\lambda')\otimes X
                    %    \end{array}\cong  \begin{array}{cc}
                     %   \bigoplus_{\gamma\in I}K(\gamma)\\
                      %  \bigoplus_{\gamma'\in I'}K(\gamma')
                      %  \end{array},\]
 %Let $I$ and $I'$ be finite sets depending only on $X$ given by the previous lemma. Clearly
 the number of Kac modules in this filtration is at most $E_X:=D_X+D'_X$. On the other hand, the projective module $P(\lambda)\otimes X$ decomposes into a finite direct sum of projective indecomposables $\bigoplus_{\alpha\in R} P(\alpha)$ where each summand has a Kac filtration. However Kac filtrations of the same module will have the same number of Kac modules. In fact, by \cite[proposition~3.3]{BKN4}, the number of times $K(\lambda)$ appears in a kac filtration of a module $M$ is equal to the $\dim \Hom(M, K'(\lambda))$, where $K'$ is the dual Kac module (cf. \cite[Subsection~3.4]{BKN4}). This dimension is not dependent on the choice of the filtration. Thus the number of the projective indecomposables, $P(\alpha)$, is bounded by $E_X$.

\end{proof}

\begin{lem}\label{t:zcompsimple}
Let $M,\, N,\,X,\, T \in \F$ such that $M\otimes X\cong N\oplus T$. Then $z_{\F}(M)\geq z_{\F}(N)$.
\end{lem}

\begin{proof}
Let $\p\rightarrow M\otimes X$ and $\mathbf{Q}.\rightarrow M$ be the minimal projective resolutions of $M\otimes X$ and $M$ respectively. Then $\mathbf{Q}.\otimes X\rightarrow M\otimes X$ is a projective resolution of $M\otimes X$. Using the above corollary, the $d^{th}$ term in this resolution decomposes as follows:
\[Q_d\otimes X = \Bigg(\bigoplus_{\lambda\in R_d} P(\lambda)\Bigg)\otimes X\cong \bigoplus_{\lambda\in R_d}\bigoplus_{\gamma\in I}P(\gamma),\]
where $|I|\leq D_X$. Thus, for each $d$, the number of summands in $Q_d\otimes X$ is at most $|R_d|.D_X$ which is a constant multiple of the number of summands in each $Q_d$. By minimality of the resolution $\p$, we have $s(\p)\leq s(\mathbf{Q}.\otimes X)$ where $s(\p)$ is the rate of growth of the number of summands in the $d^{th}$-term of the resolution $\p$. Then
\[z_{\F}(M\otimes X)=s(\p) \leq s(\mathbf{Q}.\otimes X) \leq s(\mathbf{Q}.)= z_{\F}(M).\]

%Now let $\mathbf{Q'}.\rightarrow N$ and $\mathbf{Q''}.\rightarrow T$ be the minimal projective resolutions of $N$ and $T$ respectively. Then $\mathbf{Q'}.\oplus \mathbf{Q''}.\rightarrow N\oplus T$ is a minimal projective resolution of $N\oplus T\cong M\otimes X$.
By using the definition of the $z$-complexity we can easily show that
\[z_{\F}(N\oplus T)=\operatorname{max}\Big(z_{\F}(N),\,z_{\F}(T)\Big).\]
Then
\[z_{\F}(M\otimes X)\geq z_{\F}(N),\]
thus
\[z_{\F}(M)\geq z_{\F}(M\otimes X)\geq z_{\F}(N).\]
\end{proof}

\begin{thm}\label{t2:zcompsimple}
Let $\lambda \in \x$.
\begin{enumerate}
\item [(1)] If $\lambda$ is typical, then $z_{\F}(L(\lambda))=0$.
\item [(2)] If $\l$ is atypical, $z_{\F}(L(\lambda))=2$.
\end{enumerate}
\end{thm}

\begin{proof}
If $\lambda$ is typical, then $L(\lambda)$ is projective, and hence $z_{\F}(L(\lambda))=0$. Let $\l$ be atypical, then \cite[Corollary~3.2.2]{kujawa} implies that if $\lambda$ and $\mu$ have the same atypicality, there are modules $X_1$ and $X_2$  such that $L(\mu)$ is a direct summand of $L(\lambda)\otimes X_1$ and $L(\lambda)$ is a direct summand of $L(\mu)\otimes X_2$. Lemma~\ref{t:zcompsimple} shows that $z_{\F}(L(\lambda))=z_{\F}(L(\mu))$. Thus, for an atypical $\lambda$, $z_{\F}(L(\lambda))=z_{\F}(\C)=2$.
\end{proof}

\begin{lem}\label{L1:atypzcomplexityKac}
For $\lambda\in \x$, $z_{\F}(K(\lambda))=z_{\F}(K(\lambda^{(l)}))$, $l\in \Z$.
\end{lem}

\begin{proof}
By induction it is sufficient to prove that
\[z_{\F}(K(\lambda))=z_{\F}(K(\lambda^{L})).\]
The $z$-complexity of any module $M\in \F$ is given by:
\begin{equation}\label{eq1:zcomplexityExt}
z_{\F}(M)=r(\Ext^{\bullet}_{(\fg,\fg_{\0})}(M,\bigoplus L(\mu)),
\end{equation}
where the sum is over all simple modules in $\F$. Let
$S=\bigoplus L(\mu)$. By applying the functor $\Hom(\,\,\,,S)$ to the following 2-step Kac flag:
\[0\rightarrow K(\lambda)\rightarrow P(\lambda^L)\rightarrow K(\lambda^L)\rightarrow 0,\]
we get:
\begin{align*}
& \ldots \rightarrow \Ext^{d+1}(P(\lambda^{L}),S)\rightarrow \Ext^{d+1}(K(\lambda),S)\rightarrow  \Ext^d(K(\lambda^L),S)\rightarrow  \\
& \ldots \rightarrow  \Ext^1(K(\lambda^L),S)\rightarrow \Ext^1(P(\lambda^L),S)\rightarrow\Ext^1(K(\lambda),S)\rightarrow\\
& \Hom(K(\lambda^L),S)\rightarrow \Hom(P(\lambda^L),S)\rightarrow \Hom(K(\lambda),S) \rightarrow 0.
\end{align*}
Since $P(\lambda^L)$ is projective, then $\Ext^d(P(\lambda^L),S)=0$ for all $d\geq 1$, which gives:
\[\Ext^d(K(\lambda^L),S)=\Ext^{d+1}(K(\lambda),S).\]
This implies that $\Ext^{\bullet}_{(\fg,\fg_{\0})}(K(\lambda),S)$ and $\Ext^{\bullet}_{(\fg,\fg_{\0})}(K(\lambda^L),S)$ will have the same rate of growth, hence the theorem follows.
\end{proof}

Assume $\atyp(\lambda)=1$. Let $\chi_{\lambda}$ be the central character corresponding to the weight $\lambda$ and $\F^{\chi_{\lambda}}$ be the corresponding block. Let $\F^{\chi_{0}}$ be the block containing the trivial module. The translation functors define an equivalence of blocks between $\F^{\chi_{\lambda}}$ and $\F^{\chi_{0}}$. The $z$-complexity is a categorical equivalence, hence the translation functors preserve the $z$-complexity of any $\fg$-module, thus

\begin{lem}\label{L2:atypzcomplexityKac}
Let $\lambda \in \F^{\chi_{\lambda}}$ with $\atyp(\lambda)=1$. Then
\[z_{\F}(K(\lambda))=z_{\F}(K(\mu)),\]
for some $\mu\in \F^{\chi_{0}}$.
\end{lem}

These results combined show that for atypical $\lambda\in \x$,
\begin{align*}
 z_{\F}(K(\lambda))&=z_{\F}(K(\mu))\quad\text{for some}\,\, \mu\in \F^{\chi_{0}}\\
                  &=z_{\F}(K(0^{(l)}))\quad\text{for some}\,\,l\in \Z\\
                  &=z_{\F}(K(0)).
\end{align*}

\begin{lem}
$z_{\F}(K(0))=1$.
\end{lem}

\begin{proof}
The result follows directly from the minimal projective resolution of $K(0)$ in \eqref{Ktrivialprojres}.
\end{proof}
Therefore,
\begin{thm}\label{zcompkac}
Let $\lambda \in \x$.
\begin{enumerate}
\item [(1)] If $\lambda$ is typical, then $z_{\F}(K(\lambda))=0$.
\item [(2)] If $\lambda$ is atypical, then $z_{\F}(K(\lambda))=1$.
\end{enumerate}
\end{thm}

\subsection{Detecting subsuperalgebra}
As we interpreted the complexity of the simple and the Kac modules geometrically through the dimensions of the associated variety and the support variety, we can also find a geometric interpretation of the $z$-complexity. To do so, a detecting subalgebra is introduced. Let $\ff_{\1}\subseteq \fg_{\1}$ be the span of the root vectors $x_\alpha, x_{-\alpha}$ where $\alpha=\varepsilon_1-\delta_1$. In the matrix realization, $x_\alpha=E_{1,3}-E_{n+3,2}$ and $x_{-\alpha}=E_{2,n+3}+E_{3,1}$. Set $\ff_{\0}=[\ff_{\1},\ff_{\1}]$. Then $\ff_{\0}$ is spanned by the diagonal matrix $E_{1,1}-E_{2,2}+E_{3,3}-E_{n+2,n+2}$. We define a three-dimensional subalgebra of $\fg$ by
\[\ff:=\ff_{\0}\oplus \ff_{\1}.\]
The Lie superalgebra $\ff$ is classical and of Type I and so has a support variety theory. Furthermore, as $[\ff_{\0},\ff_{\1}]=0$, it follows that these varieties admit a rank variety description and, in particular, can be identified as subvarieties of $\ff_{\1}$, i.e.,
\begin{align*}
\fv_{(\ff,\ff_{\0})}(M)&=\fv_{\ff_{\1}}^{rank}(M)\\
                       &=\{y\in \ff_{\1}\mid \text{M is not projective as $U(\langle y \rangle)$-module}\}\cup\{0\}.
\end{align*}
For example, $\fv_{(\ff,\ff_{\0})}(\C)=\ff_{\1}$. If $M$ is projective, then $\fv_{(\ff,\ff_{\0})}(M)=0$.

\begin{prop}\label{t:vardetect}
Let $\lambda \in \x$. If $\lambda \in \x$ is typical, then
\[\fv_{(\ff,\ff_{\0})}(K(\lambda))=\fv_{(\ff,\ff_{\0})}(L(\lambda))=0.\]
If $\lambda$ is atypical, then
\begin{enumerate}
\item [(1)] $\dim \fv_{(\ff,\ff_{\0})}(K(\lambda))=1$.
\item [(2)] $\dim \fv_{(\ff,\ff_{\0})}(L(\lambda))=2$.
\end{enumerate}
\end{prop}
\begin{proof}
 If $\lambda$ is typical, then $L(\lambda)$ and $K(\lambda)$ are projective. Let $\lambda$ be an atypical weight. We can use the proofs of \cite[Theorem~6.4.1,Theorem~9.2.1]{BKN1} to show
\begin{equation}\label{eq1:fvariety}
\fv_{(\ff,\ff_{\0})}(K(\lambda))\subseteq \Big(\ff_{\1}\cap \fv_{\fg_{-1}}(K(\lambda))\Big)\times \Big(\ff_{\1}\cap \fv_{\fg_{1}}(K(\lambda))\Big).
\end{equation}
 From the rank variety description we have
\begin{equation}\label{eq2:fvariety}
\Big(\ff_{\1}\cap \fv_{\fg_{-1}}(K(\lambda))\Big) \cup \Big(\ff_{\1}\cap \fv_{\fg_{1}}(K(\lambda))\Big) \subseteq \fv_{(\ff,\ff_{\0})}(K(\lambda)).
\end{equation}
We have $\fv_{\fg_{1}}(K(\lambda))=\fg_{1}$ and $\fv_{\fg_{-1}}(K(\lambda))=0$ (cf. \cite[Theorem~3.3.1]{BKN2}). Since $\ff_{\1}\cap \fg_{1}$ is one-dimensional, then equation~\eqref{eq1:fvariety} and equation~\eqref{eq2:fvariety} imply that
 \[\dim \fv_{(\ff,\ff_{\0})}(K(\lambda))=1,\]
 and, in fact, $\fv_{(\ff,\ff_{\0})}(X(\lambda))=\ff_{\1}\cap \fg_{1}$.

On the other hand, \cite[Theorem~4.1.1]{kujawa} implies that if $\lambda$ and $\mu$ have the same atypicality, then
 \[\fv_{(\ff,\ff_{\0})}(L(\lambda))=\fv_{(\ff,\ff_{\0})}(L(\mu)).\]
In particular,
 \[\fv_{(\ff,\ff_{\0})}(L(\lambda))=\fv_{(\ff,\ff_{\0})}(L(0|0,\ldots,0))=\ff_{\1},\]
 Thus
 \[\dim \fv_{(\ff,\ff_{\0})}(L(\lambda))=2.\]
\end{proof}

Note that Theorem~\ref{t2:zcompsimple}, Theorem~\ref{zcompkac}, and Proposition~\ref{t:vardetect} imply the following geometric interpretation of the $z$-complexity:
 \begin{thm}\label{ospgeometric2}
 If $X(\lambda)$ is a simple of a Kac module over $\osp(2|2n)$ then
  \[z_{\F}(X(\lambda))=\dim \fv_{(\ff,\ff_{\0})}(X(\lambda)).\]
 \end{thm}

\subsection{Kazhdan-Lusztig polynomials}\label{ss:KLpolynomial}
%To compute the $z$-complexity of the Kac modules, we will make use of Kazhdan-Lusztig Polynomials. Let $\lambda,\mu \in \x$.
The Kazhdan-Lusztig polynomial associated to a pair of weights, $\lambda,\mu \in \x$, is defined by:
\[l_{\lambda,\mu}(-q^{-1})=\sum_{i\geq 0}\dim \Ext_{\F}^i(K(\lambda),L(\mu))q^i.\]
Kazhdan-Lusztig Polynomials can be used to compute the $z$-complexity of the Kac modules. Moreover, by using these polynomials, we will show that
\[\dim \Ext^1(L(\l),L(\mu))=1\Leftrightarrow \l=\mu^L\,\,\text{or}\,\, \l^L=\mu,\]
which was used in determining the radical layer structure of the projective covers. These polynomials have been computed in \cite[Theorem~5, Remark~1]{CWZ}. If $\mu$ is typical, then $l_{\lambda,\mu}=\delta_{\lambda,\mu}$. If $\mu$ is atypical, then
\[
l_{\lambda,\mu}(q)=\begin{cases}
                 (-q)^i,& \text{if $\lambda=\mu^{(i)}$ for some $i\geq 0$};\\
            0,& \text{otherwise,}
            \end{cases}
\]
where $\mu^{(i)}$ was defined in equation~\eqref{eq1:inductivepower}. If $\mu$ is typical then
\begin{equation}\label{eq1:dimExt}
\dim \Ext_{\F}^i(K(\lambda),L(\mu))=\begin{cases}
                 1,& \text{if $\lambda=\mu$ and $i=0$};\\
            0,& \text{otherwise.}
            \end{cases}
\end{equation}
However, if $\mu$ is atypical, we get
\begin{equation}\label{eq2:dimExt}
\dim \Ext_{\F}^i(K(\lambda),L(\mu))=\begin{cases}
                 1,& \text{if $\lambda=\mu^{(i)}$};\\
            0,& \text{otherwise.}
            \end{cases}
\end{equation}
We will also get an alternative condition to get a non zero extension group. Assume $\lambda$ (or $\mu$) is atypical. Using the proof of \cite[Theorem~4.2.1]{BKN1}, we have:
\begin{align}
\dim \Ext_{\F}^d(K(\lambda),L(\mu))\leq \dim \Hom_{\fg_{\0}}(L_{\0}(\lambda), S^d(\fg_{1}^*)\otimes \Lambda^{\bullet}(\fg_{-1})\otimes L_{\0}(\mu)).
\end{align}
Now consider the element $c:=E_{1,1}-E_{2,2}\in \fg_{\0}$. Then $c$ is central in the enveloping algebra of $\fg_{\0}$. Let $\lambda=(\lambda_{-1}|\lambda_1,\dots,\lambda_n)$ and $\mu=(\mu_{-1}|\mu_1,\ldots,\mu_n)$. Note that $c$ acts on the simple module $L(\lambda)$ by $\lambda_{-1}$. However, it acts on $\fg_{\1}$ by $1$ and on $\fg_{-1}$ by $-1$. Thus $c$ acts on $S^d(\fg_{1}^*)\otimes \Lambda^{\bullet}(\fg_{-1})\otimes L_{\0}(\mu)$ by $-d-b+\mu_{-1}$ where $b\in \{0,1,\ldots,\dim \fg_{-1}=2n\}$. Therefore,
\[\dim \Hom_{\fg_{\0}}(L_{\0}(\lambda), S^d(\fg_{1}^*)\otimes \Lambda^{\bullet}(\fg_{-1})\otimes L_{\0}(\mu)) \neq 0 \Rightarrow \lambda_{-1}=-d-b+\mu_{-1},\]
for some $b\in \{0,1,\ldots,2n\}$. Hence
\begin{equation}\label{eq3:dimExt}
\dim \Ext_{\F}^d(K(\lambda),L(\mu))\neq 0 \Rightarrow \mu_{-1}=\lambda_{-1}+d+b,
\end{equation}
for some $b\in \{0,1,\ldots,2n\}$.

From the last equation in \cite[Subsection~3.1]{CWZ} we have
\[\sum_{i\geq 0} \dim \Ext_{\F}^i(L(\lambda),L(\mu))q^i=\sum_{\sigma\in \x}l_{\sigma,\lambda}(-q^{-1})l_{\sigma,\mu}(-q^{-1}),\]
which implies that
\begin{equation}\label{eq4:dimExt}
\dim \Ext_{\F}^d(L(\lambda),L(\mu))=\begin{cases}
                 1,& \text{if $\lambda^{(i)}=\mu^{(j)}$ for some $i,j$ with $i+j=d$};\\
            0,& \text{otherwise.}
            \end{cases}
\end{equation}
In other words,
\begin{equation}\label{eq5:dimExt}
\dim \Ext_{\F}^d(L(\lambda),L(\mu))=\sum_{i+j=d}\sum_{\sigma\in\mathcal{B}}\dim \Ext_{\F}^i(K(\sigma),L(\lambda))\dim \Ext_{\F}^j(K(\sigma),L(\mu)),
\end{equation}
where $\mathcal{B}$ is the block containing $L(\lambda)$ and $L(\mu)$. We can use this equation to show that if
$\dim \Ext_{\F}^d(L(\lambda),L(\mu))\neq 0$, then there exists positive integers $i,j$ with $i+j=d$ and $\sigma$ in the block  $\mathcal{B}$ containing $\lambda, \mu$ such that $\dim \Ext_{\F}^i(K(\sigma),L(\lambda))\neq 0$ and $\dim \Ext_{\F}^j(K(\sigma),L(\mu))\neq 0$. Then by equation~\eqref{eq3:dimExt}, we have
\[\sigma_{-1}=\lambda_{-1}-i-b_1\quad \text{and}\quad \sigma_{-1}=\mu_{-1}-j-b_2,\]
hence
\begin{equation}\label{eq6:dimExt}
\dim \Ext_{\F}^d(L(\lambda),L(\mu))\neq 0\Rightarrow \lambda_{-1}-\mu_{-1}=i-j+b_1-b_2,
\end{equation}
for some $b_1,b_2\in \{0,1,\dots,2n\}$ and some $i,j$ such that $i+j=d$.

We have the following alternative proof of Theorem~\ref{zcompkac} using Kazhdan-Lusztig polynomials:
\begin{proof}
If $\lambda$ is typical, then $K(\lambda)$ is projective, and hence $z_{\F}(K(\lambda))=0$. Assume $\lambda$ is atypical. Equation~\eqref{eq3:dimExt} gives
\begin{equation}
\dim \Ext_{\F}^d(K(\lambda),L(\mu))\neq 0 \Rightarrow \mu_{-1}=\lambda_{-1}+d+b,
\end{equation}
for some $b\in \{0,1,\ldots,2n\}$, which implies that
\begin{align}
\dim \Ext_{\F}^d(K(\lambda),\bigoplus L(\mu))=\sum_{\mu}\sum_{b=0}^{2n}\dim \Ext_{\F}^d(K(\l),L(\mu)),
\end{align}
where the first sum runs over all $\mu$ that are in the same block of $\lambda$. Hence, $\mu$ and $\lambda$ have the same atypicality (i.e, 1) and the same core. Let $\l=(\lambda_{-1}|\lambda_1,\ldots,\lambda_n)$ then $\mu=(\lambda_{-1}+d+b|\mu_1,\ldots,\mu_n)$. Let $\mu_k$ be the component that gives $\atyp(\mu)=1$. Then $\mu_k$ is uniquely determined by $\mu_{-1}$. However, the components $\mu_1,\mu_2,\ldots,\mu_{k-1},\mu_{k+1},\ldots,\mu_n$ are determined by $\lambda$ since $\mu$ and $\lambda$ have the same core. We have $$\Ext_{\F}^d(K(\lambda),L(\mu))=1,$$
 when it is not zero. Thus
 \[\dim \Ext_{\F}^d(K(\lambda),\bigoplus L(\mu))\leq 2n+1=(2n+1)d^{1-1},\]
  which proves $z_{\F}(K(\lambda))\leq 1$. Assume $z_{\F}(K(\lambda))=0$, then there exists $N\in \N$ such that
\[\dim \Ext_{\F}^d(K(\lambda),\bigoplus L(\mu))=0,\]
for all $d>N$. Thus if $\p\rightarrow K(\lambda)$ is a minimal projective resolution, then $P_d=0$ for all $d>N$. Assuming the resolution has at least two projective modules, it can be rewritten as:
\[0\rightarrow P_N\hookrightarrow P_{N-1}\rightarrow P_{N-2}\rightarrow \dots \rightarrow K(\lambda)\rightarrow 0.\]
Since projective modules are injective in $\F$, the short exact sequence
\[0\rightarrow P_N\hookrightarrow P_{N-1}\rightarrow P_{N-1}/P_N\rightarrow 0\]
splits, hence $P_{N-1}\cong P_N \oplus P_{N-1}/P_N$ which makes $P_{N-1}/P_N$ projective. Thus the projective resolution can be written as:
\[0\rightarrow P_N\hookrightarrow P_N\oplus P_{N-1}/P_N\rightarrow P_{N-2}\rightarrow \dots \rightarrow K(\lambda)\rightarrow 0,\]
but the exactness of the original resolution implies the exactness of the following:
\[0\rightarrow P_{N-1}/P_N\rightarrow P_{N-2}\rightarrow \dots \rightarrow K(\lambda)\rightarrow 0,\]
which contradicts the minimality of the original resolution. Now if the resolution had only one projective module, then $K(\lambda)$ is projective which is not the case since $\lambda$ is atypical. This shows that $z_{\F}(K(\lambda))=1$.

%For simple modules, we use equation~\ref{eq6:dimExt} to show that if $\dim \Ext_{\F}^d(L(\lambda),L(\mu))\neq 0$, then there exist positive integers $i,j$ with $i+j=d$ such that
%\[\lambda_{-1}-\mu_{-1}=i-j+b_1-b_2,\]
%for some $b_1,b_2\in \{0,1,\dots,2n\}$. As in the case of Kac modules, $\mu$ is determined by $\mu_{-1}$ and $\lambda$ since $\mu$ and $\lambda$ are in the same block.  We saw in Section~\ref{ss:KLpolynomial} that $\dim \Ext_{\F}^d(L(\lambda),L(\mu))\leq 1$, thus
%\[\dim \Ext_{\F}^d(L(\lambda),\bigoplus L(\mu))\leq \mathop{{\sum_{i,j}}}_{i+j=d} 1=d+1,\]
%which implies $z_{\F}(L(\lambda))\leq 2$. If $z_{\F}(L(\lambda))\leq 1 $, then there is a positive constant $C$ such that
%\[\sum_{\mu\in \mathcal{B}}\dim \Ext_{\F}^d(L(\lambda),L(\mu))\leq Cd^{1-1}=C.\]

%On the other hand, we can use Theorem~\ref{t1:Bdprojcover} to show
%\[\dim \Ext_{\F}^d(L(\lambda),\bigoplus L(\mu)^{\dim P(\mu)})=\sum_{\mu\in \mathcal{B}} \dim P(\mu)\dim \Ext_{\F}^d(L(\lambda),L(\mu))\leq C'd^{2n-1},\]
%which gives $c_{\F}(L(\lambda)\leq 2n$, contradicting Theorem~\ref{t2:atypcomplexitysimple}. Thus $z_{\F}(L(\lambda))= 2$.
\end{proof}

\iffalse
\subsection{Upper bound on the dimensions of projective covers}
We first relate the dimension of $P(\mu)$ to the dimension of $L(\mu)$. As $\fg_{\0}$-module $L(\mu)$ contains $L_{\0}(\mu)=\C_{\mu_{-1}}\boxtimes L_0(\mu)$ as a composition factor ($L_0(\mu)$ is the simple $\mathfrak{sp}(2n)$-module with the slight abuse of notation).  Thus $\dim P(\mu)\geq \dim L_0(\mu)$ (note that $\dim L_{\0}(\mu)=\dim L_0(\mu)$). On the other hand, by the PBW theorem for Lie superalgebras we see that $U(\fg)$ is a free $U(\fg_{\0})$-module and thus, $U(\fg)\otimes_{U(\fg_{\0})}L_{\0}(\mu)$ is a projective $\fg$-module. Furthermore, by applying Frobenius reciprocity we see that $U(\fg)\otimes_{U(\fg_{\0})}L_{\0}(\mu)$ surjects onto $L(\mu)$. Thus, $P(\mu)$ is a direct summand of $U(\fg)\otimes_{U(\fg_{\0})}L_{\0}(\mu)$. However, by the PBW theorem, $U(\fg)\otimes_{U(\fg_{\0})}L_{\0}(\mu)\cong \Lambda^{\bullet}(\fg_{\1})\otimes L_{\0}(\mu)$ as vector spaces. Therefore, we have
\begin{equation}\label{eq1:Bdprojcover}
\dim L_0(\mu)\leq \dim P(\mu)\leq 2^{\dim \fg_{\1}}\dim L_0(\mu).
\end{equation}

\begin{thm}\label{t1:Bdprojcover}
Let $\p\rightarrow L(\lambda)$ be a minimal projective resolution of an atypical simple module. Then there is a positive constant $C$ depending on $n$ and $\lambda$ such that if $P(\mu)$ appears as a direct summand of $P_d$, then
\[\dim P(\mu)\leq Cd^{2n-1}.\]
\end{thm}

\begin{proof}
From equation~\eqref{eq1:Bdprojcover}, it is enough to show that
\[\dim L_0(\mu)\leq Cd^{2n-1},\]
for some constant $C$ depending on $n$ and $\lambda$. From the Weyl-dimension formula, we have
\[\dim L_0(\mu)=\frac{\prod_{\alpha\in \Phi_{\0}^+}(\mu+\delta,\alpha)}{\prod_{\alpha\in \Phi_{\0}^+}(\delta,\alpha)},\]
where
\[
\Phi_{\0}^{+}=\{\delta_i+\delta_j\mid 1\leq i\leq j\leq n\}\cup\{\delta_i-\delta_j\mid 1\leq i<j \leq n\},
\]
and $\delta=\displaystyle{\sum_{i=1}^{n}(n-i+1)\delta_i}$. For atypical $\lambda$, there exists $1 \leq k\leq n$ such that $(\lambda+\rho,\varepsilon_1)=\pm(\lambda+\rho,\delta_k)$. The core of $\lambda$ can be described by the set
\[C(\lambda)=\{(\lambda+\rho,\delta_i)\mid 1 \leq i\leq n,i\neq k\}.\]
Consider the sets:
\[A_n(\lambda)=\{\alpha=\delta_i\pm \delta_j\in \Phi_{\0}^+\mid |\{i,j\}\cap\{k\}|=0\},\]
\[B_n(\lambda)=\{\alpha=\delta_i\pm \delta_j\in \Phi_{\0}^+\mid |\{i,j\}\cap\{k\}|=1\}.\]
Clearly  $\Phi_{\0}^+=A_n(\lambda)\sqcup B_n(\lambda)$. If $P(\mu)$ appears in $P_d$, then $\lambda$ and $\mu$ are in the same block, hence they have the same atypicality and core, i.e, there exists $1 \leq k'\leq n$ such that $(\mu+\rho,\varepsilon_1)=\pm(\mu+\rho,\delta_k')$ and $C(\lambda)=C(\mu)$. Since $\delta=\rho+n\varepsilon_1$, we have
\[\{(\lambda+\delta,\delta_i)\mid 1 \leq i\leq n,i\neq k\}=\{(\mu+\delta,\delta_i)\mid 1 \leq i\leq n,i\neq k'\},\]
then we can easily see that
\[\{(\lambda+\delta,\alpha)\mid  \alpha\in A_n(\lambda)\}=\{(\mu+\delta,\alpha)\mid  \alpha\in A_n(\mu)\}.\]
Note that for $\alpha\in \Phi_{\0}^+$, $(\delta,\alpha)<0$ and $(\mu+\delta,\alpha)<0$. Thus
\begin{align*}
\dim L_0(\mu)&=-\frac{1}{\prod_{\alpha\in \Phi_{\0}^+}(\delta,\alpha)}\prod_{\alpha\in A_n(\mu)}-(\mu+\delta,\alpha)\prod_{\alpha\in B_n(\mu)}-(\mu+\delta,\alpha)\\
             &= -\frac{1}{\prod_{\alpha\in \Phi_{\0}^+}(\delta,\alpha)}\prod_{\alpha\in A_n(\lambda)}-(\lambda+\delta,\alpha)\prod_{\alpha\in B_n(\mu)}-(\mu+\delta,\alpha)\\
             &=K_1\prod_{\alpha\in B_n(\mu)}-(\mu+\delta,\alpha),
\end{align*}
where $K_1$ is a positive constant depending on $n$ and $\lambda$. Since $P(\mu)$ appears in $P_d$, then $0\neq \Hom_{\F}(P_d,L(\mu))$. But as $\p$ is a minimal projective resolution, we have
\[\Ext_{\F}^d(L(\lambda),L(\mu))=\Hom_{\F}(P_d,L(\mu))\neq 0.\]
Equation~\eqref{eq6:dimExt} implies that there exist $i,j$ with $i+j=d$ such that
\[\lambda_{-1}-\mu_{-1}=i-j+b_1-b_2\geq -d+b_1-b_2,\]
 for some $b_1,b_2\in \{0,1,\dots,2n\}$. Using some algebraic manipulations, we can show
\[|\mu_{-1}|\leq d+D,\]
where is a positive constant depending only on $n$ and $\lambda$.\\

Recall that there exists $1 \leq k'\leq n$ such that $(\mu+\rho,\varepsilon_1)=\pm(\mu+\rho,\delta_k')$, so we have two cases to consider:

$\mathbf{Case\,\, I}$: Suppose $-n+\mu_{-1}=n-k'+1+\mu_{k'}$, then
\[0\leq \mu_{k'}=-2n+k'-1+\mu_{-1}\leq -n-1+\mu_{-1}\leq \mu_{-1}\leq d+D.\]

 If $\alpha=\delta_i-\delta_j\in B_n(\mu)$, then $i<j$ and $i=k'$ or $j=k'$. If $i=k'$, then
\[-(\mu,\alpha)=\mu_{k'}-\mu_{j}\leq \mu_{k'}\leq d+D.\]
If $j=k'$, then $-(\mu,\alpha)=\mu_{i}-\mu_{k'}$, but $i\neq k'$, thus $(\mu+\rho,\delta_i)=(\lambda+\rho,\delta_l)$ for some $1\leq l\neq k\leq n$ which gives $\mu_i=\lambda_l+i-l\leq \lambda_l+n$. Thus
\[-(\mu,\alpha)\leq \lambda_l+n-\mu_{k'}=\lambda_l+3n-k'+1-\mu_{-1}\leq d+D+\lambda_l+3n.\]

 If $\alpha=\delta_i+\delta_j\in B_n(\mu)$ with $i<j$, then $i=k'$ or $j=k'$. If $i=k'$, we have $\mu_j\leq \mu_{k'}$, then
\[-(\mu,\alpha)=\mu_{k'}+\mu_{j}\leq 2\mu_{k'}\leq 2(d+D).\]
If $j=k'$, then
\[-(\mu,\alpha)=\mu_i+\mu_{k'}\leq d+D+\lambda_l+n.\]

If $\alpha=2\delta_{k'}\in B_n(\mu)$, then
\[-(\mu,\alpha)=2\mu_{k'}\leq 2(d+D).\]

$\mathbf{Case\,\, II}$: Suppose $-n+\mu_{-1}=-(n-k'+1+\mu_{k'})$, then
\[0\leq \mu_{k'}=k'-1-\mu_{-1}\leq n-1-\mu_{-1}\leq d+D+n.\]

In a similar fashion to the first case, we can show that if $\alpha\in B_n(\mu)$, then
 there are positive constants $C_1,C_2$ depending only on $n$ and $\lambda$ such that
\[-(\mu,\alpha)\leq C_1.d+C_2,\]
Since $B_n(\mu)$ has $2n-1$ roots, then
\begin{align*}
\dim L_0(\mu)&=K_1\prod_{\alpha\in B_n(\mu)}-(\mu+\delta,\alpha)\\
             &\leq K_1.C_1^{2n-1}.d^{2n-1}+K_1.{C_2}^{2n-1}+\prod_{\alpha\in B_n(\mu)}-(\delta,\alpha)\\
             &=C_1'.d^{2n-1}+C'_2,
\end{align*}
where $C'_1,C'_2$ are positive constants depending only on $n$ and $\lambda$. Let $C=\operatorname{max}(C_1',C'_2)$, then $$\dim L_0(\mu)\leq C.d^{2n-1}+C\leq 2C.d^{2n-1}.$$
Combining this with equation~\eqref{eq1:Bdprojcover} will complete the proof.
\end{proof}
\fi




\chapter{Computing the complexity over $\osp(3|2)$}


After computing the complexity of the simple and Kac modules over the type $C$ Lie superalgebra, we look at the type $B$ Lie superalgebra $\osp(3|2)$. This Lie superalgebra is of \emph{Type II} and hence the Kac modules (as defined before) are not finite dimensional. The simple modules are characterized by highest weights. Using the description of the projective covers given in \cite{Germoni}, we will compute the complexity and the $z$-complexity of the simple modules over $\osp(3|2)$. We show these complexities have the same geometric interpretation as was shown for types $A$ and $C$.

\section{The Lie superalgebra $\osp(3|2)$}
Throughout this chapter, let $\fg=\osp(3|2)$. The definition of the general orthosymplectic Lie superalgebra $\osp(m|n)$ was given in Chapter~2. The bilinear form $F$ given there can be represented by the matrix:
 \[
F=\left[\begin{array}{cc}
G & 0\\
0 & H\\
\end{array}\right],\]
where \[G=\left[\begin{array}{ccc}
1 & 0 & 0\\
0 & 0 & 1\\
0 & 1 & 0\\
\end{array}\right]\quad\text{and}\quad
H= \left[\begin{array}{cc}
0 & 1\\
-1 & 0\\
\end{array}\right].
\]

Then $\fg$ consists of all the matrices
\[
\left[\begin{array}{c|c}
\begin{array}{ccc}
0 & -u & -v\\
v & a & b\\
u & c & -a\\
\end{array}  & \begin{array}{ccc}
x & x_1\\
y & y_1\\
z & z_1
\end{array}\\ \hline
\begin{array}{ccc}
-x_1 & -z_1 & -y_1\\
x & z & y\\
\end{array} & \begin{array}{cc}
d & e\\
f & -d\\
\end{array}\\
\end{array}
\right].
\]

The even part of $\fg$ is $\fg_{\0} \cong
\mathfrak{so}_3\oplus\mathfrak{sp}_2\cong \s_2\oplus\s_2$,
hence $\dim\fh^*=2$. We can choose a basis $\{\eps,\d\}$ of $\fh^*$ such that
\[\begin{array}{c}
\Phi_{\0}=\{\pm\eps ; \pm2\d\}, \qquad \Phi_{\1}=\{\pm\eps\pm\d ; \pm\d\},\\
\mathrm{and}\qquad (\eps,\eps)=-(\d,\d)=1, \quad (\eps,\d)=0.\blouk
\end{array}\]

We choose the following distinguished set of positive roots~:
$\Phi^+=\{\eps ; 2\d\}\cup\{\pm\eps+\d ; \d\}$ which corresponds to the simple roots
$-\eps+\d$ and $\eps$. Here we have $2\rho=\eps-\d$. Using the basis $\{\eps,\d\}$,
dominant weights for $\fg$ will be identified with $\N/2\times\N$. By proposition 2.3 in \cite{Kac1}, the set of dominant weights for $\fg$ is~:
$X^+=\{(a,b)\in\N/2\times\N\tq b=0\Rightarrow a=0\}$.
The atypical dominant weights are $\l_0=(0,0)$ and $\l_l=(l-1,l)$ (for $l\in\N^*$).
For example, $S(\l_0)$ is the trivial module, $S(\l_1)$ is the standard module.
%It can be checked that $\cas(\l_l)=0$ for any $l\in\N$.

\begin{thm}\cite[Theorem~2.1.1]{Germoni} \label{tru1}
 Let $\fg=\osp(3|2)$.

\toto{a}\ Up to a shift of parity, the principal block $\Ga_0=\{\l_l\tq l\in\N\}$
is the unique atypical block.

\toto{b} The projective indecomposable modules have the following radical layer structure~:

\[\raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$S(\l_0)$};
  \node (n2) at (1,0)  {$S(\l_2)$};
  %\node (n3) at (2,0)  {$[1]$};
  \node (n4) at (1,-1) {$S(\l_0)$};


  \foreach \from/\to in {n1/n2,n2/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\quad \raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$S(\l_1)$};
  \node (n2) at (1,0)  {$S(\l_2)$};
  %\node (n3) at (2,0)  {$[1]$};
  \node (n4) at (1,-1) {$S(\l_1)$};


  \foreach \from/\to in {n1/n2,n2/n4}
    \draw (\from) -- (\to);

\end{tikzpicture}}\quad \raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$S(\l_2)$};
  \node (n2) at (0,0)  {$S(\l_0)$};
  \node (n3) at (1,0)  {$S(\l_1)$};
  \node (n4) at (2,0) {$S(\l_3)$};
  \node (n5) at (1,-1)  {$S(\l_2)$};


  \foreach \from/\to in {n1/n2,n1/n3, n1/n4, n2/n5, n3/n5, n4/n5}
    \draw (\from) -- (\to);

\end{tikzpicture}}\quad \raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$S(\l_1)$};
  \node (n2) at (0,0)  {$S(\l_{l-1})$};
  \node (n3) at (2,0)  {$S(\l_{l+1})$};
  \node (n4) at (1,-1) {$S(\l_l)$};



  \foreach \from/\to in {n1/n2,n1/n3, n3/n4, n2/n4}
    \draw (\from) -- (\to);\end{tikzpicture}} \quad (l\geq3)\]
\iffalse
\[ \begin{array}{cccccccccc}
S(\l_0) & & S(\l_1) & & S(\l_2) & & S(\l_l) \\
S(\l_2) &\quad & S(\l_2) &\quad & S(\l_0)\ \ S(\l_1)\ \ S(\l_3) &\quad & S(\l_{l-1})\ \  S(\l_{l+1}) & (l\geq3) \\
S(\l_0) & & S(\l_1) & & S(\l_2) & & S(\l_l)
\end{array} \]
\fi
\end{thm}

\begin{rk}
It is worth mentioning that the author in \cite{Germoni} considers the category $\fg$-mod of finite-dimensional $\fg$-modules with
even morphisms of representations. On the other hand, we take the category $\F$ of finite-dimensional $\fg$-modules with
all morphisms of representations. However, his choice of morphisms does not affect the structure of the projective modules. We will be using his classification of the simple modules in $\fg$-mod which contains the simple modules in $\F$. This remark applies in the next chapters where we consider the three exceptional Lie superalgebras.
\end{rk}

We first relate the dimension of $P(\mu)$ to the dimension of $S(\mu)$. Note that $S_{\0}(\l_0)$ is the trivial $\fg_{\0}$-module and for $\l_l=(l-1,l)\in X^+$, we let $S_{\0}(\l_l)=V_{l-1}\otimes V_l$ where $V_m$ denotes the simple $\s_2$-module of dimension $m+1$. As $\fg_{\0}$-modules, $S(\mu)$ contains $S_{\0}(\mu)$ as a composition factor. Thus
 $$\dim P(\mu)\geq \dim S(\mu)\geq \dim S_{\0}(\mu).$$

On the other hand, by the PBW theorem for Lie superalgebras we see that $U(\fg)$ is a free right $U(\fg_{\0})$-module and thus, $U(\fg)\otimes_{U(\fg_{\0})}S_{\0}(\mu)$ is a projective $U(\fg)$-module. Hence it decomposes into a direct sum of projective indecomposable modules. Set $P:=U(\fg)\otimes_{U(\fg_{\0})}S_{\0}(\mu)$. Then by applying Frobenius reciprocity we have
\[\Hom_{\fg}(P,S(\gamma))=\Hom_{\fg_{\0}}(S_{\0}(\mu)),S(\gamma)\Big|_{\fg_{\0}})\neq 0\,\,\text{if}\,\,\mu=\gamma.\]
Thus $P$ surjects onto $L(\mu)$ which implies that $P(\mu)$ is a direct summand of $P$. However, by the PBW theorem, $$P=U(\fg)\otimes_{U(\fg_{\0})}S_{\0}(\mu)\cong \Lambda^{\bullet}(\fg_{\1})\otimes S_{\0}(\mu)$$
as vector spaces. Therefore, we have
\begin{equation}\label{eq1:Bdprojcover}
\dim S_{\0}(\mu)\leq \dim P(\mu)\leq 2^{\dim \fg_{\1}}\dim S_{\0}(\mu).
\end{equation}
 Thus
\begin{equation}\label{eq1':Bdprojcover}
l(l+1)\leq \dim P(\l_l)\leq 2^6 l(l+1).
\end{equation}

\section{Complexity of simple $\osp(3|2)$-modules}
We will use the above structure of the projective covers to give a minimal projective resolution of the trivial module $S(\l_0)$, then we use the generalized Kac-Wakimoto conjecture to show that any atypical simple module will have the same complexity as the trivial module. Recall that if the simple module is typical, then it is projective and hence has zero complexity.
\begin{thm}\label{osp(3|2)simple}
For an atypical $\l_l\in X^+$, $c_{\F}(S(\l_l))=4.$
\end{thm}

\begin{proof}
The minimal projective resolution of $S(\l_0)$ is given by:
\begin{align}\label{projres1}
 \ldots \rightarrow P_d\rightarrow \ldots \rightarrow P(\l_0)\rightarrow S(\l_0) \rightarrow 0,
\end{align}
where the $d^{th}$ term in this resolution is given by:
\begin{align*}
P_d=& P(\l_{d+1})\oplus P(\l_{d-1})\oplus \cdots\oplus P(\l_r),
    \end{align*}
where
\[r=\begin{cases}
  2 \quad \text{if $d$ is odd,}\\
   0 \quad \text{if $d\equiv 0 \mod 4$,}\\
   1 \quad \text{if $d\equiv 2 \mod 4$.}
  \end{cases}\]
The construction of this minimal projective resolution follows our construction in Theorem~\ref{t2:atypcomplexitysimple}. We use the same diagrammatic approach that was established in \cite[Section~10.3]{BC}. Using equation~\eqref{eq1':Bdprojcover} we get for all $d$:
\begin{align*}
\dim P_d&\leq 2^6\sum_{l=1}^{d+1} \Big(l(l+1)\Big)\leq C.d^3,
\end{align*}
for some positive constant $C$. On the other hand,
\begin{align*}
\dim P_d&\geq \frac{1}{2}\sum_{l=1}^{d+1} l^2 \geq C'.d^3,
\end{align*}
for some positive constant $C'$. This shows that $c_{\F}(S(\l_0))=4$. By \cite[Theorem~4.1.1]{kujawa}, all simple modules of the same atypicality have the same complexity. Thus the complexity of all atypical simple $\osp(3|2)$-modules is $4$.
\end{proof}
\subsection{A goemetric interpretation of the complexity}
We can then interpret the complexity of simple modules geometrically:
\begin{thm}\label{GMosp(3|2)}
If $S$ is a simple $\fg$-module, then
$$c_{\F}(S)=\dim \X_{S}+\dim \fv_{(\fg,\fg_{\0})}(S).$$
\end{thm}
\begin{proof}
If $S$ is typical then S is projective and both sides are zero. Let $S$ be an atypical simple module (i.e. $\atyp(S)=\df(\fg)=1$).
Let
\[\X=\{x\in \fg_{\1} \mid [x,x]=0\}.\]
Let $G_{\0}$ be an algebraic group with Lie algebra $\fg_{\0}$. Then $\X$ is invariant under the adjoint action of $G_{\0}$. It was shown that for any $x\in \X$, there exists $g\in G_{\0}$ and isotropic mutually orthogonal linearly independent roots $\alpha_1,\ldots, \alpha_k$ such that $Ad_g(x)=x_1+\ldots+x_k$ with $x_i\in \fg_{\alpha_i}$. The number $k$ does not depend on the choice of $g$ and is called the \emph{rank} of $x$ (notation $rk(x)=k$). Let
\[X_k=\{x\in \X\,|\,rk(x)=k\}.\]

Corollary~2.5 in \cite{S} implies that for an atypical simple module $S$, we have  $\X_{S}=\overline{X_1}=\X$ by the above stratification. Using \cite[Theorem~4.5, Cor~4.8]{DS}, $\dim \X=3$ in case $\fg=\osp(3|2)$. On the other hand, by \cite[Cor~4.4.2]{kujawa} we have $\dim \fv_{(\fg,\fg_{\0})}(S)=\atyp(S)=1$. Thus
$$c_{\F}(S)=\dim \X_{S}+\dim \fv_{(\fg,\fg_{\0})}(S).$$
\end{proof}


\section{$z$-complexity}
Using the projective resolution given in the proof of Theorem~\ref{osp(3|2)simple}, we have
\begin{prop}
$z_{\F}(S(\l_0))=2$.
\end{prop}
\begin{proof}
The $z$-complexity of a module is the rate of growth of the number of direct summands in a minimal projective resolution of the module. In the proof of Theorem~\ref{osp(3|2)simple}, the number of summands in $P_d$ is a either $\dspt{\frac{d}{2}+1}$ if $d$ is even or $\dspt{\frac{d+1}{2}}$ if $d$ is odd. Thus the rate of growth of this number is $2$.
 \end{proof}
 We will also show that:
 \begin{prop}
 For $l\geq 1$, we have $z_{\F}(S(\l_l))=2$.
 \end{prop}

 \begin{proof}
 Since the projective cover of $S(\l_1)$ has the same structure as the projective cover of $S(\l_0)$, it follows that $z_{\F}(S(\l_1))=2$. The minimal projective resolution of $S(\l_2)$ is given by:
\begin{align}\label{projres1}
 \ldots \rightarrow P_d\rightarrow \ldots \rightarrow P(\l_2)\rightarrow S(\l_2) \rightarrow 0,
\end{align}
where the $d^{th}$ term in this resolution is given by:
\[P_d=\begin{cases}
  P(\l_{d+2})\oplus 2.P(\l_{d})\oplus \cdots\oplus 2.P(\l_2) \quad \text{if $d\geq 2$ is even,}\\
  P(\l_{d+2})\oplus 2.P(\l_{d})\oplus \cdots\oplus 2.P(\l_3)\oplus P(\l_1)\oplus P(\l_0)  \quad \text{if $d\geq 1$ is odd.}\\
  \end{cases}\]
In the above, $2.P(\l)$ means $P(\l)\oplus P(\l)$. Thus the number of summands in $P_d$ is either $d+1$ if $d\geq 2$ is even or it is $d+2$ if $d\geq 1$ is odd. This shows that $z_{\F}(S(\l_l))=2$.

The $d^{th}$ term in the minimal projective resolution of $S(\l_3)$ is given by:
\[P_d=\begin{cases}
  P(\l_{d+3})\oplus P(\l_{d+1})\quad \text{if $d=1$,}\\
   P(\l_{d+3})\oplus P(\l_{d+1})\oplus P(\l_1)\oplus P(\l_0) \quad \text{if $d=2$,}\\
  P(\l_{d+3})\oplus P(\l_{d+1})\oplus 2.P(\l_{d-1})\oplus \cdots\oplus 2.P(\l_2) \quad \text{if $d\geq 3$ is odd,}\\
  P(\l_{d+3})\oplus P(\l_{d+1})\oplus 2.P(\l_{d-1})\oplus \cdots\oplus 2.P(\l_3)\oplus P(\l_1)\oplus P(\l_0)  \quad \text{if $d\geq 4$ is even.}\\
  \end{cases}\]
 Thus the number of summands in $P_d$ is either $d+2$ if $d\geq 2$ is even or it is $d+1$ if $d\geq 1$ is odd. This shows that $z_{\F}(S(\l_l))=2$.\\
For $l\geq 3$, the projective cover of $S(\l_l)$ has the same structure as the projective cover of $S(\l_3)$. This gives  $z_{\F}(S(\l_l))=2$
 \end{proof}

\subsection{Detecting subsuperalgebra}
 Let $\ff_{\1}\subseteq \fg_{\1}$ be the span of the root vectors $x_\alpha, x_{-\alpha}$ where $\alpha=\varepsilon_1+\delta$. In the matrix realization, $x_\alpha=E_{1,4}-E_{5,2}$ and $x_{-\alpha}=E_{2,5}+E_{3,1}$. Set $\ff_{\0}=[\ff_{\1},\ff_{\1}]$. Then $\ff_{\0}$ is spanned by the diagonal matrix $E_{1,1}-E_{2,2}+E_{4,4}-E_{5,5}$. We define a three-dimensional subalgebra of $\fg$ by
\[\ff:=\ff_{\0}\oplus \ff_{\1}.\]
The Lie superalgebra $\ff$ is classical and so has a support variety theory. Furthermore, as $[\ff_{\0},\ff_{\1}]=0$, it follows that these varieties admit a rank variety description and, in particular, can be identified as subvarieties of $\ff_{\1}$, i.e.,
\begin{align*}
\fv_{(\ff,\ff_{\0})}(M)&=\fv_{\ff_{\1}}^{rank}(M)\\
                       &=\{y\in \ff_{\1}\mid \text{M is not projective as $U(\langle y \rangle)$-module}\}\cup\{0\}.
\end{align*}
Note that $\fv_{(\ff,\ff_{\0})}(S(\l_0))=\ff_{\1}$. We can use this detecting subsuperalgebra to give a geometric interpretation of the $z$-complexity. For instance,
\begin{cor}
$z_{\F}(S(\l_0))=\dim \fv_{(\ff,\ff_{\0})}(S(\l_0))=2$.
\end{cor}
In general:
\begin{thm}\ \\
\toto {1} If $S$ is a typical simple module, then
$$z_{\F}(S)=\dim \fv_{(\ff,\ff_{\0})}(S)=0.$$

\toto{2} If $S$ is an atypical simple module, then
$$z_{\F}(S)=\dim \fv_{(\ff,\ff_{\0})}(S)=2.$$

\end{thm}
\begin{proof}
If $S$ is a typical simple module, then $S$ is projective and the first part follows. If $S$ is atypical, then by \cite[Theorem~4.1.1]{kujawa},
$$\fv_{(\ff,\ff_{\0})}(S)=\fv_{(\ff,\ff_{\0})}(S(\l_0)),$$
thus $\dim \fv_{(\ff,\ff_{\0})}(S)=2$. The second part follows.
\end{proof}





\chapter{Computing the complexity over $\dd$}


In this chapter we give the structure of the exceptional Lie superalgebra $\dd$ as discussed in \cite{Musson}. In \cite{Germoni} the author establishes an equivalence between the principal blocks of $\dd$ and $\osp(3|2)$. We use this equivalence to compute the complexity and the $z$-complexity of simple modules over $\dd$. We also give these complexities a geometric interpretation as in the earlier chapters.


 \section{The Lie superalgebra $\dd$}
 Let $\fg$ be the basic classical Lie superalgebra
 $\dd$  (see \cite{Kac1}). Scheunert \cite{Sch} denotes these algebras by $\Gamma(\sigma_1,\sigma_2,\sigma_3)$ where $\sigma_i\in \C$ for $i=1,2,3$. We begin with Scheunert's construction.\\
%It is denoted by $\Ga(-(1+\a)/2,1/2,\a/2)$ in \cite{VJ} and \cite{Zou}. Recall from \cite{VJ} that if $\a=1$, it is isomorphic to $\osp(4,2)$, and that the algebras corresponding to $\a$, $-1-\a^{-1}$ and $\a^{-1}$ are isomorphic (hence if $\a\in\Q$ we can assume $\a>0$, or even $\a>1$).
Let $V$ be a two-dimensional vector space over $\C$ with basis $\{u_1,\,u_{-1}\}$ and let $\psi$ be a non-degenerate skew symmetric bilinear form on $V$ defined by $\psi(u_1,u_{-1})=1$. Then $\mathfrak{sl}(V)=\mathfrak{sp}(\psi)\cong \s_2$ is the algebra of linear transformations on $V$ preserving $\psi$. Now for $i=1,2,3$ take copies $(V_i,\psi_i)$ of $(V,\psi)$ and set
 \[\fg_{\0}\cong\mathfrak{sp}(\psi_1)\oplus \mathfrak{sp}(\psi_2) \oplus \mathfrak{sp}(\psi_3)\cong \s_2\oplus\s_2\oplus\s_2,\]
  \[\fg_{\1}=V_1\boxtimes V_2\boxtimes V_3.\]
  Then $\fg_{\1}$ is a $\fg_{\0}$-module. Define the bilinear map $p_i:V_i\times V_i\rightarrow \mathfrak{sp}(\psi_i)$ by
 \[p_i(x,y)z=\psi_i(y,z)x-\psi_i(z,x)y\]
 for $x,y,z\in V_i$. We then define a symmetric $\fg_{\0}$-invariant bilinear form $p:\fg_{\1}\times \fg_{\1}\rightarrow \fg_{\0}$ by
 \begin{equation}\label{ddbracket}
 p(x_1\otimes x_2 \otimes x_3,y_1\otimes y_2 \otimes y_3)=\sum\sigma_k\psi_i(x_i,y_i)\psi_j(x_j,y_j)p_k(x_k,y_k),
 \end{equation}
 where the sum is over all even permutations of $\{1,2,3\}$.

 \begin{lem}\cite[Lemma~4.2.1]{Musson}\label{ddbracket1}
 $p$ defines a Lie super bracket if and only if $\sigma_1+\sigma_2+\sigma_3=0$.
 \end{lem}

 Thus if $(\sigma_1,\sigma_2,\sigma_3)$ satisfies $\sigma_1+\sigma_2+\sigma_3=0$, then $\fg=\fg_{\0}\oplus \fg_{\1}$ is a Lie superalgebra which we denote by $\Gamma(\sigma_1,\sigma_2,\sigma_3)$. Let $\alpha \in\C\setminus\{0,-1\}$, and set $\dd=\Ga(-(1+\a)/2,1/2,\a/2)$. It can be shown that if $\a=1$, then $\dd$ is isomorphic to $\osp(4|2)$, and that the algebras corresponding to $\a$, $-1-\a^{-1}$ and $\a^{-1}$ are isomorphic (hence if $\a\in\Q$ we can assume $\a>0$, 114 or even $\a>1$).\\
Let
 \[H=\begin{bmatrix}
 1 & 0\\
 0 & -1
 \end{bmatrix},\quad E=\begin{bmatrix}
 0 & 1\\
 0 & 0
 \end{bmatrix},\quad F=\begin{bmatrix}
 0 & 0\\
 1 & 0
 \end{bmatrix}
 \]
 be the usual basis of $\s_2$ and identify $V$ with the vector space of column vectors with basis
 \[u_1=\begin{bmatrix}
 1 \\
 0
 \end{bmatrix},\quad u_{-1}=\begin{bmatrix}
 0 \\
 1
 \end{bmatrix}.\]

 Denote by $H_i,E_i,F_i$ the copies of $H,E,F$ in the $i^{th}$ copy of $\s_2$. Then
 \[p_i(u_1,u_{-1})=-H_i,\,\,p_i(u_1,u_1)=2E_i,\,\,p_i(u_{-1},u_{-1})=-2F_i.\]
The root system of $\fg$ can be realized in the space $\fh^*=\C^3$ endowed with a basis
 $\{\eps_1,\eps_2,\eps_3\}$ and with the bilinear whose matrix is
 $\diag(-(1+\a)/2,1/2,\a/2)$, the roots are given by~:
  $$\Phi_{\0}=\{\pm2\eps_i\tq 1\leq i\leq3\}\quad \text{and}\quad\Phi_{\1}=\{\pm\eps_1\pm\eps_2\pm\eps_3\}.$$
  We choose as simple roots
 $\b_4=\eps_1-\eps_2-\eps_3$, $2\eps_2$, and $2\eps_3$. The non-simple positive odd roots are~:
 $\b_1=\eps_1+\eps_2-\eps_3$, $\b_2=\eps_1-\eps_2+\eps_3$ and $\b_3=\eps_1+\eps_2+\eps_3$.
 We have $\rho=-\b_4$.\\
The corresponding root vectors are
 \[e_1=E_2,\quad e_2=u_1\otimes u_{-1}\otimes u_{-1}, \quad e_3=E_3\]
 and
 \[f_1=F_2,\quad f_2=u_{-1}\otimes u_1\otimes u_1, \quad f_3=F_3.\]
 Set
 \[h_1=H_2,\quad h_2=-\sigma_1H_1+\sigma_2H_2+\sigma_3H_3,\quad h_3=H_3.\]
 These elements satisfy the relations
 \[[e_i,f_j]=\delta_{ij}h_i,\,\, [h_i,h_j]=0,\,\, [h_i,e_j]=a_{ij}e_j,\,\,[h_i,f_j]=-a_{ij}f_j,\]
 where
 \[A=(a_{ij})=\begin{bmatrix}
 2 & -1 & 0\\
 2\sigma_2 & 0 & 2\sigma_3\\
 0 & -1 & 2
 \end{bmatrix}.\]
We identify the set of dominant weights for $\fg_{\0}$  with
 $\N\times\N\times\N$.
 By proposition 2.2 in \cite{Kac1}, the set of dominant weights for $\fg$ is~:
 $X^+=\{(a,b,c)\in\N\times\N\times\N\tq a=0\Rightarrow b=c=0\ \mathrm{and}\
 a=1\Rightarrow (b+1)=\pm \a(c+1)\}$.
 Besides, the eigenvalue
 of the Casimir element is~:
 \[\cas(a,b,c)= -\frac{1+\a}{2}(a-1)^2+\frac{1}{2}(b+1)^2+\frac{\a}{2}(c+1)^2 .\]
A dominant weight $\l=(a,b,c)$ is atypical if and
 only if one of the following scalar products is zero~:
 \[\begin{array}{c}
 2(\l+\rho,\b_1)=-(1+\a)(a-1)+(b+1)-\a(c+1) \\
 2(\l+\rho,\b_2)=-(1+\a)(a-1)-(b+1)+\a(c+1) \\
 2(\l+\rho,\b_3)=-(1+\a)(a-1)+(b+1)+\a(c+1)\\
 2(\l+\rho,\b_4)=-(1+\a)(a-1)-(b+1)-\a(c+1) \\
 \end{array}\]
 Notice that if $\a\not\in\Q$, the only atypical dominant weights are $\l_0=(0,0,0)$
 (corresponding to the trivial module) and $\l_l=(l+1,l-1,l-1)$ for $l\geq1$.
 For example, $\l_1$ is the highest root of $\fg$, {\em i.e.}\ $S(\l_1)$ is the adjoint
 representation.

  If $\a\in\Q$, we assume $\a=p/q$, with $p$ and $q$ relatively prime positive
 integers. For $k\in\N$, let $\Ga_k$ be the set of atypical simple modules
 $\l\in X^+$ such that $\cas(\l)=p(p+q)k^2/2$. For $l\in\Z$ we set moreover~:
 $$\l_{k,l}=\left\{\begin{array}{ccl}
 (-l+2,-l-kp,-l+kq) & \hbox{if}& l\leq-kp ,\\
 (-l+1,l+kp-1,-l+kq-1) & \hbox{if}& -kp+1\leq l\leq0, \blouk\\
 (l+1,l+kp-1,-l+kq-1) & \hbox{if}& 0\leq l\leq kq-1, \\
 (l+2,l+kp,l-kq) & \hbox{if}& kq\leq l.\blouk\\
 \end{array}\right.$$
 \begin{thm}\cite[Theorem~3.1.1]{Germoni}
 Let $\a\in\C\setminus\{-1,0\}$ and let $\fg=\dd$.

 \noi{\em (i)}\ The principal block of $\fg$ is~: $\Ga_0=\{\l_l\tq l\in\N\}$.
 This block is equivalent to the principal block of $\osp(3|2)$~:
 statement {\em (b)} in Theorem~\ref{tru1} holds for the principal
 block of $\dd$.

 \noi{\em (ii)} If $\a\not\in\Q$, $\Ga_0$ is the unique atypical block.

 \noi{\em (iii)} Let $\a\in\Q$, $\a>0$, written as $\a=p/q$ with $p$ and $q$
 relatively prime. Then any atypical block is one
 of the $\Ga_k$. Besides, let $k\geq1$.

 {\em (a)} The map $\Z\rightarrow\Ga_k$, $l\mapsto\l_{k,l}$ is a bijection~;

 {\em (b)} The projective cover of $S(\l_{k,l})$ ($l\in\Z$) has the following radical
 layer structure~:
 \[\raisebox{-3em}{\begin{tikzpicture}

  \node (n1) at (1,1) {$S(\l_{k,l})$};
  \node (n2) at (0,0)  {$S(\l_{k,l-1})$};
  \node (n3) at (2,0)  {$S(\l_{k,l+1}) $};
  \node (n4) at (1,-1) {$S(\l_{k,l})$};



  \foreach \from/\to in {n1/n2,n1/n3, n3/n4, n2/n4}
    \draw (\from) -- (\to);\end{tikzpicture}} \]
\iffalse
 \[ \begin{array}{c}
 S(\l_{k,l})\\
  S(\l_{k,l-1})\ \ S(\l_{k,l+1}) \\
 S(\l_{k,l})
 \end{array} \]
 \fi
 \end{thm}

 Recall that we have the following bounds on the dimensions of the projective covers:
 \begin{equation}\label{eq1:Bdprojcover}
 \dim S_{\0}(\mu)\leq \dim P(\mu)\leq 2^{\dim \fg_{\1}}\dim S_{\0}(\mu), \quad (\mu \in X^+),
 \end{equation}
 where $S_{\0}(\mu)$ is the simple $\fg_{\0}$-module of weight $\mu$. Note that $S_{\0}(\l_0)$ is the trivial $\fg_{\0}$-module and for $\l_l=(l+1,l-l,l)\in X^+$, $S_{\0}(\l_l)=V_{l+1}\otimes V_{l-1}\otimes V_l$ where $V_m$ denotes the simple $\s_2$-module of dimension $m+1$. Thus
\begin{equation}\label{eq1'':Bdprojcover}
(l+2)(l)(l+1)\leq \dim P(\l_l)\leq 2^8 (l+2)(l)(l+1).
\end{equation}

\begin{rk}\label{KacWakconj}
In \cite[Example~3.4]{KW}, the authors compute the super-dimensions of the finite-dimensional simple atypical modules over the exceptional Lie superalgebras. Their computations show that the super-dimension of these modules is not zero which shows that the Kac-Wakimoto conjecture
\[\operatorname{sdim}V=0\Leftrightarrow \atyp(V)< \df(\fg)\]
holds over these Lie superalgebras. Since the conjecture is valid, then \cite[Theorem~4.1.1]{kujawa} holds.
\end{rk}

\section{Complexity of simple $\dd$-modules}
Since the principal block $\Ga_0=\{\l_l\tq l\in\N\}$ is equivalent to the principal block of $\osp(3|2)$ and statement {\em (b)} in Theorem~\ref{tru1} holds for $\Ga_0$, $S(\l_0)$ will have the same minimal projective resolution \eqref{projres1} as in Chapter~4. Then,
\begin{thm}\label{Dsimple}
If $S$ is an atypical simple $\dd$-module, we have $c_{\F}(S)=5.$
\end{thm}

\begin{proof}
Using the projective resolution \eqref{projres1} and the bounds in equation~\eqref{eq1'':Bdprojcover}, we have
\begin{align*}
C'.d^4\leq \dim P_d\leq C.d^4,
\end{align*}
for some positive constants $C,\, C'$. This shows that $c_{\F}(S(\l_0))=5$. Since \cite[Theorem~4.1.1]{kujawa} holds over $\dd$, all simple modules of the same atypicality have the same complexity and the same support variety. Thus the complexity of all atypical simple $\dd$-modules is $5$.
 \end{proof}
 \subsection{A geometric interpretation of the complexity}\label{ss: Dvar}
 The atypical simple module, $S$, over $\dd$ has $\atyp(S)=\df(\dd)=1$. Using the proof of Proposition~\ref{GMosp(3|2)}, we have $ \X_{S}=\X$. Using \cite[Theorem~4.5, Cor~4.8]{DS}, $\dim \X=4$.  By \cite[Theorem~4.1.1]{kujawa} we have
 $$\dim \fv_{(\fg,\fg_{\0})}(S)=\dim \fv_{(\fg,\fg_{\0})}(S(\lambda_0))=1.$$
  \begin{thm}
  Let $S$ be a simple $\dd$-module. Then
 $$c_{\F}(S)=\dim \X_{S}+\dim \fv_{(\fg,\fg_{\0})}(S).$$
\end{thm}
 \section{$z$-complexity}
 The $z$-complexity is a categorical invariant, thus using the equivalence between the principal blocks of $\osp(3|2)$ and $\dd$ we have:
 \begin{prop}
 For all the simple modules $S(\l)$ in the principal block of $\dd$, we have $z_{\F}(S(\l))=2$.
 \end{prop}

 For the simple modules in the other atypical blocks $\Ga_k$, we can compute the $z$-complexity by writing an explicit minimal projective resolution. We then have:
 \begin{thm}
 For $\l_{k,l}\in \Ga_k$, we have $z_{\F}(S(\l_{k,l}))=2$.
 \end{thm}
 \begin{proof}
 The $d^{th}$ term in the minimal projective resolution of $S(\l_{k,l})$ is given by:
\[P_d=\begin{cases}
   P(\l_{k,l\pm d})\oplus P(\l_{k,l\pm (d-2)})\oplus \cdots\oplus P(\l_{k,l\pm 1}) \quad \text{if $d\geq 1$ is odd,}\\
   P(\l_{k,l\pm d})\oplus P(\l_{k,l\pm (d-2)})\oplus \cdots\oplus P(\l_{k,l})  \quad \text{if $d\geq 2$ is even.}\\
   \end{cases}\]
 In the above, $P(\l_{k,l\pm i})$ means $P(\l_{k,l+i})\oplus P(\l_{k,l-i})$. Thus the number of summands in $P_d$ is $d+1$. This shows that $z_{\F}(S(\l_{k,l}))=2$.
 \end{proof}

 \subsection{Detecting subsuperalgebra}\label{ss:Ddetecting}
  Let $\ff_{\1}\subseteq \fg_{\1}$ be the span of the root vectors $x_\alpha, x_{-\alpha}$ where $\alpha=\varepsilon_1+\varepsilon_2+\varepsilon_3$. Set $\ff_{\0}=[\ff_{\1},\ff_{\1}]$.  We define a subalgebra of $\fg$ by
\[\ff:=\ff_{\0}\oplus \ff_{\1}.\]
The Lie superalgebra $\ff$ is classical and so has a support variety theory. Furthermore,
\begin{lem}
$\ff$ is three-dimensional. Moreover, $[\ff_{\0},\ff_{\1}]=0$.
 \end{lem}
 \begin{proof}
 The root vectors $x_\a$ and $x_{-\a}$ are in fact given by
 \[x_\a=u_1\otimes u_1\otimes u_1\quad x_{-\a}=u_{-1}\otimes u_{\1}\otimes u_{-1},\]
 Then by equation~\eqref{ddbracket} we have
 \[[x_\a,x_{-\a}]=\sum \sigma_k \psi_i(u_i,u_{-i})\psi_j(u_j,u_{-j})p_k(u_k,u_{-k})\]
 where the sum is over the even permutations of $\{1,2,3\}$. We have
 \[\psi_i(u_i,u_{-i})=\psi_j(u_j,u_{-j})=1\]
 %the even permutations are (1) (1 2 3) and (1 3 2)
 thus
 \begin{align*}
 [x_\a,x_{-\a}] & =\sigma_1p_1(u_1,u_{-1})+\sigma_2p_2(u_1,u_{-1})+\sigma_3p_3(u_1,u_{-1})\\
                & = \sigma_1(-H_1)+\sigma_2(-H_2)+\sigma_3(-H_3)\\
                & = -(\sigma_1H_1+\sigma_2H_2+\sigma_3H_3)\in \fg_{\0}.
 \end{align*}
   Having $[x_\a,x_{\a}]=[x_{-\a},x_{-\a}]=0$ shows that $\ff$ is three-dimensional. Moreover,
   \begin{align*}
   [[x_\a,x_{-\a}],x_\a] & =-[\sigma_1H_1+\sigma_2H_2+\sigma_3H_3, u_1\otimes u_1\otimes u_1]\\
                         & = -(\sigma_1u_1\otimes u_1\otimes u_1+\sigma_2u_1\otimes u_1\otimes u_1+\sigma_3u_1\otimes u_1\otimes u_1)\\
                         & = -(\sigma_1+\sigma_2+\sigma_3)u_1\otimes u_1\otimes u_1\\
                         & = 0.
 \end{align*}
 The last equality follows from the definition of $\dd$ (Lemma~\ref{ddbracket1}). Similarly we can show
 \[[[x_\a,x_{-\a}],x_{-\a}]=0.\]
 This shows that $[\ff_{\0},\ff_{\1}]=0$.
 \end{proof}

 It follows that these varieties admit a rank variety description and, in particular, can be identified as subvarieties of $\ff_{\1}$, i.e.,
\begin{align*}
\fv_{(\ff,\ff_{\0})}(M)&=\fv_{\ff_{\1}}^{rank}(M)\\
                       &=\{y\in \ff_{\1}\mid \text{M is not projective as $U(\langle y \rangle)$-module}\}\cup\{0\}.
\end{align*}
Note that $\fv_{(\ff,\ff_{\0})}(S(\l_0))=\ff_{\1}$. We can use this detecting subsuperalgebra to give a geometric interpretation of the $z$-complexity. For instance,
\begin{cor}
$z_{\F}(S(\l_0))=\dim \fv_{(\ff,\ff_{\0})}(S(\l_0))=2$.
\end{cor}
In general:
\begin{thm}\ \\
\toto {1} If $S$ is a typical simple module, then
$$z_{\F}(S)=\dim \fv_{(\ff,\ff_{\0})}(S)=0.$$

\toto{2} If $S$ is an atypical simple module, then
$$z_{\F}(S)=\dim \fv_{(\ff,\ff_{\0})}(S)=2.$$

\end{thm}
\begin{proof}
If $S$ is a typical simple module, then $S$ is projective and the first part follows. If $S$ is atypical, then by \cite[Theorem~4.1.1]{kujawa},
$$\fv_{(\ff,\ff_{\0})}(S)=\fv_{(\ff,\ff_{\0})}(S(\l_0)),$$
thus $\dim \fv_{(\ff,\ff_{\0})}(S)=2$. The second part follows.
\end{proof}



\chapter{Computing the complexity over $G(3)$}
In this chapter we calculate the complexity of simple modules over the exceptional Lie superalgebra $G(3)$. We introduce the structure of $G(3)$ as given in \cite{Musson}. We use the characterization of the simple modules and their projective covers given in \cite{Germoni} to find the complexity and the $z$-complexity. We then verify the geometric interpretation of these complexities in this case.
\section{The Lie superalgebra $G(3)$}

Let $\fg$ be the basic classical Lie superalgebra
$G(3)$  (see \cite{Kac1}). We will consider the construction introduced in \cite[Section~4.4]{Musson}. Let $\mathbb{H}=\R\oplus \R i\oplus \R j\oplus \R k$ be the quaternion algebra over $\R$, with $i^2=j^2=k^2=-1$ and $ij=k,\,jk=i,\,ki=j$. If $x=ai+bj+ck+dk$, we set $\bar{x}=a-bj-cj-dk$. Let $\mathbb{O}=\mathbb{H}\oplus \mathbb{H}$ be the octonion algebra with multiplication
\[(a,b)(c,d)=(ac-\bar{d}b,da+b\bar{c}),\]
for $a,b,c,d\in \mathbb{H}$. For $x=(a,b)\in \mathbb{O}$, define $\bar{x}=(\bar{a},-b)$ and the trace of $x$ by $t(x)=x+\bar{x}=(a+\bar{a},0)\in \R$. Set $\mathbb{O}_{(0)}=\{x\in \mathbb{O}\,|\,t(x)=0\}$, the space of traceless octonions. A derivation $D$ on $\mathbb{O}$ is a linear map satisfying
\[D(xy)=D(x)y+xD(y).\]
It is easy to check that $D(\mathbb{O})\subseteq \mathbb{O}_{(0)}$. Let $\operatorname{Der}\mathbb{O}$ be the set of all derivations on $\mathbb{O}$. We define the associator $(x,y,z)$ on $\mathbb{O}$ by
\[(x,y,x)=(xy)z-x(yz),\]
and the commutator
\[[x,y]=xy-yx.\]
For $x,y,z\in \mathbb{O}$, set $D_{x,z}(y)=[y,[x,z]]-3(x,y,z)$. Now we can give an explicit construction of $G(3)$~:
\begin{thm}\cite[Theorem~4.4.5]{Musson}\label{G(3)}
Let $V_1$ be a two-dimensional vector space over $\C$. Let $\psi_1:V_1\times V_1\rightarrow \C$ be a nondegenerate skew symmetric bilinear form defined by $\psi_1(u_1,u_{-1})=1$ if we let $\{u_1,u_{-1}\}$ as a basis of $V_1$. Define $p_1:V_1\times V_1\rightarrow \s_2$ by
\[p_1(x,y)(z)=4\{\psi_1(y,z)x-\psi_1(z,x)y\}\]
for $x,y,z\in V_1$.

Let $G_2=\operatorname{Der}\mathbb{O}\otimes_{\R} \C$, $V_2=\mathbb{O}_{(0)}\otimes_{\R} \C$ and let $\psi_2$ be the bilinear map $\psi_2:V_2\times V_2\rightarrow \C$ given by $\psi_2(x,y)=t(x\bar{y})/2$. Define $p_2:V_2\times V_2\rightarrow G_2$ by $p_2(x,y)=D_{x,y}$.

Let $\fg_{\0}=\s_2\oplus G_2$ and $\fg_{\1}=V_1\boxtimes V_2$. Then $\fg=\fg_{\0}\oplus \fg_{\1}$ becomes a Lie superalgebra when we define
\[[x_1\otimes x_2, y_1\otimes y_2 ]= \psi_1(x_1,y_1)p_2(x_2,y_2)+\psi_2(x_2,y_2)p_1(x_1,y_1).\]
\end{thm}

Consider the following basis of $\mathbb{O}_{(0)}$:
\begin{equation*}
\begin{split}
v_3=(0,-i)&,\, v_2=(0,-j),\,v_1=(0,-k),\,v_0=(0,1)\\
&v_{-1}=(k,0),\,v_{-2}=(j,0),\,v_{-3}=(i,0).
\end{split}
\end{equation*}
Together with the identity $1=(1,0)$ these elements form a basis for $\mathbb{O}_{(0)}$. We will write $D_{j,k}$ for $D_{v_j,v_k}$. Define a basis for $V_2=\mathbb{O}_{(0)}\otimes_{\R} \C$ by
\[e_3=v_3+iv_{-3},\,e_2=v_1-iv_{-1},\,e_1=v_2-iv_{-2},\,e_0=v_0,\]
\[e_{-1}=v_2+iv_{-2},\,e_{-2}=v_1+iv_{-1},\,e_{-3}=v_3-iv_{-3}.\]
Using this basis we identify $G_2$ with its image in $\gl_7(\C)$. The following computations are taken from \cite[Section~4.7]{Musson}~:
\begin{thm}\label{G(3)structure}\
\begin{enumerate}
\item Using the basis $\{e_3,\ldots,e_{-3}\}$, the derivations $D_{1,-1}$ and $D_{2,-2}$  are represented by diagonal matrices with entries
\[(-2i,-4i,2i,0,-2i,4i,2i)\quad\text{and}\quad (-2i,2i,-4i,0,4i,-2i,2i)\]
respectively.
\item $D_{1,-1}+D_{2,-2}+D_{3,-3}=0$.
\item Set $h_1=iD_{2,-2}/2$ and $h_2=i(D_{1,-1}-D_{2,-2})/2$. Then $\{h_1,h_2\}$ is a basis for a Cartan subalgebra of $G_2$.
\item Define $\a_1,\a_2$ in the dual of this Cartan subalgebra by $\a_j(h_k)=a_{jk}$ where $A=(a_{jk})=\begin{bmatrix} 2 & -3\\ -3 & 6 \end{bmatrix}$.
Then $e_{\pm3},\,e_{\pm2},\,e_{\pm 1},\,e_0$ are weight vectors of weights $\pm(2\a_1+\a_2),\,\pm(\a_1+\a_2),\,\pm \a_1,$ and $0$ respectively. As a set of positive roots of $G_2$ we can take $\{\a_1,\a_2,\a_1+\a_2,2\a_1+\a_2,3\a_1+\a_2,3\a_1+2\a_2\}$.
\item The derivation $\Delta_1=D_{1,3}-iD_{1,-3}-iD_{-1,3}-D_{-1,-3}$ is represented by the matrix $x_1:=-4E_{1,2}+4E_{3,4}-8E_{4,5}+4E_{6,7}$.
\item The derivation $\Delta_2=D_{1,2}-iD_{1,-2}-iD_{-1,2}-D_{-1,-2}$ is represented by the matrix $x_1:=12E_{2,3}-12E_{5,6}$.
\item For every weight $\mu$ of $V_2$ we have $x_j.V^{\mu}\subseteq V^{\mu+\a_j}$ and hence $x_j\in G_2^{\a_j}$ for $j=1,2$.
\item The remaining generators of $G_2$ as a Lie algebra are:
\[y_1=-k(x_1)/2\quad \text{and}\quad y_2=-k(x_2)/48,\]
where $K(x_i)$ is the complex conjugate of $x_i$. Then $y_j\in G_2^{-\a_j}$ for $j=1,2$.
\item For $j=1,2$ and $h$ in the Cartan subalgebra of $G_2$ we have
\[[h_i,h_j]=0,\,[x_i,y_j]=\delta_{ij}h_i,\,[h,x_i]=\a_i(h)x_i,\,[h,y_i]=-\a_i(h)y_i.\]
\item Let $H=\begin{bmatrix}
1 & 0\\
0 & -1
\end{bmatrix}$
and consider the basis $\{u_1,u_{-1}\}$ for $V_1$. In addition to the generators $x_i,y_i,h_i$ for $i=1,2$, $G(3)$ has generators
\[x_0=(u_1\otimes e_{-3})/2,\quad y_0=(u_{-1}\otimes e_3)/2,\quad \text{and}\quad h_0=-2H-2h_1-h_2.\]
\item Let $\psi_1,\,\psi_2,\,p_1,\,p_2$ be as defined in Theorem~\ref{G(3)}. Then
 \[p_1(u_1,u_{-1})=-4H,\quad p_2(e_{-3},e_3)=-8h_1-4h_2,\quad \psi_2(e_{-3},e_3)=2.\]
 \item Let $\fh$ be the Cartan subalgebra of $G(3)$ with basis $\{h_0,h_1,h_2\}$ and define $\a_j\in \fh^*$ by $\a_j(h_k)=a_{jk}$ where
 \[A=(a_{jk})=\begin{bmatrix}
  0 & -1 & 0\\ -1 & 2 & -3 \\ 0 & -3 & 6\end{bmatrix}.\]
Then for $j=0,1,2$ and $h\in \fh$ we have
\[[h_i,h_j]=0,\,[x_i,y_j]=\delta_{ij}h_i,\,[h,x_i]=\a_i(h)x_i,\,[h,y_i]=-\a_i(h)y_i.\]
\item Let $2\d$ be the positive root of $\s_2$ and set $\a_0=\d-2\a_1-\a_2$. Then $\{\a_0,\a_1,\a_2\}$ is a system of simple roots for $G(3)$.
\end{enumerate}
\end{thm}

If we set
\[\a_0=\d+\e_3,\quad \a_1=\e_1,\quad \a_2=\e_2-\e_1,\]
the root system in Theorem~\ref{G(3)structure} can be realized in the space $\h^*=\C^3$ endowed with
a basis $\{\d,\eps_1,\eps_2\}$ and with the bilinear defined by~:
\[(\e_1,\e_1)=(\e_2,\e_2)=-2(\e_1,\e_2)=-(\d,\d)=2.\]

We can then take as simple roots $\e_1$, $\e_2-\e_1$ and $-\e_1-\e_2+\d$.
Then positive even roots are~:
$\Phi_{\0}^+=\{\e_1, 2\e_1+\e_2,\e_1+\e_2,\e_1+2\e_2,\e_2,\e_2-\e_1,2\d\}$
and positive odd roots are~:
$\Phi_{\1}^+=\{(\pm(u\e_1+v\e_2)+\d\tq (u,v)\in\{0,1\}^2\}$.
We have $\rho=2\e_1+3\e_2-5\d/2$. We identify the set of dominant weights for $\fg_{\0}$ with
$\N\times\N\times\N$ by means of the fundamental
weights $(\w_1=\e_1+\e_2;\w_2=\e_1+2\e_2;\w_3=\d)$.
For $\lambda=m_1\omega_1+m_2\omega_2$, the dimension of the simple $G(2)$-module, $L(\lambda)$, can be computed by the Weyl-dimension formula \cite[Section~24.3]{HUM}:
\begin{multline}\label{Gdimformula}
\dim L(\lambda)=\frac{1}{5!}(m_1+1)(m_2+1)(m_1+m_2+2)(m_1+2m_2+3)\\
(m_1+3m_2+4)(m_1+3m_2+4)(2m_1+3m_2+5).
\end{multline}
By proposition 2.2 in \cite{Kac1}, the set of dominant weights for $\fg$ is~:
$X^+=\{(a,b,c)\in\N\times\N\times\N\tq a=0\Rightarrow b=c=0,\ a\neq1\ \mathrm{and}\
a=2\Rightarrow b=0\}$.
Besides, the eigenvalue
of the Casimir element is~:
\[\cas(a,b,c)= 2b^2+6bc+10b+6c^2+18c-2a^2+10a.\]
For $k\in\N$, we denote by $\Ga_k$ the set of dominant weights $\l\in X^+$ such that
$\cas(\l)=6k(k+1)$. For $l\in\N$, we set~:
\[\begin{array}{ccl}
\l_{0,0} &=& (0,0,0),\\
\l_{0,1} &=& (5,0,0), \\
\l_{k,0} &=& (2,0,k-1) \quad \mathrm{if}\ k\geq1,\\
\l_{k,1} &=& (3,0,k-1) \quad \mathrm{if}\ k\geq1,\\
\l_{k,l} &=& \left\{ \begin{array}{ll}
(l+2,2l-2,k-l) & \mathrm{if}\ 2\leq l\leq k,\\
(l+3,3k-l,l-k-1) & \mathrm{if}\ k+1\leq l\leq 3k,\\
(l+4,l-3k-1,2k) & \mathrm{if}\ 3k+1\leq l.
\end{array}\right.
\end{array} \]

\begin{thm}(\cite[Theorem~4.1.1]{Germoni})\label{tru3}
Let $\fg=G(3)$.
Every atypical block of $\fg$ is one of the $\Ga_k$. For every $k\in\N$, $\Ga_k$ is
equivalent to the principal block of $\osp(3,2)$.

The map $\N\rightarrow\Ga_k$, $l\mapsto\l_{k,l}$ is bijective~; statements {\em (b)} and
{\em (c)} of Theorem~\ref{tru1} apply to $\Ga_k$.
\end{thm}

\section{Complexity of simple $G(3)$-modules}
Since the principal block $\Ga_0=\{\l_{0,l}\tq l\in\N\}$ is equivalent to the principal block of $\osp(3|2)$ and statement {\em (b)} in Theorem~\ref{tru1} holds for $\Ga_0$ (\cite[Lemma~4.3.1]{Germoni}), we will have the same minimal projective resolution \eqref{projres1} for $S(\l_{0,0})$. Then
\begin{thm}\label{Gsimple}
If $S$ is an atypical simple $G(3)$-module, we have $c_{\F}(S)=8.$
\end{thm}

\begin{proof}
Using the projective resolution \eqref{projres1}, the bounds in equation~\eqref{eq1:Bdprojcover}, and the dimension formula given in \eqref{Gdimformula}, we have
\begin{align*}
C'.d^7\leq \dim P_d\leq C.d^7,
\end{align*}
for some positive constants $C,\, C'$. These bounds are obtained by multiplying the dimension formula given in \eqref{Gdimformula} by an extra factor from the $\s_2$-part. This shows that $c_{\F}(S(\l_0))=8$. Since \cite[Theorem~4.1.1]{kujawa} holds over $G(3)$, then all simple modules of the same atypicality have the same complexity and the same support variety. Thus the complexity of all atypical simple $G(3)$-modules is $8$.
\end{proof}

\subsection{A geometric interpretation of the complexity}
The same discussion as in \ref{ss: Dvar} shows that we have the following geometric interpretation in this case: If $S$ is a simple $G(3)$-module, then
$$c_{\F}(S)=\dim \X_{S}+\dim \fv_{(\fg,\fg_{\0})}(S).$$
%Using \cite[Theorem~4.5, Cor~4.8]{DS}, $\dim \X_{S(\l_0)}=\dim \X=4$. On the other hand, by \cite[Cor~4.4.2]{kujawa} we have $\dim \fv_{(\fg,\fg_{\0})}(S(\lambda_0))=\atyp(\l_0)=1$. Thus
%$$c_{\F}(S(\lambda_0)=\dim \X_{S(\lambda_0)}+\dim \fv_{(\fg,\fg_{\0})}(S(\lambda_0)).$$

\section{$z$-complexity}
The $z$-complexity is a categorical invariant, thus using the equivalence between the principal block of $\osp(3|2)$ and the atypical blocks $G(3)$ we have:
\begin{prop}
If $S$ is an atypical simple $G(3)$-module, then $z_{\F}(S)=2$.
\end{prop}

\subsection{Detecting subsuperalgebra}
 Let $\ff_{\1}\subseteq \fg_{\1}$ be the span of the root vectors $x_0=x_{\alpha_0},y_0= x_{-\alpha_0}$ where $\alpha_0=\varepsilon_3+\delta$. Set $\ff_{\0}=[\ff_{\1},\ff_{\1}]$. We define a three-dimensional subalgebra of $\fg$ by
\[\ff:=\ff_{\0}\oplus \ff_{\1}.\]
The Lie superalgebra $\ff$ is classical and so has a support variety theory. Furthermore,

 \begin{lem}
 $[\ff_{\0},\ff_{\1}]=0$.
  \end{lem}

  \begin{proof}
The even part $\ff_{\0}$ is spanned by the vector $[x_0,y_0]=h_0$. From Theorem~\ref{G(3)structure} we have $[h_0,x_0]=\a_0(h_0)x_0=0$ and $[h_0,y_0]=\a_0(h_0)y_0=0$. The result follows.
  \end{proof}

 It follows that these varieties admit a rank variety description and, in particular, can be identified as subvarieties of $\ff_{\1}$, i.e.,
\begin{align*}
\fv_{(\ff,\ff_{\0})}(M)&=\fv_{\ff_{\1}}^{rank}(M)\\
                       &=\{y\in \ff_{\1}\mid \text{M is not projective as $U(\langle y \rangle)$-module}\}\cup\{0\}.
\end{align*}
Note that $\fv_{(\ff,\ff_{\0})}(S(\l_{0,0}))=\ff_{\1}$. We can use this detecting subsuperalgebra to give a geometric interpretation of the $z$-complexity. For instance,
\begin{cor}
$z_{\F}(S(\l_{0,0}))=\dim \fv_{(\ff,\ff_{\0})}(S(\l_{0,0}))=2$.
\end{cor}
In general:
\begin{thm}\ \\
\toto {1} If $S$ is a typical simple module, then
$$z_{\F}(S)=\dim \fv_{(\ff,\ff_{\0})}(S)=0.$$

\toto{2} If $S$ is an atypical simple module, then
$$z_{\F}(S)=\dim \fv_{(\ff,\ff_{\0})}(S)=2.$$

\end{thm}
\begin{proof}
If $S$ is a typical simple module, then $S$ is projective and the first part follows. If $S$ is atypical, then by \cite[Theorem~4.1.1]{kujawa},
$$\fv_{(\ff,\ff_{\0})}(S)=\fv_{(\ff,\ff_{\0})}(S(\l_{0,0})),$$
thus $\dim \fv_{(\ff,\ff_{\0})}(S)=2$. The second part follows.
\end{proof}


\chapter{Computing the complexity over $F(4)$}

In this chapter we calculate the complexity of simple modules over the exceptional Lie superalgebra $F(4)$. We introduce the structure of $F(4)$ as given in \cite{Musson}. We use the characterization of the simple modules and their projective covers given in \cite{Mart} to find the complexity and the $z$-complexity. The geometric interpretation of these complexities in this case agrees with our earlier results.

\section{The Lie superalgebra $F(4)$}
The construction of the classical Lie superalgebra $F(4)$ involves the use of Clifford algebras and spinor representations.
\begin{dfn}
Let $\C^{2n}$ be the vector space of column vectors with $2n$ entries and let $\beta$ be the bilinear form on $\C^{2n}$ defined by $\beta(x,y)=x^ty$. Let $T(\C^{2n})$ be the tensor algebra on $\C^{2n}$ and $J$ the ideal of this tensor algebra generated by the elements
 \[xy+yx-\beta(x,y)\]
 for $x,y\in \C^{2n}$. The Clifford algebra associated to $\beta$ is $C(\beta)~:~=T(\C^{2n}/J$.
 \end{dfn}
 \begin{prop}\cite[Proposition~A.3.5]{Musson}
 If $\{x_1,\ldots,x_{2n}\}$ is a basis for $\C^{2n}$ then $C(\beta)$ has a basis consisting of the monomials $x_1^{r_1}x_2^{r_2}\cdots x_{2n}^{r_{2n}}$ where $r_i=0$ or $1$ for $i=1,2,\ldots, 2n$.
 \end{prop}

 \begin{thm}\cite[Theorems~A.3.7 and A.3.8]{Musson}
 $C(\beta)$ has a unique simple module $S$ with $\dim_{\C} S= 2^n$. $S$ is called the \textbf{space of spinors}.
 \end{thm}

 Let $R~:~ C(\beta) \rightarrow \End(S)$ be the corresponding representation. Set $\Ga_i=R(x_i)$ for $1\leq i\leq 2n$. Then
 \begin{equation}\label{spinor1}
 \Ga_k\Ga_+\Ga_j\Ga_k=2\d_{jk}
 \end{equation}
 for $1\leq j,k\leq 2n$. If we set $\Ga_{2n+1}=i^{n}\Ga_1\cdots \Ga_{2n}$ then equation~\eqref{spinor1} holds for $1\leq j,k\leq 2n+1$.

 Define $\Ga_j^t\in \End S^*$ by $(\Ga_j^tf)(v)=f(\Ga_jv)$, for $f\in S^*,\,v\in S$. The relation \eqref{spinor1} also holds if we replace the elements $\Ga_j\in \End S$ by $(-1)^n\Ga_j^t\in \End S^*$. By the uniqueness of $S$ there is an isomorphism $f~:~S\rightarrow S^*$ such that
 \[(-1)^n\Ga_j^t\circ f-f\circ \Ga_j\]
 for $1\leq j\leq 2n$. Define a nondegenerate bilinear form $\psi$ on $S$ by $\psi(u,v)=(f(u))(v)$. Then
 \begin{equation}\label{spinor2}
 \psi(\Ga_ju,v)=(-1)^n\psi(u,\Ga_jv)
 \end{equation}
 for all $u,v\in S$ and $ 1\leq j\leq 2n+1$. Let $\mathfrak{so}(2n+1)$ be the Lie algebra of all skew-symmetric $2n+1\times 2n+1$ matrices. The spin representation of $\mathfrak{so}(2n+1)$ is the Lie algebra homomorphism $\mathfrak{so}(2n+1)\rightarrow \End S$ sending $E_{jk}-E_{kj}$ to $\frac{1}{2}\Ga_j\Ga_k$ when $j\neq k$. We identify $\mathfrak{so}(2n+1)$ with its image in $\End S$. Then $S$ is a simple $\mathfrak{so}(2n+1)$-module and $\psi$ is an $\mathfrak{so}(2n+1)$-invariant bilinear form on $S$. It is known that $\psi$ is symmetric if $n\equiv 0,3\mod 4$ and it is skew-symmetric otherwise.

 The following theorem gives an explicit construction of $F(4)$~:
 \begin{thm}\cite[Theorem~4.5.5]{Musson}\label{F(4)}
 Let $V_1$ be a two-dimensional vector space over $\C$. Let $\psi_1:V_1\times V_1\rightarrow \C$ be a nondegenerate skew symmetric bilinear form defined by $\psi_1(u_1,u_{-1})=1$ if we let $\{u_1,u_{-1}\}$ as a basis of $V_1$. Define $p_1:V_1\times V_1\rightarrow \s_2$ by
\[p_1(x,y)(z)=3\{\psi_1(y,z)x-\psi_1(z,x)y\}\]
for $x,y,z\in V_1$.

Let $V_2=S$ be the spin representation of $\mathfrak{so}_7$ and let $\psi_2:V_2\times V_2\rightarrow \C$ be a nondegenerate bilinear form such that $ \psi_2(\Ga_ju,v)=-\psi_2(u,\Ga_jv)$ for $u,v\in V_2$. Identify $\mathfrak{so}_7$ with its image in $\End S$ and define $p_2:V_2\times V_2\rightarrow \mathfrak{so}_7$ by
 \[p_2(u,v)=\sum_{1\leq j< k\leq 7}\psi_2(u,\Ga_j\Ga_kv)\Ga_j\Ga_k.\]

Let $\fg_{\0}=\s_2\oplus \mathfrak{so}_7$ and $\fg_{\1}=V_1\boxtimes V_2$. Then $\fg=\fg_{\0}\oplus \fg_{\1}$ becomes a Lie superalgebra when we define
\[[x_1\otimes x_2, y_1\otimes y_2 ]= \psi_1(x_1,y_1)p_2(x_2,y_2)+\psi_2(x_2,y_2)p_1(x_1,y_1).\]
\end{thm}

The root system can be realized in the space $\h^*=\C^4$ endowed with
a basis $\{\d,\eps_1,\eps_2,\eps_3\}$ and with the bilinear defined by~:
$(\e_i,\e_j)=\d_{ij},\,\,(\d,\d)=-3,\,\,(\e_i,\d)=0\,\,\text{for all $i,\,j$}$.
We take as simple roots $\alpha_1=\frac{1}{2}(-\e_1-\e_2-\e_3+\d)$, $\alpha_2=\e_3$, $\alpha_3=-\e_2-\e_3$, and $\alpha_4=\e_1-\e_2$.
Then positive even roots are~:
$\Phi_{\0}^+=\{\d,\e_i, \e_i\pm\e_j\,|\, i<j\}$
and positive odd roots are~:
$\Phi_{\1}^+=\{\frac{1}{2}(\pm\e_1\pm\e_2\pm\e_3+\d)\}$.
We have $\rho=\frac{1}{2}(5\e_1+3\e_2+\e_3-3\d)$.

The even part of $\fg$ is $\fg_{\0}\cong\s_2\oplus \mathfrak{so}_7$ of type $A_1\oplus B_3$. The set of dominant weights and the atypical blocks are described in \cite{Mart}. The maximal atypicality is $1$ as the defect of $F(4)$ is $1$. The fundamental weights are $\omega_1=\e_1,\,\omega_2=\e_1+\e_2,\,\omega_3=\frac{1}{2}(\e_1+\e_2+\e_3),$ and $\omega_4=\frac{1}{2}\delta$. For $\lambda=m_1\omega_1+m_2\omega_2+m_3\omega_3$, the dimension of the simple $\mathfrak{so}(7)$-module, $L(\lambda)$, can be computed by the Weyl-dimension formula \cite[section~24.3]{HUM}:
\begin{multline}\label{Fdimformula}
\dim L(\lambda)=\frac{1}{720}(m_1+1)(m_2+1)(m_3+1)(m_1+m_2+2)(m_2+m_3+2)\\
(2m_2+m_3+3)(m_1+m_2+m_3+3)(m_1+2m_2+m_3+4)(2m_1+2m_2+m_3+5).
\end{multline}

Note that the spin representation of $\mathfrak{so}_7$ has the following explicit construction. Let $W$ be a three-dimensional vector space over $\C$, then $\bigwedge^\bullet W$ is an eight-dimensional vector space over $\C$~:

\begin{prop} \cite[prop~20.20]{FH}
The vector space $\bigwedge^\bullet W$ is the irreducible representation of $\mathfrak{so}_7$ with highest weight $\frac{1}{2}(\e_1+\e_2+\e_3)$. Moreover, $e_I=e_{i_1} \wedge e_{i_2}\wedge \ldots \wedge e_{i_k}\in \bigwedge^\bullet W$ has the weight $\frac{1}{2}(\sum_{i\in I} \e_i - \sum_{j\notin I} \e_j)$.
\end{prop}

\section{Complexity of simple $F(4)$-modules}

The description of the projective indecomposable modules over $F(4)$ given in \cite[Lemma~11.1]{Mart} is the same as the one given in \cite{Germoni} over $G(3)$, except for a small difference in the notation. For example, in \cite{Mart}, $\l_1$ corresponds to the trivial module, while in \cite{Germoni} $\l_0$ corresponds to the trivial module. This similarity means that the projective resolution \eqref{projres1} over $\osp(3|2)$ will carry over to $F(4)$.

\begin{thm}\label{Fsimple}
If $S$ is an atypical simple $F(4)$-module, we have $c_{\F}(S)=9.$
\end{thm}

\begin{proof}
Using the projective resolution \eqref{projres1} and the bounds in equation~\eqref{eq1:Bdprojcover}, and the dimension formula in equation~\eqref{Fdimformula}, we have
\begin{align*}
C'.d^8\leq \dim P_d\leq C.d^8,
\end{align*}
for some positive constants $C,\, C'$. These bounds are obtained by multiplying the dimension formula in equation~\eqref{Fdimformula} by an extra factor from the $\s_2$-part. This shows that $c_{\F}(S(\l_1))=9$. Since \cite[Theorem~4.1.1]{kujawa} holds over $F(4)$, then all simple modules of the same atypicality have the same complexity and the same support variety. Thus the complexity of all atypical simple $F(4)$-modules is $9$.
\end{proof}

\subsection{A geometric interpretation of the complexity}
The same discussion as in \ref{ss: Dvar} shows that we have the following geometric interpretation in this case: If $S$ is an atypical simple $F(4)$-module, then
$$c_{\F}(S)=\dim \X_{S}+\dim \fv_{(\fg,\fg_{\0})}(S).$$
%Using \cite[Theorem~4.5, Cor~4.8]{DS}, $\dim \X_{S(\l_0)}=\dim \X=4$. On the other hand, by \cite[Cor~4.4.2]{kujawa} we have $\dim \fv_{(\fg,\fg_{\0})}(S(\lambda_0))=\atyp(\l_0)=1$. Thus
%$$c_{\F}(S(\lambda_0)=\dim \X_{S(\lambda_0)}+\dim \fv_{(\fg,\fg_{\0})}(S(\lambda_0)).$$

\section{$z$-complexity}
The radical layer structure of the projective indecomposable modules over $F(4)$ is the same as that over $G(3)$. This will give the same projective resolutions over $F(4)$. Thus
\begin{prop}
If $S$ is an atypical simple $F(4)$-module, then $z_{\F}(S)=2$.
\end{prop}

\subsection{Detecting subsuperalgebra}
 Let $\ff_{\1}\subseteq \fg_{\1}$ be the span of the root vectors $x_\alpha, x_{-\alpha}$ where $\alpha=1/2(\e_1+\e_2+\e_3+\delta)$. Set $\ff_{\0}=[\ff_{\1},\ff_{\1}]$. We define a subalgebra of $\fg$ by
\[\ff:=\ff_{\0}\oplus \ff_{\1}.\]
The Lie superalgebra $\ff$ is classical and so has a support variety theory. Furthermore,
 \begin{lem}
 $\ff$ is three-dimensional and satisfies $[\ff_{\0},\ff_{\1}]=0$.
 \end{lem}
 \begin{proof}
 The root vectors $x_\a$ and $x_{-\a}$ are given by
 \[x_\a=u_1\otimes (e_1\wedge e_2\wedge e_3), \quad x_{-\a}=u_{-1}\otimes 1,\]
 where $\{u_1,u_{-1}\}$ is a basis of $V_1$ and $\{e_1,e_2,e_3\}$ is a basis of $W$, with $\bigwedge^\bullet W=V_2$. Recall that we have $\psi_1(u_1,u_{-1})=1$ in Theorem~\ref{F(4)}. We can easily show that
 \[p_1(u_1,u_{-1})=\begin{bmatrix} -3 & 0\\ 0 & 3\end{bmatrix}\in \s_2.\]
 Since $e_1\wedge e_2\wedge e_3$ and $1$ have opposite weights, %and $\psi_2:V_2\otimes V_2\otimes V_2\rightarrow \C$ is $\mathfrak{so}_7$-invariant,
 then $\psi_2(e_1\wedge e_2\wedge e_3, 1)\neq 0$. Choose $\psi_2$ to be normalized in the sense that $\psi_2(e_1\wedge e_2\wedge e_3, 1)=1$. To compute the bracket $[x_\a,x_{-\a}]$ we need to find all the elements, $\Ga_j\Ga_k$ in $\mathfrak{so}_7$ such that
 \[\psi_2(e_1\wedge e_2\wedge e_3, \Ga_j\Ga_j.1)\neq 0\]
 which is equivalent to having $\Ga_j\Ga_k$ as a Cartan element. These elements are
 \[\Ga_2\Ga_5=2(E_{25}-E_{52})\quad \Ga_3\Ga_6=2(E_{36}-E_{63})\quad\Ga_4\Ga_7=2(E_{47}-E_{74}),\]
 and they act on the eigenvector $1$ by $-i$. Thus
 \begin{align*}
 p_2(e_1\wedge e_2\wedge e_3, 1)&=\psi_2(e_1\wedge e_2\wedge e_3, -i.1)\Ga_2\Ga_5+\psi_2(e_1\wedge e_2\wedge e_3, -i.1)\Ga_3\Ga_6\\
                                &+\psi_2(e_1\wedge e_2\wedge e_3, -i.1)\Ga_4\Ga_7\\
                               &= -i(\Ga_2\Ga_5+\Ga_3\Ga_6+\Ga_4\Ga_7)\\
                               &=-2i(E_{25}-E_{52}+E_{36}-E_{63}+E_{47}-E_{74})\in \mathfrak{so}_7.
 \end{align*}
If we use the definition of $\mathfrak{so}_7$ given in \cite{HUM}, then
  \[ p_2(e_1\wedge e_2\wedge e_3, 1)=2(h_1+h_2+h_3),\]
  where $h_i=E_{i+1,i+1}-E_{i+4,i+4}$ for $i=1,2,3$. Let $h=\begin{bmatrix} 1 & 0\\ 0 & 1\end{bmatrix}\in \s_2$, then
\[[x_\a,x_{-\a}]=\Big(-3h, 2(h_1+h_2+h_3)\Big)\in \s_2\oplus \mathfrak{so}_7.\]
 Having $[x_\a,x_{\a}]=[x_{-\a},x_{-\a}]=0$ shows that $\ff$ is three-dimensional. Moreover,
   \begin{align*}
   [[x_\a,x_{-\a}],x_\a] & =[\Big(-3h, 2(h_1+h_2+h_3)\Big),u_1\otimes (e_1\wedge e_2\wedge e_3)]\\
                         & = -3h.u_1\otimes (e_1\wedge e_2\wedge e_3) + u_1\otimes (2h_1+2h_2+2h_3)(e_1\wedge e_2\wedge e_3)\\
                         & = -3x_\a+u_1\otimes (\e_1+\e_2+\e_3)(h_1+h_2+h_3)(e_1\wedge e_2\wedge e_3)\\
                         & = -3x_\a+3x_\a=0
  \end{align*}
  Similarly we can show $[[x_\a,x_{-\a}],x_{-\a}]=0.$ This shows that $[\ff_{\0},\ff_{\1}]=0$.
 \end{proof}
 It follows that these varieties admit a rank variety description and, in particular, can be identified as subvarieties of $\ff_{\1}$, i.e.,
\begin{align*}
\fv_{(\ff,\ff_{\0})}(M)&=\fv_{\ff_{\1}}^{rank}(M)\\
                       &=\{y\in \ff_{\1}\mid \text{M is not projective as $U(\langle y \rangle)$-module}\}\cup\{0\}.
\end{align*}
Note that $\fv_{(\ff,\ff_{\0})}(S(\l_1))=\ff_{\1}$. We can use this detecting subsuperalgebra to give a geometric interpretation of the $z$-complexity. For instance,
\begin{cor}
$z_{\F}(S(\l_1))=\dim \fv_{(\ff,\ff_{\0})}(S(\l_1))=2$.
\end{cor}
In general:
\begin{thm}\ \\
\toto {1} If $S$ is a typical simple module, then
$$z_{\F}(S)=\dim \fv_{(\ff,\ff_{\0})}(S)=0.$$

\toto{2} If $S$ is an atypical simple module, then
$$z_{\F}(S)=\dim \fv_{(\ff,\ff_{\0})}(S)=2.$$

\end{thm}
\begin{proof}
If $S$ is a typical simple module, then $S$ is projective and the first part follows. If $S$ is atypical, then by \cite[Theorem~4.1.1]{kujawa},
$$\fv_{(\ff,\ff_{\0})}(S)=\fv_{(\ff,\ff_{\0})}(S(\l_1)),$$
thus $\dim \fv_{(\ff,\ff_{\0})}(S)=2$. The second part follows.
\end{proof}




%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% BIBLIOGRAPHY %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%



\end{document}
